Gromov-Witten invariants of symmetric products of projective space

by

Robert Silversmith

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Mathematics) in the University of Michigan 2017

Doctoral Committee:

Professor Yongbin Ruan, Chair Professor William Fulton Professor Leopoldo Pando Zayas Professor Karen Smith Associate Professor David Speyer



Robert Silversmith

rsilvers@umich.edu

ORCID id: 0000-0003-0508-5958

© Robert Silversmith 2017

My thesis is dedicated to the amazing Rohini Ramadas. It is impossible to describe the ways in which Rohini has contributed to this thesis and my life over the last six years. More than anyone else, Rohini has taught me how to learn math, explain math, enjoy math, do research, and navigate academia. Rohini is the center of my life and my perfect partner. I am lucky.

ACKNOWLEDGMENTS

I spent five years working with Yongbin Ruan, and it has been a journey. At the beginning, I didn't even know what area of math I was getting myself into, except that it had something to do with "orbifolds." Through my years of learning background material, and years of struggling with the concept of research, and finally some time actually doing research, Yongbin was there in our weekly meetings, answering my partly-formed questions and pushing me towards thinking from a researcher's point of view. It was a transformative experience and I am very grateful.

There is a long list of people from whose knowledge, wisdom, and generosity I have benefited. Karen Smith has been a constant force for good in my mathematical life, and without knowing it she chose my advisor for me. I have learned from Melissa Liu, Yu-jong Tzeng, Bill Fulton, Renzo Cavalieri, David Speyer, Davesh Maulik, Yaim Cooper, Aaron Pixton, Diane MacLagan, Sarah Koch, Anand Deopurkar, and Hsian-Hua Tseng, all of whom have taken the time to slowly answer questions and point out interesting directions. In particular, Melissa helped me through some of the computations in Chapter 5, and without her beautifully explained paper [46], I could not have written Section 4.6. David worked through Section 4.5 with me and made many helpful suggestions.

A great part of working with Yongbin was the buzz of activity that surrounds him; I have learned a lot from his other (former) students Mark Shoemaker, Yefeng Shen, Emily Clader, and Pedro Acosta, as well as his (former) postdoc Dustin Ross.

Many thanks to the staff in the math office: Stephanie Carroll, Carrie Berger, Kathryn Beeman, Chad Gorski, Beste Erel Windes, Brenae Smith, Molly Long, Wendy Summers, and especially Tara McQueen, whose time I have taken up far too much of.

I'm fortunate to have parents who understand the workings of graduate school and academia. They have given me amazing guidance and advice — when I was applying to grad schools, when I was taking classes, when I was doing research, and finally when I was applying for postdocs. I am even more fortunate to have wonderful, supportive, loving parents, who never doubted for a second that I could get a Ph.D. I am so grateful to them.

TABLE OF CONTENTS

Dedicatio	n	ii	
Acknowle	dgments	ii	
List of Figures			
Abstract			
Chapter			
1 Introdu	iction	1	
1.1	Enumerating curves	1	
1.2	Mirror theorems	3	
1.3	Statement of main theorem	7	
1.4	Motivation: the crepant resolution conjecture	8	
1.5	Proof techniques	9	
1.6	Organization of the thesis	.1	
1.7	Notation and conventions	.2	
	1.7.2 Cramba	2	
	1.7.2 Graphs	. <i>L</i>	
2 Orbifold Gromov-Witten theory			
2.1	Orbifolds and orbifold maps	4	
2.2	Orbifold curves	8	
2.3	Orbifolds and stacks	9	
2.4	Moduli spaces of stable maps	22	
	2.4.1 Orbifold stable maps	24	
2.5	Gromov-Witten invariants and twisted Gromov-Witten invariants	25	
2.6	The Givental cone	26	
3 Proper	ties of $\operatorname{Sym}^d \mathbb{P}^r$	9	
3.1	Symmetric product stacks	29	
	3.1.1 Two definitions of $\operatorname{Sym}^d X$	29	
	3.1.2 The tangent bundle to $\operatorname{Sym}^d X$	51	
	3.1.3 The inertia stack and Chen-Ruan cohomology of $\operatorname{Sym}^d X$	51	
3.2	Symmetric products of projective space	3	
	3.2.1 Notation for projective space	63	
	3.2.2 The Chen-Ruan cohomology of $\operatorname{Sym}^d \mathbb{P}^r$	3	

5.2.5	The action of $(\mathbb{C}^*)^{r+1}$ on Sym ^{<i>d</i>} \mathbb{P}^r
4 Localization ar	d virtual localization
 4.1 The loc 4.2 The T- 4.3 The T- 4.4 T-Fixe 4.5 The T- 4.5.1 4.5.2 4.6 The vir 	alization and virtual localization theorems
5 The Givental c	one is characterized by recursion relations
5.1 Charac 5.2 Charac	erization of the Givental cone $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$
6 The <i>1</i> -function	and mirror theorem for $\operatorname{Sym}^a \mathbb{P}^r$
6 The <i>1</i> -function 6.1 Statem 6.2 Proof c 6.2.1 6.2.2 6.2.3 6.3 $I_{\text{Sym}^d \mathbb{P}^{7}}$ 6.4 The ori 6.4.1 6.4.2 6.4.3 6.4.3 6.4.4	and mirror theorem for $\operatorname{Sym}^{a} \mathbb{P}^{r}$ 82ent of the main theorem82f Theorem 6.1.283 $I_{\operatorname{Sym}^{d} \mathbb{P}^{r}}(\mathbf{x}, Q, -z)$ satisfies condition (I)83 $I_{\operatorname{Sym}^{d} \mathbb{P}^{r}}(\mathbf{x}, Q, -z)$ satisfies condition (II)83 $I_{\operatorname{Sym}^{d} \mathbb{P}^{r}}(\mathbf{x}, Q, -z)$ satisfies condition (II)83 $I_{\operatorname{Sym}^{d} \mathbb{P}^{r}}(\mathbf{x}, Q, -z)$ satisfies condition (III)83 $I_{\operatorname{Sym}^{d} \mathbb{P}^{r}}(\mathbf{x}, Q, -z)$ satisfies condition (III)86 (t, \mathbf{x}, Q, z) is equal to the J-function of $\operatorname{Sym}^{d} \mathbb{P}^{r}$ 88gins of the I-function88Quasimaps88Graph spaces and I-functions89I-functions of orbifolds89I-functions of orbifolds, attempt 291

LIST OF FIGURES

1.1	A point of $\operatorname{Sym}^9 \mathbb{P}^2$ (or $\operatorname{Sym}^9 \mathbb{P}^2$) and a point of $\operatorname{Hilb}^{(9)}(\mathbb{P}^2)$ mapping to it	9
4.1	The T-graph of $\mathbb{P}^2 \times \mathbb{P}^2$ with the action of its dense torus, and the diagonal action of	
	$(\mathbb{C}^*)^3$, respectively	37
4.2	The torus orbits in $(\mathbb{P}^1)^2$	40
4.3	The graph $\Gamma_{\mathbb{P}^2,\mathbb{C}^*}$ and a picture of the 1-dimensional \mathbb{C}^* -orbits	42
4.4	The graph $\Gamma_{\mathcal{B}\ell_{P_1}\mathbb{P}^2,\mathbb{C}^*}$ and a picture of the 1-dimensional \mathbb{C}^* -orbits	44
4.5	A point with one dimensional T-orbit in Sym ¹⁷ \mathbb{P}^2 , with $i_1 = 1$ and $i_2 = 2$	46
4.6	The T-graph, and a picture of 1-dimensional orbits, for $\text{Sym}^3 \mathbb{P}^2$	47
4.7	Combining edges	52
4.8	Combining two pairs of edges	52
4.9	A portion of a map in $\overline{\Psi^{-1}(\tilde{\Gamma}_0)}$, with $\eta = 1$ and $q(e) = 3$	55

ABSTRACT

Gromov-Witten invariants are numbers that roughly count curves of a fixed type on an algebraic variety X. For example, for 3 general points and 6 general lines in $X = \mathbb{P}^3$, there are exactly 190 twisted cubics intersecting all of them, so 190 is a Gromov-Witten invariant of \mathbb{P}^3 . Gromov-Witten invariants appear in algebraic geometry and string theory. In the special case when X is a toric variety, Kontsevich found a method to compute any Gromov-Witten invariant of X. Givental and Lian-Liu-Yau used Kontsevichs algorithm to prove a mirror theorem, which states that Gromov-Witten invariants of X have an interesting rigid structure predicted by physicists. The main result of this thesis is a mirror theorem for the nontoric orbifold $X = \text{Sym}^d(\mathbb{P}^r)$, the symmetric product of projective space, which parametrizes unordered d-tuples of points in \mathbb{P}^r .

CHAPTER 1

Introduction

Over the last 20 years, following predictions from string theory [10], mathematicians have proven a series of results known as *mirror theorems*. These theorems reveal elegant patterns and structures embedded in the collection of (usually genus-zero) Gromov-Witten invariants, or counts of curves satisfying geometric constraints, of a fixed target variety or orbifold X. They also allow for indirect computation of these invariants in many cases where direct computation is combinatorially difficult. However, the scope of these results is essentially limited to the world of toric geometry; in every case, X is either a complete intersection in a toric stack, or a stack admitting a toric degeneration.

The goal of this thesis is to develop tools for mirror symmetry outside of toric geometry. The main theorem (Theorem 6.1.2) is a genus-zero mirror theorem for the orbifold $\operatorname{Sym}^d \mathbb{P}^r$ parametrizing unordered *d*-tuples of points in projective *r*-space. Because $\operatorname{Sym}^d \mathbb{P}^r$ is nontoric, and further nonabelian, this result is not possible using existing techniques. Theorem 6.1.2 gives an explicit formula for a generating function of Gromov-Witten invariants of $\operatorname{Sym}^d \mathbb{P}^r$. This is the only known mirror theorem for a nonabelian orbifold, besides single-point orbifolds $[\bullet/G]$.

1.1 Enumerating curves

From an algebro-geometric perspective, Gromov-Witten stems from enumerative geometry of curves in varieties, motivated by questions like these¹:

- 1. How many plane conic curves pass through five general specified points?
- 2. How many twisted cubic curves in 3-space pass through three general specified points and intersect six general specified lines?
- 3. How many lines are contained in a general cubic surface in 3-space?

¹Here "plane" refers to \mathbb{CP}^2 and "3-space" refers to \mathbb{CP}^3 . We work over \mathbb{C} throughout.

Classically, the answers are 1, 190, and 27 respectively. A natural kind of question generalizing these is the following:

Question 1.1.1. Fix² a smooth complex projective variety X, a genus $g \in \mathbb{Z}_{\geq 0}$, a homology class $\beta \in H_2(X, \mathbb{Z})$, and a list of cohomology classes $\gamma_1, \ldots, \gamma_n \in H^*(X, \mathbb{Z})$. How many smooth genus g curves in X of homology class β intersect generic subvarieties V_1, \ldots, V_n representing the classes $\gamma_1, \ldots, \gamma_n$?

Note that in the case $X = \mathbb{P}^2$, g = 0, $\beta = 2[L]$, n = 5, and $\gamma_1 = \cdots = \gamma_5 = [pt]$, Question 1.1.1 is precisely Question (1) above, and in the case $X = \mathbb{P}^3$, g = 0, $\beta = 3[L]$, n = 9, $\gamma_1 = \gamma_2 = \gamma_3 = [pt]$, and $\gamma_4 = \cdots = \gamma_9 = [L]$ it is precisely Question 2. For a general cubic surface X = S, with g = 0, $\beta = [L]$, and n = 0, Question 1.1.1 is the same as question 3.

In practice, these questions are usually very difficult to answer. In Gromov-Witten theory, the objects of study are not the correct answers, but certain natural incorrect answers called *Gromov-Witten invariants*.

In detail, the counting problem can be set up as an intersection theory problem on a moduli space $\mathcal{M}_g(X,\beta)$ parametrizing genus g curves of degree β in X, as follows. The condition that a curve intersects V_i defines a closed subvariety $\mathcal{M}_i \subseteq \mathcal{M}_g(X,\beta)$, where

$$\mathcal{M}_i = \{ C \in \mathcal{M}_g(X, \beta) : C \cap V_i \neq \emptyset \}.$$

The intersection $\bigcap_{i=1}^{n} \mathcal{M}_i$ is the set of genus g, degree d curves in X that intersect all of the subvarieties V_1, \ldots, V_n . Therefore the number of points in this intersection is the correct answer to Question 1.1.1.

A strategy to compute this number is the following. $\mathcal{M}_g(X,\beta)$ is not compact. We compactify $\mathcal{M}_g(X,\beta)$ to get a space $\overline{\mathcal{M}}_g(X,\beta)$, and we obtain compactified subvarieties $\overline{\mathcal{M}}_i \subseteq \overline{\mathcal{M}}_g(X,\beta)$ such that $\overline{\mathcal{M}}_i \cap \mathcal{M}_g(X,\beta) = \mathcal{M}_i$. We calculate the class of the intersection $\bigcap_{i=1}^n \overline{\mathcal{M}}_i$ using the intersection product $\prod_{i=1}^n [\overline{\mathcal{M}}_i]$ in the cohomology ring of $\overline{\mathcal{M}}_g(X,\beta)$. The answer is some multiple of the class of a point, and this number is a Gromov-Witten invariant, denoted $\langle \gamma_1, \ldots, \gamma_n \rangle_{g,n,\beta}^X$. The three motivational questions correspond to the Gromov-Witten invariants:

$$\langle [pt], [pt], [pt], [pt], [pt], [pt] \rangle_{0,5,2[L]}^{\mathbb{P}^2}, \quad \langle [pt], [pt], [pt], [L], [L], [L], [L], [L], [L] \rangle_{0,9,3[L]}^{\mathbb{P}^3}, \text{ and } \langle \rangle_{0,0,[L]}^{S}.$$

The reason that $\langle \gamma_1, \ldots, \gamma_n \rangle_{g,n,\beta}^X$ may be an incorrect³ answer to Question 1.1.1 is that even if the subspaces $\overline{\mathcal{M}}_i$ intersect transversely for general choices of V_i , the intersection $\bigcap_{i=1}^n \overline{\mathcal{M}}_i$ may

²To be precise, this is not exactly a generalization of the question (3), since (3) works with a *generic* cubic surface, whereas here we have fixed a variety X. Making sense of this discrepancy (i.e. making the answer invariant under deformations of X) leads to the "virtual" intersection theory discussed in Section 2.4.

³Indeed, $\langle \gamma_1, \ldots, \gamma_n \rangle_{g,n,\beta}^X$ is a rational number, not even an integer. This is because $\overline{\mathcal{M}}_g(X,\beta)$ is an Deligne-Mumford stack/orbifold rather than a variety.

contain points of the boundary $\overline{\mathcal{M}}_g(X,\beta) \smallsetminus \mathcal{M}_g(X,\beta)$.

The compactification $\overline{\mathcal{M}}_g(X,\beta)$ and the marked versions $\overline{\mathcal{M}}_{g,n}(X,\beta)$ are introduced in Section 2.4; they are spaces of maps $(f : C \to X)$ from (possibly singular) curves to X, called *stable* maps. (The interior $\mathcal{M}_g(X,\beta) \subseteq \overline{\mathcal{M}}_g(X,\beta)$ consists of *embeddings* of *smooth* curves, i.e. actual smooth curves contained in X.) In fact, the spaces $\overline{\mathcal{M}}_{g,n}(X,\beta)$ are not really compactifications, in that $\overline{\mathcal{M}}_{g,n}(X,\beta)$ may contain irreducible components other than the Zariski closure of $\mathcal{M}_{g,n}(X,\beta)$. Similarly, the subvariety $\overline{\mathcal{M}}_i$ may be larger than the Zariski closure of \mathcal{M}_i .

In this thesis, the main case of interest is $X = \text{Sym}^d \mathbb{P}^r$, which is not a smooth projective variety; it is a smooth complex orbifold. The entire discussion above works with little modification (see Section 2.4) for X an orbifold.

1.2 Mirror theorems

A mirror theorem for a smooth variety or orbifold X states that a generating function f of Gromov-Witten invariants of X is equal to an explicit analytic function I_X (an "I-function"). Such a statement asserts that there is a rigid structure to the various Gromov-Witten invariants of X. Because a Gromov-Witten invariant has many inputs (i.e. the genus g, the class β , and the list of cohomology classes $\gamma_1, \ldots, \gamma_n$), the generating function f and the explicit function I will have many variables.

The statement of a mirror theorem usually involves generating functions of *descendant Gromov-Witten invariants*, a generalization of Gromov-Witten invariants. These generating functions are easier to compute with than generating functions of usual Gromov-Witten invariants. A descendant Gromov-Witten invariant has extra inputs a_1, \ldots, a_n , and is denoted $\langle \psi^{a_1} \gamma_1, \ldots, \psi^{a_n} \gamma_n \rangle_{g,n,\beta}$. The symbols $\psi^{a_1}, \ldots, \psi^{a_n}$ represent certain cohomology classes $\psi_1^{a_1}, \ldots, \psi_n^{a_n} \in H^*(\overline{\mathcal{M}}_{g,n}(X,\beta), \mathbb{Q})$ (see Definition 2.4.3), and the descendant Gromov-Witten invariant is defined as an intersection product with extra factors (see Section 2.5 for the precise definition):

$$\langle \psi^{a_1} \gamma_1, \dots, \psi^{a_n} \gamma_n \rangle_{g,n,\beta} := \prod_{i=1}^n [\overline{\mathcal{M}}_i] \cdot \psi_i^{a_i}.$$

Gromov-Witten invariants as discussed in the previous section are the case $a_1 = \cdots = a_n = 0$ of descendant Gromov-Witten invariants. From now on, we use the term "Gromov-Witten invariants" to include descendant Gromov-Witten invariants.

Example 1.2.1. Givental [27] (and Lian-Liu-Yau [45] in slightly different language) proved the following mirror theorem for projective space.

Theorem 1.2.2 (Givental, Lian-Liu-Yau). Let

$$J(t,Q,z) := z + tH + \sum_{n,\beta \ge 0} \sum_{i=0}^{r} \frac{Q^{\beta}}{n!} \langle tH, \dots, tH, \frac{H^{i}}{z - \psi} \rangle_{0,n,\beta}^{\mathbb{P}^{r}} H^{r-i}.$$
 (1.1)

Here the symbol $\frac{1}{z-\psi}$ means the series expansion $\sum_{m\geq 0} \frac{\psi^m}{z^{m+1}}$. The series J(t,Q,z) is a cohomology-valued generating function; explicitly, $J(t,Q,z) \in H^*(\mathbb{P}^r,\mathbb{Q})[[Q]]((z^{-1}))$. Then

$$J(t,Q,z) = I(t,Q,z) := z \sum_{\beta \ge 0} \frac{e^{t(H/z+\beta)}Q^{\beta}}{\prod_{\gamma=1}^{\beta} (H+\gamma z)^{r+1}}.$$
(1.2)

We should note that this example (unlike the next one, Example 1.2.3) is not useful from the point of view enumerating curves in \mathbb{P}^r , for two reasons. First, the cohomology classes tH appearing in the coefficients of J(t, Q, z) mean that we are counting curves that intersect a fixed hyperplane — but in projective space, every curve meets every hyperplane, so this is no restriction at all. Second, almost all of the coefficients are descendant Gromov-Witten invariants, which do not have a clear enumerative significance.

In fact, this theorem can be significantly generalized; the equality (1.2) is obtained from the equality on the bottom of page 20 in [14] by setting many variables equal to zero. The generating function in that generalization *does* have enumerative significance, since every nondescendant Gromov-Witten invariant appears as a coefficient. (For example, the number 190 of twisted cubics intersecting 6 general lines and 3 general points in space is a coefficient, when r = 3.) We did not state it here because the statement requires the formalism of the Givental cone, which we discuss next.

Usually, mirror theorems cannot be stated as simply as Theorem 1.2.2; instead, they are usually stated using a framework introduced by Givental to encode genus-zero Gromov-Witten invariants of X. Givental's insight was to focus on a special class of generating functions, those of the form:

$$\mathbf{f} := -z + \mathbf{t}(t, Q, z) + \sum_{\substack{n \ge 0, \beta \text{ effective}\\\beta \neq 0 \text{ or } n \ge 2}} \frac{Q^{\beta}}{n!} \langle \mathbf{t}(t, Q, \psi), \dots, \mathbf{t}(t, Q, \psi), \frac{\gamma_{\phi}}{-z - \psi} \rangle_{0, n, \beta}^{X} \gamma^{\phi}.$$
(1.3)

Some explanation is required. The formal variable Q keeps track of the degree $\beta \in H_2(X, \mathbb{Z})$. Here $\mathbf{t}(t, Q, z)$ is a polynomial in z with coefficients in $H^*(X, \mathbb{Q})[[t, Q]]$. Note that the coefficients are well-defined (sums of) (descendant) Gromov-Witten invariants, where $\mathbf{t}(t, Q, \psi)$ means that ψ is formally substituted into $\mathbf{t}(t, Q, z)$. In Theorem 1.2.2, J(t, Q, -z) is of the form (1.3), with $\mathbf{t}(t, Q, z) = tH$.

Givental gave a geometric interpretation of generating functions of the form (1.3). Namely,

there is a (germ of an) infinite-dimensional scheme \mathcal{L}_X , called the *Givental cone* of X (see Section 2.6), such that generating functions **f** as in (1.3) are in natural bijection with $\mathbb{C}[[t, Q]]$ -valued points of \mathcal{L}_X . Using this setup, the typical statement of a mirror theorem is of the form, "the following explicit function $I_X(t, Q, -z) \in H^*(X, \mathbb{Q})[[t, Q]]((z^{-1}))$ is a $\mathbb{C}[[t, Q]]$ -valued point of the Givental cone \mathcal{L}_X ." One may then write $I_X(t, Q, -z)$ as a generating function of Gromov-Witten invariants of the form (1.3), where $-z + \mathbf{t}(t, Q, z)$ is the part of $I_X(t, Q, -z)$ with nonnegative powers of z in the Laurent expansion of $I_X(t, Q, -z)$ in z^{-1} .

Example 1.2.3. Givental [28] and Lian-Liu-Yau [44] also proved a mirror theorem for the general quintic threefold $\mathscr{Q} \subseteq \mathbb{P}^4$. Stating this theorem requires the language of the Givental cone.

Theorem 1.2.4 (Givental, Lian-Liu-Yau). Let

$$I(t,Q,z) := z \sum_{\beta \ge 0} e^{t(H/z+\beta)} Q^{\beta} \frac{\prod_{\gamma=1}^{5\beta} (5H+\gamma z)}{\prod_{\gamma=1}^{\beta} (H+\gamma z)^5},$$
(1.4)

where *H* is the pullback of the hyperplane class on \mathbb{P}^4 . Then I(t, Q, -z) is a $\mathbb{C}[[t, Q]]$ -valued point of the Givental cone $\mathcal{L}_{\mathcal{Q}}$.

By the definition of the $\mathcal{L}_{\mathscr{Q}}$, I(t, Q, -z) is equal to a generating function of Gromov-Witten invariants, but in this example it is difficult to say which one. In order to find out, we need to identify the terms with nonnegative power of z, in the Laurent expansion of I(t, Q, -z) in z^{-1} . Let $-z + \mathbf{t}(t, Q, z) \in H^*(\mathscr{Q}, \mathbb{Q})[[t, Q, z]]$ be the sum of all such terms. We calculate the coefficient of Q in $\mathbf{t}(t, Q, z)$ as follows. The first terms of I(t, Q, z) are (using $H^4 = 0$):

$$I(t,Q,z) = ze^{tH/z} + zQe^{t(H/z+1)}\frac{(5H+z)(5H+2z)(5H+3z)(5H+4z)(5H+5z)}{(H+z)^5} + O(Q^2)$$
$$= ze^{tH/z} + zQe^{t(H/z+1)}\frac{10625H^3z + 5625H^2z^2 + 1370Hz^3 + 120z^4}{(H+z)^4} + O(Q^2)$$

Writing $\frac{1}{(H+z)^5} = \frac{1}{z^5} - \frac{5H}{z^6} + 15\frac{H^2}{z^7} - 35\frac{H^3}{z^8}$, and $e^{t(H/z+1)} = e^t(1 + tH/z + O(z^{-2}))$ we have

$$I(t,Q,-z) = -z + tH + Qe^{t}(-120z + (770 + 120t)H) + O(z^{-1}) + O(Q^{2}).$$

In other words, $\mathbf{t}(t,Q,z) = tH + Qe^t(-120z + (770 + 120t)H) + O(Q^2)$. Let $\mathbf{t}_1(t,Q,z) = tH + Qe^t(-120z + (770 + 120t)H) + O(Q^2)$.

 $tH + Qe^t(-120z + (770 + 120t)H)$. Then by the definition of $\mathcal{L}_{\mathscr{Q}}$, we find

$$I(t,Q,-z) = -z + \mathbf{t}_{1}(t,Q,z) + \sum_{\substack{n,\beta \ge 0 \\ \beta > 0 \text{ or } n \ge 2}} \sum_{i=0}^{3} \frac{Q^{\beta}}{n!} \langle \mathbf{t}_{1}(t,Q,\psi), \dots, \mathbf{t}_{1}(t,Q,\psi), \frac{H^{i}}{-z-\psi} \rangle_{0,n+1,\beta}^{\mathscr{Q}} \frac{H^{3-i}}{5} + O(Q^{2}).$$
(1.5)

Repeating this process with the Q^2 term of I(t, Q, -z) (and subsequently higher order terms), we can calculate t(t, Q, z) to arbitrarily high order in Q. That is, we can identify I(t, Q, -z) as a generating function of Gromov-Witten invariants.

The invariants in this generating function do not immediately appear to have enumerative significance, similarly to the last example. However, one can use the fact that \mathscr{Q} is a Calabi-Yau threefold to express any degree β Gromov-Witten invariant of \mathscr{Q} explicitly in terms the invariant $\langle \rangle_{0,0,\beta}^{\mathscr{Q}}$. (This is the Gromov-Witten invariant corresponding to the enumerative question of how many degree β rational curves lie on \mathscr{Q} . This is essentially the only enumerative question one can ask about genus zero curves on \mathscr{Q} .) The process uses certain recursive identities called the *string equation, dilaton equation,* and *divisor equation,* see Section 26.3 of [32]. This allows one to recover all "enumerative" (non-descendant) Gromov-Witten invariants, in particular the invariants $\langle \rangle_{0,0,\beta}^{\mathscr{Q}}$, from Theorem 1.2.4. (This is how Givental and Lian-Liu-Yau verified the predictions of [10].) We illustrate this is Example 2.6.5, which computes $\langle \rangle_{0,0,1}^{\mathscr{Q}} = 2875$ and $\langle \rangle_{0,0,2}^{\mathscr{Q}} = 609250 + 2875/8$.

So far, mirror theorems have been proven in genus zero for:

- Quintic threefolds [28, 44],
- Complete intersections in toric varieties [27, 45],
- Complete intersections in homogeneous spaces [39],
- Toric bundles [8],
- Toric orbifolds/stacks and certain complete intersections, and [12, 18, 17]
- Toric stack bundles [35].

There are also some results in higher genus, see e.g. [57, 43, 25]. The proofs of all of these theorems rely on toric geometry, specifically the fact that toric varieties and orbifolds admit a torus action with isolated fixed points and 1-dimensional orbits. (See Section 1.5 and 4.2.)

1.3 Statement of main theorem

The main result is a mirror theorem for $X = \text{Sym}^d \mathbb{P}^r$, the orbifold parametrizing unordered *d*-tuples of points of \mathbb{P}^r . Using the language of Section 1.2, the statement is as follows.

Theorem 6.1.2. Introduce formal variables Q, z, $\{t_i\}_{0 \le i \le r}$, and $\{x_{\varpi}\}_{\varpi \in \text{Part}(d)}$, where Part(d) is the set of partitions of d. Let

$$I_{\text{Sym}^{d} \mathbb{P}^{r}}(t, \mathbf{x}, Q, z) \coloneqq z \sum_{\sigma \in \text{Part}(d)} 1_{\sigma} \sum_{\beta \ge 0} \exp\left(\sum_{i=0}^{r} t_{i}([H_{\sigma,i}]/z + \beta)\right) Q^{\beta} \sum_{\substack{\mathbb{Z}_{>0}\text{-labels } L = (L_{\eta}) \text{ of the parts of } \sigma \text{ with sum } \beta}} \left(\sum_{\substack{\mathbf{k} = (k_{\varpi})_{\varpi} \in \text{Part}(d) \\ k_{\varpi} \ge 0}} H(\sigma \prod_{\varpi} \varpi^{k_{\varpi}}) \prod_{\varpi \in \text{Part}(d)} \frac{x_{\varpi}^{k_{\varpi}}}{z^{k_{\varpi}} k_{\varpi}!}\right)$$

$$\cdot \left(\frac{|S_{\sigma}|}{|S_{\sigma,L}|}\right) \left(\prod_{\eta \in \sigma} \frac{1}{\prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(H_{\sigma,\eta,i} + \frac{\gamma}{\eta}z\right)}\right).$$
(1.6)

Then $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, -z)$ is a $\mathbb{C}[[\{t_i\}, Q, \{x_{\varpi}\}]]$ -valued point of the Givental cone $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$. Here we use the notation:

- 1_σ ∈ H^{*}_{CR,T}(Sym^d ℙ^r, ℚ) is the Chen-Ruan cohomology class of the twisted sector corresponding to the partition σ (see Section 3.2.2),
- $[H_{\sigma,i}]$ and $[H_{\sigma,\eta,i}]$ are hyperplane classes defined in Section 3.1.3,
- H(σ Π_∞ ∞<sup>k_∞</sub>) is the number of ways of factoring 1 ∈ S_d as a product a₁ ··· a_{1+∑k_∞}, where the conjugacy classes (i.e. partitions) of the permutations a_j are given by the list (σ Π_∞ ∞^{k_∞}), and
 </sup>
- S_{σ} and $S_{\sigma,L}$ are automorphism groups, of the partition S_{σ} and the labeled partition $S_{\sigma,L}$, respectively (see Section 1.7).

As in Example 1.2.3, we may write $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, -z)$ as a generating function of Gromov-Witten invariants by calculating the terms $-z + \mathbf{t}(t, \mathbf{x}, Q, z)$ of its Laurent expansion in z^{-1} with nonnegative powers of z. We get

$$\mathbf{t}(t, \mathbf{x}, Q, z) = \theta := \sum_{\sigma} \sum_{i=0}^{r} t_i [H_{\sigma,i}] + \sum_{\varpi \in \operatorname{Part}(d)} x_{\varpi} 1_{\varpi}.$$
(1.7)

The following corollary is immediate:

Corollary 6.3.1. Let

$$J_{\operatorname{Sym}^{d}\mathbb{P}^{r}}(Q,\theta,z) = z + \theta + \sum_{\beta,n} \frac{Q^{\beta}}{n!} \left\langle \theta, \dots, \theta, \frac{\gamma_{\phi}}{z - \overline{\psi}} \right\rangle_{0,n+1,\beta}^{\operatorname{Sym}^{d}\mathbb{P}^{r},T} \gamma^{\phi},$$

where θ is as in (1.7). Then $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, z) = J_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, z)$.

Corollary 6.3.1 allows us to recover some, but not all, non-descendant Gromov-Witten invariants. This is because, when we vary the coefficients t_i and x_{ϖ} , θ takes values in a proper subspace of $H^*_{CR,T}(\text{Sym}^d \mathbb{P}^r, \mathbb{Q})$. In the future we hope to generalize Theorem 6.1.2 along the lines of Ciocan-Fontanine and Kim's "big" mirror theorem for toric varieties [14]. For such a theorem, there would be no analog of Corollary 6.3.1, since the *I*-function would contain arbitrarily higher powers of *z*.

1.4 Motivation: the crepant resolution conjecture

Mirror theorems show that the Gromov-Witten invariants of an orbifold X have *internal* structure, a natural question is how Gromov-Witten invariants of X are related to those of *another* orbifold Y, when there is a geometric relationship between X and Y. We focus on one example of such a relationship.

An orbifold X has an underlying *singular* variety (or algebraic space) \underline{X} , called the coarse moduli space of X, with a natural map $X \to \underline{X}$ (see Section 2.3). Since \underline{X} is not smooth, it does not have Gromov-Witten invariants. However, suppose Y is a crepant resolution of singularities of \underline{X} , i.e. a resolution such that K_X pulls back to K_Y :



(Such resolutions have long been studied in algebraic geometry, see [52].) Physicists predicted that X and Y are equivalent from the point of view of mirror symmetry. This claim was formulated as a precise mathematical conjecture, the crepant resolution conjecture, by Ruan and others ([53, 9, 20]). The conjecture states that the the Gromov-Witten invariants of X are related in an explicit way to those of Y: there is a certain reversible transformation that turns a generating function of Gromov-Witten invariants of X into a generating function of Gromov-Witten invariants of Y.

The crepant resolution conjecture has been proved for toric orbifolds X, when Y is a toric crepant resolution [19]. The proof is via the mirror theorem for toric orbifolds [18]; the mirror theorem provides explicit analytic functions I_X and I_Y , which are directly compared. The conjecture has also been proven for the non-proper orbifolds $\operatorname{Sym}^d \mathbb{C}^2$ [9] (whose moduli spaces of stable maps $\overline{\mathcal{M}}_{q,n}(\operatorname{Sym}^d \mathbb{C}^2, \beta)$ exist only for $\beta = 0$), and by explicit computations for $\operatorname{Sym}^2 \mathbb{P}^2$ [56].





Unordered 9-tuple of points in \mathbb{P}^2

Schematic picture of the ideal: $(x + y, x^2) \cap (yz - (x - 2z)^2 + z^2, x^3)$ $\cap (x - z, y - 2z) \cap (z^2, z(x - 2y), (x - 2y)^2)$

Figure 1.1: A point of $\operatorname{Sym}^9 \mathbb{P}^2$ (or $\operatorname{Sym}^9 \mathbb{P}^2$) and a point of $\operatorname{Hilb}^{(9)}(\mathbb{P}^2)$ mapping to it

We are interested in the following case of this conjecture. The Hilbert scheme $\operatorname{Hilb}^{(d)}(\mathbb{P}^2)$ of points in \mathbb{P}^2 is a smooth scheme that is a moduli space for length d subschemes of \mathbb{P}^2 . Such a subscheme S is a disjoint union of (at most d) subschemes S_1, \ldots, S_k , of lengths $d_1 + \cdots + d_k = d$, concentrated at distinct single points P_1, \ldots, P_k . This defines an unordered d-tuple of points of \mathbb{P}^2 , consisting of d_i copies of P_i for $i = 1, \ldots, k$. (See Figure 1.1.) This does not define a regular map $\operatorname{Hilb}^{(d)}(\mathbb{P}^2) \to \operatorname{Sym}^d \mathbb{P}^2$, but it does define a regular (birational) map $\operatorname{Hilb}^{(d)}(\mathbb{P}^2) \to \operatorname{Sym}^d \mathbb{P}^2$, and one can show that this map is a crepant resolution. This gives a diagram



(More generally, such a resolution exists for symmetric products of any smooth surface. For symmetric products of higher-dimensional varieties X, there is still a map $\operatorname{Hilb}^{(d)}(X) \to \operatorname{Sym}^d X$, but is not generally birational, nor is $\operatorname{Hilb}^{(d)}(X)$ smooth.) The goal is to prove the crepant resolution conjecture in this case using the method of [19], but to do so one must prove two mirror theorems; one for $\operatorname{Sym}^d \mathbb{P}^2$ and one for $\operatorname{Hilb}^{(d)}(\mathbb{P}^2)$. The case r = 2 of Theorem 6.1.2 is the first of these.

1.5 Proof techniques

There is a tool called Atiyah-Bott torus localization, which can be used to simplify computations of integrals on an orbifold \mathcal{M} when \mathcal{M} admits an action of a torus T (see Chapter 4). Specifically, torus localization expresses any integral over \mathcal{M} as an integral over the T-fixed locus \mathcal{M}^T (See Section 4.1.

In Gromov-Witten theory, if an orbifold X admits a T-action, then so does the moduli space $\overline{\mathcal{M}}_{g,n}(X,\beta)$. Specifically, $\overline{\mathcal{M}}_{g,n}(X,\beta)$ is a space of maps to X, and T acts by postcomposition. Since Gromov-Witten invariants are defined as intersection products (integrals) on $\overline{\mathcal{M}}_{g,n}(X,\beta)$, torus localization can be applied. Using the natural $T = (\mathbb{C}^*)^{r+1}$ -action on \mathbb{P}^r , Kontsevich [41] developed an algorithm to compute any Gromov-Witten invariant of a complete intersection in \mathbb{P}^r . The algorithm involves a combinatorial description of the geometry of the T-fixed locus $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^r,\beta)^T$.

Givental found a recursive structure in Kontsevich's algorithm, which implies that when X is a complete intersection in \mathbb{P}^r (or more generally a smooth projective toric variety), any generating function of the form (1.3) satisfies a certain recursion relation. Furthermore, Givental showed that an element of $H^*(X, \mathbb{Q})[[t, Q]]((z^{-1}))$ is of the form (1.3) *if any only if* it satisfies this recursion. In other words, the recursion relation exactly characterizes $\mathbb{C}[[t, Q]]$ -valued points of \mathcal{L}_X .

Using predictions from string theory, Givental produced *I*-functions $I_X(t, Q, z)$, and showed via a direct check that $I_X(t, Q, -z)$ satisfies the recursion relation. This proved Theorem 1.2.4, and after finding the initial terms t(t, Q, z), proved Theorem 1.2.2.

This method has been streamlined since Givental's proof, by Kim [39], Brown [8] (while proving the mirror theorem for toric bundles), and Coates-Corti-Iritani-Tseng [18] (while proving the mirror theorem for toric orbifolds). The argument in [18] relies on calculations of Johnson [36] and Liu ([46], Section 9) involving toric orbifolds, which are analogous to (but much more subtle than) Kontsevich's calculations in projective space.

We use torus localization to prove Theorem 6.1.2, and our proof has the same overall structure as that of the mirror theorem for toric stacks [18]. We use the natural diagonal action of $T = (\mathbb{C}^*)^{r+1}$ on $\operatorname{Sym}^d \mathbb{P}^r$, which as above induces a *T*-action on the moduli spaces $\overline{\mathcal{M}}_{g,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ for all g, n, and β (though we need only g = 0).

We immediately run into one of the main obstacles in the proof of Theorem 6.1.2. Applying Kontsevich's algorithm or Givental's recursive argument requires a description of the fixed locus $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)^T$. (In particular, one must be able to effectively compute integrals over the fixed locus, and be able to compute the normal bundle.) For X a toric orbifold, Kontsevich's description of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$ was completely combinatorial. This is essentially due to the fact that T-fixed stable maps to X are closely related to 1-dimensional T-orbits in X, and in a toric orbifold (with its dense torus action) these orbits are isolated. However, in $\operatorname{Sym}^d \mathbb{P}^r$ with its diagonal $T = (\mathbb{C}^*)^{r+1}$ -action, the 1-dimensional T-orbits are not isolated; they come in positive dimensional families. (In other words, one can find points that are very close together, whose T-orbits are 1-dimensional and distinct.) In order to carry out the argument, we must first describe each family explicitly (see Theorem 4.5.23). Our description applies in more generality, for example to symmetric products of toric varieties. In Chapters 4 and 5, we carry out the analog of the Johnson and Liu's toric computations, which are needed to compute the relevant integrals over the fixed loci $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)^T$. Altogether, the description of the *T*-fixed locus and these calculations allow us to prove the analog of Givental's recursion, which says that an element of $H^*_{CR,T}(\operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q})[[t, Q]]((z^{-1}))$ is a point of $\mathcal{L}_{\operatorname{Sym}^d \mathbb{P}^r}$ if and only if it satisfies a certain recursion (Theorem 5.1.4).

Finally, we show directly, via Tseng's orbifold quantum Riemann-Roch theorem [54], that the series $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, -z)$ also satisfies these recursions.

It is natural to ask where these complicated functions (1.2), (1.4), and (1.6) come from. Besides preditions by physicists, there is recent work of Ciocan-Fontanine, Kim, and others [15, 13, 14, 12] (with origins in Bertram's [7] for $X = \mathbb{P}^r$) showing that in many cases, the *I*-function can be written in terms of *quasimap invariants* [13]. In particular, the authors produce *I*-functions for a large class of targets *X*, including toric orbifolds and certain complete intersections in toric orbifolds. However, we could not make rigorous sense of the method for $X = \text{Sym}^d \mathbb{P}^r$. In Section 6.4, we informally discuss how we found $I_{\text{Sym}^d \mathbb{P}^r}$, and conjecture the existence of a particular moduli space that would allow one to produce a candidate function I_X for quite general orbifolds *X*.

Finally, we would like to draw the reader's attention to a technical aspect of the recursion in Theorem 5.1.4, condition (II). The recursion expresses Laurent coefficients in f with negative exponents in terms of those with positive exponents. We have not seen this type of recursive structure before — it is a certain generalization of the one in Theorem 41 of [18] — but suspect it will be useful in more generality. We give a more detailed discussion of this point in Section 5.2.

1.6 Organization of the thesis

Section 1.7 sets up notational and combinatorial conventions. Chapter 2 gives background on orbifolds and orbifold Gromov-Witten theory. Chapter 3 defines and gives important properties of symmetric products, especially of projective space. In Chapter 4, we introduce Atiyah-Bott torus localization and virtual localization for moduli spaces of stable maps, and carry out the various calculations required to apply localization to compute Gromov-Witten invariants. Chapter 5 proves the recursive characterization (Theorem 5.1.4) of the Givental cone $\mathcal{L}_{Sym^d \mathbb{P}^r}$. Finally, Chapter 6 defines the explicit series $I_{Sym^d \mathbb{P}^r}$, and proves that it lies on $\mathcal{L}_{Sym^d \mathbb{P}^r}$.

1.7 Notation and conventions

1.7.1 Multisets, partitions, and multipartitions

A *finite multiset* ϖ is an unordered finite collection of elements a (we write $a \in \varpi$), possibly appearing more than once. Multisets are denoted with braces, e.g. $\{a, a, b\}$, and we will clarify if an object is a set or a multiset when necessary. We write $\text{Mult}(\varpi, a)$ for the number of times that a appears in ϖ . If a multiset is indexed by another multiset, we write e.g. $\{a_i\}_{i\in I}$. A *submultiset* $\varpi' \subseteq \varpi$ is a multiset such that for $\text{Mult}(\varpi', a) \leq \text{Mult}(\varpi, a)$ for $a \in \varpi'$. Unions are always taken to be disjoint, e.g. $\{a, a, b\} \cup \{a, c\} = \{a, a, a, b, c\}$. The *cardinality* or *length* of a multiset is the number of elements, including multiplicities. We take sums and products over multisets with multiplicity, e.g. $\sum_{\eta \in \{1,2,2\}} \eta = 5$.

For $d \in \mathbb{Z}_{>0}$, a partition of d is a multiset of positive integers whose sum (with multiplicities) is d. The (finite) set of partitions of d is denoted Part(d). The ones partition of d is the multiset $\{1, ..., 1\}$ of size d. A nonnegative ordered partition of d is an ordered tuple of nonnegative integers whose sum is d. The (finite) set of nonnegative ordered partitions of d of length r is denoted ZPart(d, r).

If D is a tuple of positive integers, a *multipartition of* D is a tuple $(\varpi_d)_{d\in D}$, with ϖ_d a partition of d. The (finite) set of multipartitions of D is denoted MultiPart(D). The *ones multipartition* of D is the multipartition of D each of whose elements ϖ_d is the ones partition of d. We think of a multipartition as a multiset, via the forgetful map MultiPart(D) \rightarrow Part($\sum_{d\in D} d$) sending $(\varpi_d)_{d\in D}$ to $\bigcup_{d\in D} \varpi_d$. For example, we write $\eta \in (\varpi_d)_{d\in D}$ to mean that η is a part of some ϖ_d .

If ϖ is a partition, we write S_{ϖ} for the group of automorphisms of the partition; e.g. for the partition $\varpi = \{1, 1, 1, 2, 2\}$ of 7, we have $S_{\varpi} \cong S_3 \times S_2$. For $\sigma = (\varpi_d)_{d \in D}$ a multipartition of D, we define $S_{\sigma} := \prod_{d \in D} S_{\varpi}$.

Let A be a set, and let $\sigma = (\varpi_d)_{d \in D}$ be a multipartition of D. An A-labeling L of σ is an assignment $\{L_\eta\}_{\eta \in \sigma}$ of an element of A to each part η of each ϖ_d . Precisely, it is the data of a tuple $\tilde{\sigma} = (\tilde{\varpi}_d)_{d \in D}$, where $\tilde{\varpi}_d$ is a multiset of pairs (η, a) with $\eta \in \mathbb{Z}_{>0}$ and $a \in A$, such that the tuple $\sigma = (\varpi_d)_{d \in D}$ obtained by forgetting the second entry of each pair in each $\tilde{\varpi}_d$ is equal to σ . We define $S_{\sigma,L}$ to be the subgroup of S_{σ} of permutations that preserve labels.

1.7.2 Graphs

Let Γ be a graph. We write $V(\Gamma)$ for the set of vertices and $E(\Gamma)$ for the set of edges. A *flag* of Γ is a pair (v, e) with e incident to v; we write $F(\Gamma)$ for the set of flags of Γ . We write $E(\Gamma)_v$ for the set of edges incident to v, and $E(\Gamma)_{v_1,v_2}$ for the set of edges connecting v_1 to v_2 .

A map of graphs $\phi: \Gamma \to \Gamma'$ is a pair of maps $\phi_V: V(\Gamma) \to V(\Gamma')$ and $\phi_E: E(\Gamma) \to E(\Gamma')$,

such that $\phi_E(E(\Gamma)_{v_1,v_2}) \subseteq E(\Gamma')_{\phi_V(v_1),\phi_V(v_2)}$.

CHAPTER 2

Orbifold Gromov-Witten theory

This chapter is intended as an introduction to the parts of orbifold Gromov-Witten theory that we require. Since this material can be found elsewhere [54, 46, 36, 2], we concentrate on an intuitive elementary discussion of the behaviors of orbifolds/stacks, as compared to varieties. Of course, this material can also be found elsewhere [4].

2.1 Orbifolds and orbifold maps

Question 2.1.1. What does the moduli space \mathcal{M} of unordered pairs of real numbers look like?

By calling \mathcal{M} a moduli space, one thing we mean is that any point of \mathcal{M} corresponds to an unordered pair of real numbers, and that for any unordered pair of real numbers, there is a corresponding point of \mathcal{M} . We may visualize \mathcal{M} as follows.

For any unordered pair $\{a, b\}$ of real numbers, we may uniquely put the pair in (weakly) ascending order. In other words, there is a bijective map P from \mathcal{M} to the half-plane $\{(x, y) \in \mathbb{R}^2 : y \geq x\}$.

Is "manifold with boundary" the correct geometric structure on \mathcal{M} ? We claim that \mathcal{M} comes with extra structure along the line y = x. Intuitively, this line behaves as a mirror. To see this, we must consider not only single unordered pairs (points of \mathcal{M}), but also 1-dimensional *families* of unordered pairs (paths in \mathcal{M}).

Consider the family $\{t, 3 - 2t\}$ of unordered pairs. Via the map P, this defines a path in the half-plane $\{y \ge x\}$: here are some values.

$$t = 0 \quad (0,3) t = 1 \quad (1,1) t = 2 \quad (-1,2)$$

Between t = 0 and t = 2, here is the path:



This is what we meant by mirror behavior. An equivalent way of saying this is the following. Rather than choosing an ordering, we could simply keep track of both orderings. For a pair $\{a, b\}$, we obtain two (unordered) points of \mathbb{R}^2 , namely (a, b) and (b, a). These points are permuted by an action of S_2 on \mathbb{R}^2 , where the nontrivial element σ of S_2 acts by reflecting over the line x = y. In other words, \mathcal{M} is in bijection with the *space of orbits* \mathbb{R}^2/S_2 .

As a topological space, \mathbb{R}^2/S_2 is isomorphic to the half-plane $y \ge x$. However, \mathbb{R}^2/S_2 has extra structure, giving way to the "mirror" behavior above. In particular, there is a notion of a smooth curve on \mathbb{R}^2/S_2 ; given a smooth curve in \mathbb{R}^2 , we just take its image under the quotient map. The path above is smooth, since it is the image of a line in \mathbb{R}^2 .

In general, "well-behaved" moduli spaces behave like the quotient space of a manifold V by a group G acting by automorphisms — at least locally. An *orbifold* is the correct geometric object to capture this behavior. An orbifold has a cover by charts U whose data is a manifold together with a group action, and the transition functions must respect the group actions. (See [4], Section 1.1 for the precise definition.) As an orbifold, \mathcal{M} "remembers" the group S_2 , in that any orbifold chart containing a point on the line y = x can be identified with a small neighborhood of that point in \mathbb{R}^2 , together with the action of S_2 fixing that point. We say that S_2 is the *isotropy group* at that point. In general, for a point x of an orbifold X, we write G_x for the isotropy group at x. If G_x is nontrivial we call x an *orbifold point*.¹

The simplest orbifolds are quotients of a manifold by a finite group; these are called *global* quotients. We write X = [V/G] for such an orbifold, where V is the manifold and G the group. The main topic of this thesis, the example $\text{Sym}^d \mathbb{P}^r := [(\mathbb{CP}^r)^d/S_d]$, is one of these. Also, any manifold is a global quotient orbifold, taking G to be trivial. In Section 2.2 we will see orbifolds that are not global quotients.

¹Since G_x depends upon a choice of orbifold chart, it is only defined up to isomorphism.

Example 2.1.2. Consider $[\mathbb{R}^2/(\mathbb{Z}/n/\mathbb{Z})]$, where $a \in \mathbb{Z}/n\mathbb{Z}$ acts by rotation on \mathbb{R}^2 by $2\pi a/n$. The result is in bijection with a half-open sector in \mathbb{R}^2 , with the edges identified. (As a topological space, this isomorphic to \mathbb{R}^2 , but not as an orbifold.) We may picture $[\mathbb{R}^2/(\mathbb{Z}/n\mathbb{Z})]$ as a cone, with cone angle $2\pi/n$, and isotropy group $\mathbb{Z}/n\mathbb{Z}$ at the cone point:



Note that (unlike in some literature) the group G need not act faithfully. (For a general orbifold, this translates to G_x being nontrivial for every point x.²) As an extreme example, one of the orbifolds we will often need to consider is BG, the quotient of a point by the trivial action of the group G. Another example is $X \times BG$ for any orbifold X and finite group G; this is the same as the quotient [X/G] by the trivial action of G.³ These orbifolds are sometimes called *ineffective orbifolds*, or G-gerbes, where G is the isomorphism class of the generic point of X.

Definition 2.1.3. A structure that can be defined on a manifold can usually be defined on an orbifold. We can make sense of the notion of complex (or Riemannian or symplectic) structures, vector bundles/sheaves, tangent spaces, maps of orbifolds, covering maps, fundamental groups, homology and cohomology of orbifolds, etc. For global quotient orbifolds, these all simplify:

- A complex structure on [V/G] is a complex structure on V such that G acts by biholomorphisms.
- A vector bundle E on [V/G] is a G-equivariant vector bundle \tilde{E} on V. (Note: For $v \in V$, G_v acts on the fiber \tilde{E}_v . This is called the *monodromy action*.)
- The tangent space at x ∈ [V/G] is the disjoint union of the tangent spaces at all points of the corresponding G-orbit in V, modulo the derivative of the action of G. This has the structure of a vector space modulo a group acting by linear automorphisms, or equivalently a representation (of G_x).
- A map of orbifolds $f : [V/G] \to [V'/G']$ is a map⁴ $\tilde{f} : V \to V'$, together with a map $\phi : G \to G'$, such that $\tilde{f}(g \cdot v) = \phi(g) \cdot \tilde{f}(v)$ for all $v \in V$ and $g \in G$. (Two maps (\tilde{f}_1, ϕ_1)

²Here and elsewhere, we assume X is connected.

³It is straightforward to define the quotient of an *orbifold* by a finite group; the result is another orbifold.

⁴There is a subtle technical detail here. It may be possible to write $[V_1/G_1] \cong [V_2/G_2]$, but the set of maps $[V_1/G_1] \rightarrow [V'/G']$ is not in bijection with the set of maps $[V_2/G_2] \rightarrow [V'/G']$ (in our definition). This is fixed by taking a "common refinement", i.e. a presentation $[V_0/G_0]$ where G_0 contains G_1 and G_2 . In general, one defines orbifold maps using *Morita equivalences*, see [4].

and (\tilde{f}_2, ϕ_2) are identified if there exists $h \in G'$ such that $\tilde{f}_1(v) = h \cdot \tilde{f}_2(v)$ for all $v \in V$; in this case we have

$$\tilde{f}_2(g \cdot v) = h^{-1}\tilde{f}_1(g \cdot v) = h^{-1}\phi_1(g)\tilde{f}_1(v) = h^{-1}\phi_1(g)h \cdot \tilde{f}_2(v),$$

i.e. ϕ_2 is the postcomposition of ϕ_1 with conjugation by h.) The map $\phi: G \to G'$, which is well-defined up to postcomposition by conjugation, is called the *monodromy map*.

- A covering map f : [V/G] → [V'/G'] of orbifolds is a map such that f̃ : V → V' is a covering map. In particular, the natural quotient map V → [V/G] (of orbifolds) given by f̃ = id is a covering map for any V and G.
- The universal cover \tilde{V} of V is also the universal cover of [V/G], and the (orbifold) fundamental group of [V/G] is the group of deck transformations of \tilde{V} over [V/G].
- The homology and cohomology of [V/G] are canonically isomorphic to the G-equivariant homology and G-equivariant cohomology of V.

The following fact is helpful for computing intersection products: if x, x' are points of a (connected) orbifold X, then $|G_x|[x] = |G_{x'}|[x'] \in H_0(X, \mathbb{Q})$. In particular, [x] is $\frac{1}{|G_x|}$ times the class of a nonorbifold point of X.

There is an important construction associated to any orbifold X, called its *inertia orbifold* IX. (See Example 2.5 in [4].)

Definition 2.1.4. Let X = [V/G] be a global quotient orbifold. The inertia orbifold I[V/G] is the quotient of $\{(v, g) \in V \times G | g \cdot v = v\}$ by G, where $h \in G$ acts by $h \cdot (v, g) = (h \cdot v, hgh^{-1})$.

For a general orbifold X, this construction glues to form an orbifold IX. As a set, IX consists of pairs (x, (g)) with $x \in X$ and (g) a conjugacy class in G_x .

Example 2.1.5. Let X = BG, for G a finite group. Then IX is the quotient of G (as a set of points) by itself, acting by conjugation. Points of IX are in bijection with conjugacy classes (g) of G, and the isotropy group of a point (g) is $C_G(g)$. There is a distinguished conjugacy class (e), which corresponds to a point of IX with isotropy group BG (since $C_G(e) = G$). For example, if $G = S_3$, then IX is the disjoint union of three orbifold points $(e) \cong BS_3$, $(tr) \cong B(\mathbb{Z}/2\mathbb{Z})$, and $(cy) \cong B(\mathbb{Z}/3\mathbb{Z})$.

Example 2.1.6. Let $X = [\mathbb{R}^2/S_2]$ as above. Then IX is the quotient of $\mathbb{R}^2 \times \{e\} \cup \Delta \times \{\sigma\} \subseteq \mathbb{R}^2 \times S_2$ by S_2 , where Δ is the diagonal $\{x = y\}$, and $\sigma \in S_2$ is the transposition. Thus, IX is the union of a copy of X with the quotient $[\Delta/S_2]$, where S_2 acts trivially.

Definition 2.1.7. In general, IX has a copy of X as a connected component, corresponding to pairs $(v, e) \in V \times G$. (These automatically glue in the case where X is not a global quotient.) This connected component is called the *untwisted sector*, and other connected components are called *twisted sectors*.

2.2 Orbifold curves

Besides $\operatorname{Sym}^d \mathbb{P}^r$, the other orbifolds we will use most often are (complex) *orbifold curves*, or *orbifold Riemann surfaces*. These are similar to Riemann surfaces, except that instead of being locally isomorphic to \mathbb{C} with biholomorphic transition functions, they are locally isomorphic to $[\mathbb{C}/\mu_n]$ (with its complex structure) for some n, where μ_n is the group of nth roots of unity acting by multiplication. (Again, the biholomorphic transition functions have to be compatible with the group actions on charts.) The point $0 \in [\mathbb{C}/\mu_n]$ has isotropy group μ_n , and all other points in the chart have trivial isotropy group; this local behavior also appeared in Example 2.1.2.

There is a natural map $[\mathbb{C}/\mu_n] \to \mathbb{C}$ that sends the orbit $\{a, \zeta_n a, \zeta_n^2 a, \ldots, \zeta_n^{n-1} a\}$ to a^n , which implies that there is a natural holomorphic map from an orbifold Riemann surface X to a Riemann surface <u>X</u>. (This is a special case of the "coarse moduli space" map mentioned in the introduction; see Section 2.3.) We may think of X as being the data of <u>X</u> endowed with "extra structure" at finitely many points P, namely the data of the size r_P of the isotropy group at P.

Remark 2.2.1. There is a close relationship between covering maps of orbifold curves and ramified covers of (nonorbifold) curves. Let $f : C \to D$ be a degree d ramified cover of curves, and let $y \in D$. The ramification profile of f at y is a partition $\varpi_f \in Part(d)$, and for each part η of ϖ_f , there is a point x_η of C mapping to y with local degree η .

We modify C and D. We replace a small neighborhood U of y with a copy of $[\mathbb{C}/\mu_N]$, where $N = \operatorname{lcm}(\varpi_f)$ is the least common multiple $\operatorname{lcm}(\varpi_f)$ of all parts $\eta \in \varpi_f$. For each $x_\eta \in C$, we replace the neighborhood $f^{-1}(U)$ with a copy of $[\mathbb{C}/\mu_{N/\eta}]$, mapping down to $[\mathbb{C}/\mu_N] \subseteq D$ by the pair (\tilde{f}, ϕ) , with $\tilde{f}(z) = z$ and $\phi(\zeta) = \zeta^{-1}$.

After doing this operation for all ramification points of D, we are left with a canonical covering map of orbifold curves (in the sense of Definition 2.1.3, at least locally). The original map is recovered as the induced map on coarse moduli spaces.

Some orbifold curves are global quotient orbifolds. For example, there is a curve, constructed below, whose coarse moduli space is isomorphic to \mathbb{P}^1 , with whose only orbifold points are over 0 and ∞ , both with isotropy group isomorphic to μ_n . This orbifold curve is the quotient of \mathbb{P}^1 by μ_n , where μ_n acts by scaling on \mathbb{P}^1 . However, if we had introduced only one orbifold point, the result would not be a global quotient. (If it were, then Remark 2.2.1 would assign to the nontrivial covering map $V \to [V/G]$ a ramified cover of \mathbb{P}^1 , ramified at a single point. No such ramified cover exists.) Similarly, if the isotropy groups at the two orbifold points had different sizes, then the curve would not be a global quotient.

These curves are constructed as follows. The complement $[\mathbb{C}^*/\mu_n]$ of the cone point in $[\mathbb{C}/\mu_n]$ is isomorphic to \mathbb{C}^* , via the diagram



with $\tilde{f} = z \mapsto z^n$ (and $\phi : \mu_n \to 1$ trivial). Therefore we may glue $[\mathbb{C}^*/\mu_n]$ to $[\mathbb{C}^*/\mu_m]$ by identifying the respective copies of \mathbb{C}^* via the isomorphism $z \mapsto z^{-1}$.

2.3 Orbifolds and stacks

We will need to consider singular objects, so we use the algebraic analog of orbifolds: Deligne-Mumford stacks. For the remainder of the thesis, we use the terms "orbifold" and "smooth Deligne-Mumford stack" interchangeably. Working with stacks requires dealing with unfortunate technical difficulties. For example, morphisms of stacks do not behave like arrows in a category, but rather like 1-arrows in a 2-category.⁵ Luckily one can usually avoid these issues. For more details, see e.g. [21].

One way to define a (separated) Deligne-Mumford stack (over \mathbb{C}) is by mimicking the last section, using (étale) charts and transition functions, where each chart is the quotient of an affine scheme by a finite group (Lemma 2.2.3 of [3]). This definition is difficult to state precisely, and it is also more natural to define Deligne-Mumford stacks via maps from schemes, along the lines of the "functor of points" perspective. That is, a stack X is usually defined as a category \mathfrak{C}_X , with a functor $F_X : \mathfrak{C}_X \to \text{Sch}$ to the category of schemes, that satisfies certain properties. The objects η of \mathfrak{C}_X with $F_X(\eta) = S$ are thought of as "maps from S to X." Morphisms of stacks are functors that commute with F_X , and 2-morphisms are natural transformations. Any scheme is naturally a stack, and indeed, considering a scheme S as a stack, the category of 1-morphisms $f : S \to X$ is equivalent to the subcategory $F_X^{-1}(S)$ of \mathfrak{C}_X .

Example 2.3.1. For V a scheme and G a group acting on V by automorphisms, [V/G] is a Deligne-Mumford stack. (This is obvious using the first definition of a Deligne-Mumford stack. To use the

⁵This means e.g. that morphisms may themselves have automorphisms. In fact this already came up in Definition 2.1.3, when we identified certain morphisms via an action of G. One has to treat these identifications not as an equivalence relation, but as a collection of 2-arrows.

second definition, we need to define a map $S \to [V/G]$. Such a map is defined to be a principal G-bundle $\tilde{S} \to S$, together with a G-equivariant map $\tilde{S} \to V$.) Similarly any orbifold curve has the natural structure of a Deligne-Mumford stack.

As with orbifolds, properties of a scheme can usually be extended to Deligne-Mumford stacks [21]:

- A Deligne-Mumford stack is reduced/smooth if it has a covering of charts $[U_i/G_i]$, where U_i is a reduced/smooth scheme.
- A Deligne-Mumford stack is separated if its diagonal morphism is proper after base change to any scheme. (Part of the defition of a Deligne-Mumford stack is that this base change is a map of schemes.)
- A separated Deligne-Mumford stack is proper if it admits a surjective map from a proper scheme.

As mentioned in the introduction, an orbifold has an underlying variety. More precisely and generally, any separated Deligne-Mumford stack X admits a *coarse moduli space map* $X \to \underline{X}$, where \underline{X} is an algebraic space, and the map is bijective on \mathbb{C} -points and universal with respect to maps to algebraic spaces (see [38].) If $\{[U_i/G_i]\}_i$ is an atlas for X with $U_i = \text{Spec}(A_i)$, then \underline{X} has an étale atlas consisting of the charts $\{U_i/G_i = \text{Spec}(A_i^G)\}$. This allows us to think of a separated Deligne-Mumford as an algebraic space, endowed with some extra structure, in the spirit of the first example (Question 2.1.1).

Example 2.3.2. It will be important later that even if X is smooth, <u>X</u> is usually singular. (Exceptions include orbifold curves, or more generally Deligne-Mumford stacks whose only isotropy is cyclic and along a divisor.) Let $X = [\mathbb{C}^2/(\mathbb{Z}/2\mathbb{Z})]$, where $1 \in \mathbb{Z}/2\mathbb{Z}$ acts by $1 \cdot (x, y) = (-x, -y)$. Then X is a smooth Deligne-Mumford stack by definition. By the description of coarse moduli spaces on charts, we have

$$\underline{X} \cong \operatorname{Spec}(\mathbb{C}[x, y]^{\mathbb{Z}/2\mathbb{Z}}) \cong \operatorname{Spec}(\mathbb{C}[x^2, xy, y^2]) \cong \operatorname{Spec}(\mathbb{C}[a, b, c]/(b^2 - ac)).$$

The latter is a quadric cone in \mathbb{C}^3 , a singular variety.

A map $X \to Y$ of Deligne-Mumford stacks is *representable* if for every map $S \to Y$ from a scheme S, the base change $X \times_Y S$ is a scheme. Equivalently, the map of isotropy groups at every point is injective.

Finally, we see a few more examples.

Example 2.3.3. Orbifold curves are exactly the smooth, proper, connected, 1-dimensional Deligne-Mumford stacks.

Example 2.3.4. A balanced twisted curve or prestable orbifold curve C is a connected, proper, 1-dimensional Deligne-Mumford stack that is (étale) locally isomorphic to:

- 1. isomorphic to $[\mathbb{C}/\mu_n]$ for some *n*, where μ_n acts by multiplication, or
- 2. isomorphic to $[\mathbb{V}(xy)/\mu_n]$, where $\mathbb{V}(xy) \subseteq \mathbb{C}^2$ is the union of the coordinate axes, and μ_n acts by multiplication by opposite roots of unity on x and y.

We often consider these curves with labeled marked points b_1, \ldots, b_n , where every smooth orbifold point is marked, and no nodes are marked. If every rational component has at least three nodes or markings (collectively known as special points), then we say C is *stable*. Prestable orbifold curves are the building blocks of orbifold stable maps.

Remark 2.3.5. One can generalize Remark 2.2.1 to covers of prestable orbifold curves, as follows. One may check that any connected étale cover of a node $[\mathbb{V}(xy)/\mu_n]$ from Example 2.3.4 is of the form $[\mathbb{V}(xy)/\mu_n] \to [\mathbb{V}(xy)/\mu_n]$ for some *a*, where the map is the quotient by the action of μ_a by multiplication by opposite roots of unity. The induced map on coarse moduli spaces is the map $\mathbb{V}(xy) \to \mathbb{V}(xy)$ given by $(x, y) \mapsto (x^a, y^a)$. This is the local (*balanced*) structure of an *admissible cover* in the sense of [30]. (An admissible cover is a map of nodal curves whose restriction to the smooth locus of the source is a ramified cover, and whose restriction to an étale neighborhood of a node is that just described.)

Indeed, given an étale cover of a prestable orbifold curve, the coarse moduli map is an admissible cover, by Lemma 4.2.1 of [1]. However, [1] shows that unlike in Remark 2.2.1, the process of taking the coarse moduli spaces is not quite reversible. In particular, the induced map from the moduli stack of étale covers of prestable orbifold curves to the moduli stack of admissible covers is the normalization map. (The moduli stack of admissible covers is in general singular, whereas the moduli stack of étale covers of prestable orbifold curves is smooth.)

Example 2.3.6. Consider the quotient $X = [V(xy, yz, xz)/(\mathbb{Z}/3\mathbb{Z})]$, where V(xy, yz, xz) is the union of the coordinate axes in \mathbb{CP}^3 , and μ_3 acts by cycling the coordinates x, y and z. Then X is birational to \mathbb{P}^1 , and has a single singular point whose tangent space has dimension 3. One can check that the coarse moduli space of X is isomorphic to \mathbb{P}^1 .

Similarly, we could replace μ_3 by S_3 , acting on the coordinates by permutations. The result X' is an S_2 -gerbe over X. (Incidentally, the normalization of X' is the "nonbanded" gerbe $[\mathbb{P}^1/S_3]$ in Example II.9 of [36].)

Example 2.3.7. Consider the quotient of the scheme $V = \text{Spec}(\mathbb{C}[\epsilon]/(\epsilon^n))$ by the group $G = \mu_2$, acting by multiplication on ϵ . This is a nasty nonreduced Deligne-Mumford stack, that might arise as a substack of a smooth stack, based at a point with isotropy group μ_2 . We can calculate its coarse moduli space:

$$[V/G] \cong \operatorname{Spec}((\mathbb{C}[\epsilon^2]/(\epsilon^n))^{\mu_2}) \cong \operatorname{Spec}\mathbb{C}[\delta]/(\delta^{\lceil n/2 \rceil}).$$

2.4 Moduli spaces of stable maps

The fundamental objects of Gromov-Witten theory are the moduli spaces of stable maps $\overline{\mathcal{M}}_{g,n}(X,\beta)$, developed by Kontsevich and Manin. Fix X a smooth projective variety, nonnegative integers g and n, and a class $\beta \in H_2(X,\mathbb{Z})$. The moduli space $\overline{\mathcal{M}}_{g,n}(X,\beta)$ parametrizes the data of:

- A genus g connected curve C with at worst nodal singularities,
- A collection of n distinct labeled marked smooth points b_1, \ldots, b_n on C, and
- A map $f: C \to X$ with $f_*[C] = \beta$,

subject to the condition that if $C' \subseteq C$ is a rational irreducible component, and $f|_{C'}$ is a constant map, then C' is *stable*, i.e

(# nodes on
$$C'$$
) + (# marked points on C') ≥ 3 .

By work of Kontsevich [41], and Fulton-Pandharipande [26], $\overline{\mathcal{M}}_{g,n}(X,\beta)$ is a proper and separated Deligne-Mumford stack. (It may be viewed as a "compactification" of the space $\mathcal{M}_{g,n}(X,\beta)$ of smooth genus g curves in β , but in reality may have extra irreducible or connected components, possibly of large dimensions; see Example 2.4.1.)

There is a natural universal curve C over $\overline{\mathcal{M}}_{g,n}(X,\beta)$, with sections s_1, \ldots, s_n , and a universal map f:

$$C \xrightarrow{f} X$$

$$\overset{s_i}{\leftarrow} \downarrow_{\pi}$$

$$\overline{\mathcal{M}}_{g,n}(X,\beta)$$

There are "evaluation maps" $ev_i = f \circ \sigma_i : \overline{\mathcal{M}}_{g,n}(X,\beta) \to X.$

While $\overline{\mathcal{M}}_{g,n}(X,\beta)$ is singular, it has a structure called a *perfect obstruction theory* (see [6] and [42]), which gives rise to a *virtual fundamental class* $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\text{vir}} \in H_{\text{vd}}(\overline{\mathcal{M}}_{g,n}(X,\beta))$, where

vd =
$$(1 - g)(\dim X - 3) + n + \int_{\beta} c_1(TX).$$

There is a useful long exact sequence of sheaves on $\overline{\mathcal{M}}_{g,n}(X,\beta)$:

$$0 \to \operatorname{Aut}(C) \to R^0 \pi_* f^* T X \to \mathcal{T}^1 \to$$
$$\to \operatorname{Def}(C) \to R^1 \pi_* f^* T X \to \mathcal{T}^2 \to 0,$$

where $\operatorname{Aut}(C)$ and $\operatorname{Def}(C)$ parametrize infinitesimal automorphisms and deformations of the marked curve (C, b_1, \ldots, b_n) , $R^0 \pi_* f^* TX$ and $R^1 \pi_* f^* TX$ parametrize infinitesimal deformations and obstructions of the map $f: C \to X$, and \mathcal{T}^1 and \mathcal{T}^2 are the "tangent sheaf" and "obstruction sheaf" of $\overline{\mathcal{M}}_{g,n}(X,\beta)$. One reason this is useful is that in the special case when \mathcal{T}^1 and \mathcal{T}^2 are vector bundles, we have $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\operatorname{vir}} = e(\mathcal{T}^2)$ (using Poincaré duality).

Example 2.4.1. We describe $\overline{\mathcal{M}}_{0,0}(\mathcal{Q}, 2)$, where $\mathcal{Q} \subseteq \mathbb{P}^4$ is a general quintic threefold. It is a classical fact that \mathcal{Q} contains exactly 2875 nonintersecting lines (see also Example 2.6.5 below), and it was shown in [37] that \mathcal{Q} contains exactly 609250 smooth conic curves. This implies that there are three types of genus zero, degree 2, unmarked stable maps to \mathcal{Q} :

- 1. Maps $f: \mathbb{P}^1 \to \mathscr{Q}$ that map isomorphically to a conic in \mathscr{Q} (there are 609250 of these),
- 2. Maps $f : \mathbb{P}^1 \to \mathcal{Q}$ that map to a line in \mathcal{Q} as a degree 2 cover, ramified over two distinct points, and
- 3. Maps $f: C \to \mathcal{Q}$, where C consists of two copies of \mathbb{P}^1 joined at a node, and f maps each \mathbb{P}^1 with degree 1 to a line in \mathcal{Q} .

One can check by explicit construction that stable maps of type 3 are precisely the limits of those of type 2 in families when the two ramification points collide. Thus $\overline{\mathcal{M}}_{0,0}(\mathcal{Q}, 2)$ consists of 609250 points (from maps of type 1), together with 2875 copies of $[\mathbb{P}^2/S_2]$, where S_2 acts trivially on \mathbb{P}^2 . Here an open subset (the complement of the discriminant conic) of each \mathbb{P}^2 parametrizes maps of type 2 up to automorphism, and the discriminant conic parametrizes maps of type 3 (which are uniquely defined by the image point of the node). The reason for the S_2 action is that each of these maps has a symmetry, namely switching the two branches of the source curve.

One can also compute the virtual fundamental class $[\overline{\mathcal{M}}_{0,0}(\mathcal{Q},2)]^{\text{vir}}$ (see [32], Chapter 27): it is the sum of the fundamental classes of the 609250 points, plus the sum of [pt]/8 over each copy of $[\mathbb{P}^2/S_2]$. (Here [pt] is the class of a nonorbifold point, i.e. the pushforward of the fundamental class along any map Spec $\mathbb{C} \to [\mathbb{P}^2/S_2]$.) The total degree of the class is 609250 + 2875/8; this number is the Gromov-Witten invariant $\langle \rangle_{0,0,2}^{\mathcal{Q}}$ defined in Section 2.5. (This invariant is calculated via the Theorem 1.2.4 in Example 2.6.5.)

Example 2.4.2. Another important case is $X = \mathbb{P}^r$. If g = 0, then the spaces $\overline{\mathcal{M}}_{0,n}(\mathbb{P}^r,\beta)$ are smooth irreducible Deligne-Mumford stacks, whose coarse moduli spaces are projective varieties.

There is a natural inclusion $j : \overline{\mathcal{M}}_{0,n}(\mathcal{Q},\beta) \hookrightarrow \overline{\mathcal{M}}_{0,n}(\mathbb{P}^r,\beta)$, and one can use j to compute intersection numbers on $\overline{\mathcal{M}}_{0,n}(\mathcal{Q},\beta)$. In particular, the cohomology of the ambient space $\overline{\mathcal{M}}_{0,n}(\mathbb{P}^r,\beta)$ is relatively well-understood, so after calculating $j_*[\overline{\mathcal{M}}_{0,n}(\mathcal{Q},\beta)]^{\text{vir}}$, one can compute any intersection product on $\overline{\mathcal{M}}_{0,n}(\mathcal{Q},\beta)$ of classes pulled back from $\overline{\mathcal{M}}_{0,n}(\mathbb{P}^r,\beta)$.

In general (especially in high genus), one should not expect $\overline{\mathcal{M}}_{g,n}(X,\beta)$ to be smooth, though it happened to be smooth in the previous two examples. (In fact, there is a precise sense in which it can be "arbitrarily singular," even for $X = \mathbb{P}^r$ [55].)

Definition 2.4.3. We define natural classes in $H^2(\overline{\mathcal{M}}_{g,n}(X,\beta),\mathbb{Q})$:

$$\psi_i := c_1(s_i^*(T_{C|\overline{\mathcal{M}}_{q,n}(X,\beta)}^*)),$$

where $T^*_{C|\overline{\mathcal{M}}_{q,n}(X,\beta)}$ is the relative cotangent bundle of π .

2.4.1 Orbifold stable maps

All of the above can be extended to the case where X is a smooth proper Deligne-Mumford stack, due to Abramovich-Graber-Vistoli [3, 2]. There are a few modifications:

- (i) The curve C is allowed to have orbifold structure at marked points and nodes: it must be a prestable orbifold curve as in Example 2.3.4.
- (ii) The map $f: C \to X$ is required to be representable.
- (iii) The maps ev_i take values in the inertia stack IX rather than in X. Intuitively, at a marked point $P \in C$, there is a canonical generator $g \in G_P$, and the monodromy map takes g to a well-defined conjugacy class in $G_{f(P)}$, namely a point of IX.
- (iv) Depending on the definition one uses, the maps s_i may not exist, due to the possibility that the marked points form a *nontrivial gerbe* over the moduli space. We adopt the convention as in Section 7.1 of [46], that marked gerbes come with the data of a trivialization.⁶ Because of this convention, the classes ψ_i may be defined as above.

Example 2.4.4. The case $X = BS_d$ (and more generally X = BG) is studied in detail [1]. Spaces of orbifold stable maps to BS_d are closely related to Hurwitz spaces and Harris-Mumford spaces of admissible covers [30]. An orbifold stable map $f : C \to BS_d$ is (by definition) a principal

⁶If we did not take this convention, then the maps ev_i would land in a *rigidification* of IX, i.e. a stack that looks the same as IX geometrically, but has smaller isotropy groups.

 S_d -bundle $\mathcal{P} \to C$. There is a correspondence between principal S_d -bundles and degree d étale covers, that sends

$$(\mathcal{P} \to C) \mapsto (\mathcal{P} \times_{S_d} \{1, \dots, d\} \to C).$$

Example 2.2.1 says that there is, in turn, a correspondence between these degree d étale maps $D \to C$ of orbifold covers, and degree d ramified covers $\underline{D} \to \underline{C}$ of nonorbifold curves. These two correspondences identify an open subset of $\overline{\mathcal{M}}_{g,n}(BS_d, 0)$ with a Hurwitz space parametrizing (possibly disconnected) degree d ramified covers of genus g nonorbifold curves, with arbitrary ramification profiles at marked points.

The monodromy map in Definition 2.1.3 at a marked point of C with isotropy group μ_n is a map $\mu_n \to S_d$, up to conjugacy. These maps are in bijection with conjugacy classes of S_d , which in turn are in bijection with partitions of d. Indeed, the partition of d associated to an orbifold marked point of C by the monodromy map is precisely the ramification profile at the corresponding nonorbifold marked point of \underline{C} .

2.5 Gromov-Witten invariants and twisted Gromov-Witten invariants

A Gromov-Witten invariant is an integral of the form

$$\langle \overline{\psi}_{1}^{a_{1}} \gamma_{1}, \dots, \overline{\psi}_{n}^{a_{n}} \gamma_{n} \rangle_{g,n,\beta}^{X} := \int_{[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathrm{vir}}} \prod_{j=1}^{n} \overline{\psi}_{j}^{a_{j}} \operatorname{ev}_{j}^{*} \gamma_{j} \in \mathbb{Q},$$
(2.1)

where

- $\overline{\psi}_j$ is the *j*th cotangent class on $\overline{\mathcal{M}}_{g,n}(X,\beta)$, coming from the cotangent space to the *coarse* moduli space of C,⁷ and
- the "insertions" γ_j are in the Chen-Ruan cohomology H^{*}_{CR}(X, Q) := H^{*}(IX, Q) (see item (iii) of Section 2.4.1).⁸

If X has an action of a torus T, it induces a natural T-action on IX and $\overline{\mathcal{M}}_{g,n}(X,\beta)$, and $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\text{vir}}, \overline{\psi}_j$, and $\mathrm{ev}_j^* \gamma_j$ are naturally equivariant classes (where $\gamma_j \in H^*_{CR,T}(X,\mathbb{Q})$). In this case (2.1) defines an *equivariant Gromov-Witten invariant* (an element of $H^*_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q})$, denoted by $\langle \cdots \rangle_{g,n,\beta}^{X,T}$) via T-equivariant integration.

⁷Note that $\overline{\psi}_j = r_j \psi_j$, where r_j is the size of the isotropy group at the mark b_j , and ψ_j is the usual ψ class.

⁸In enumerative terms, we may think of these insertions as not only specifying incidence conditions (as in Section 1.1), but also specifying the failure of a curve, locally in a chart $[V/G] \subseteq X$, to lift to V.

There is also a notion of a *twisted Gromov-Witten invariant*. Let E be a vector bundle on X, equipped with the linear action of a torus T (that acts trivially on X). Consider the class $R\pi_*f^*E \in K_T^0(\overline{\mathcal{M}}_{g,n}(X,\beta))$). This has a well-defined invertible Euler class $e_T(R\pi_*f^*E)$, see [33]. We define twisted Gromov-Witten invariants:

$$\langle \overline{\psi}_1^{a_1} \gamma_1, \dots, \overline{\psi}_n^{a_n} \gamma_n \rangle_{g,n,\beta}^{X,T,E} := \int_{[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathrm{vir}}} \prod_{j=1}^n \overline{\psi}_j^{a_j} \operatorname{ev}_j^* \gamma_j \cup e_T^{-1}(R\pi_*f^*E).$$

2.6 The Givental cone

For the rest of the thesis, we restrict to the case g = 0.

Notation 2.6.1. We write $H^*_{T,\text{loc}}(\text{Spec }\mathbb{C},\mathbb{Q})$ for the fraction field of $H^*_T(\text{Spec }\mathbb{C},\mathbb{Q})$, and

$$H^*_{T,\mathrm{loc}}(X,\mathbb{Q}) := H^*_T(X,\mathbb{Q}) \otimes_{H^*_T(\mathrm{Spec}\,\mathbb{C},\mathbb{Q})} H^*_{T,\mathrm{loc}}(\mathrm{Spec}\,\mathbb{C},\mathbb{Q}).$$

Definition 2.6.2. Following [18], the *T*-equivariant Novikov ring of X is

$$\Lambda_T^{\text{nov}} := H^*_{T,\text{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})[[\operatorname{NE}(X) \cap H_2(X, \mathbb{Z})]].$$

We write $\mathbf{Q} := \{Q_i\}_i$ for a set of generators of $NE(X) \cap H_2(X, \mathbb{Z})$, and use the shorthand

$$H^*_{T,\mathrm{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})[[\mathbf{Q}]] := H^*_{T,\mathrm{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})[[\operatorname{NE}(X) \cap H_2(X, \mathbb{Z})]].$$

When $X = \text{Sym}^d \mathbb{P}^r$, there is a single generator, denoted Q. (See Section 3.2.2.)

Definition 2.6.3. *Givental's symplectic*⁹ *vector space* is

$$\mathcal{H} := H^*_{CR,T,\mathrm{loc}}(X,\mathbb{Q})[[\mathbf{Q}]]((z^{-1})) = \mathcal{H}^+ \oplus \mathcal{H}_{-1}$$

where $\mathcal{H}^+ = H^*_{CR,T,\text{loc}}(X,\mathbb{Q})[[\mathbf{Q}]][z]$ and $\mathcal{H}^- = z^{-1}H^*_{CR,T,\text{loc}}(X,\mathbb{Q})[[\mathbf{Q}]][[z^{-1}]].$

Inside \mathcal{H} , there is a special subscheme $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$ called the *Givental cone* of $\text{Sym}^d \mathbb{P}^r$, which encodes the genus-zero Gromov-Witten invariants of $\text{Sym}^d \mathbb{P}^r$. (Precisely, $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$ is a formal germ of a subscheme, defined at $-1 \cdot z$, where $1 \in H^*_{CR,T,\text{loc}}(\text{Sym}^d \mathbb{P}^r, \mathbb{Q})$ is the fundamental class of the untwisted sector.) Fix a basis γ_{ϕ} of $H^*_{CR,T,\text{loc}}(\text{Sym}^d \mathbb{P}^r, \mathbb{Q})$, with Poincaré dual basis γ^{ϕ} .

⁹The symplectic structure is not relevant to us, so we do not define it.

Definition 2.6.4. A $\Lambda_{nov}^T[[x]]$ -valued point of $\mathcal{L}_{Sym^d \mathbb{P}^r}$ is defined to be a formal Laurent series

$$-1z + \mathbf{t}(z) + \sum_{n=0}^{\infty} \sum_{\beta=0}^{\infty} \sum_{\phi} \frac{Q^{\beta}}{n!} \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{\gamma_{\phi}}{-z - \overline{\psi}} \right\rangle_{0, n+1, \beta}^{\operatorname{Sym}^{d} \mathbb{P}^{r}, T} \gamma^{\phi} \in \mathcal{H}[[x]],$$

where $\mathbf{t}(z) \in (x, Q) \subseteq \mathcal{H}_+[[x]]$. (For brevity, we suppress the variables x and Q in t.)

 $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$ has several important geometric properties that follow from relations between Gromov-Witten invariants: see Appendix B of [16], which also defines $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$ rigorously as a non-Noetherian formal scheme. For example, it is a cone in a certain sense, hence the name (Proposition B.2 of [16]). It is also (formal) Lagrangian with respect to the symplectic structure on \mathcal{H} , so is often called the *Lagrangian cone* of X.

Example 2.6.5. We illustrate how the Theorem 1.2.4, the mirror theorem for quintic threefolds, can be used to recover Gromov-Witten invariants. In the discussion after Theorem 1.2.4, we saw that

$$I(t,Q,-z) = -z + \mathbf{t}_{1}(t,Q,z) + \sum_{\substack{n,\beta \ge 0\\\beta > 0 \text{ or } n \ge 2}} \sum_{i=0}^{3} \frac{Q^{\beta}}{n!} \langle \mathbf{t}_{1}(t,Q,\psi), \dots, \mathbf{t}_{1}(t,Q,\psi), \frac{H^{i}}{-z-\psi} \rangle_{0,n+1,\beta}^{\mathscr{Q}} \frac{H^{3-i}}{5} + O(Q^{2}),$$
(2.2)

where $\mathbf{t}_1(t, Q, z) = tH + Qe^t(-120z + (770 + 120t)H)$ and I(t, Q, z) is defined in (1.4). For simplicity, we consider (2.2) with t set to 0. Equating the coefficients of Q in (2.2) gives

$$-z \frac{(5H-z)(5H-2z)(5H-3z)(5H-4z)(5H-5z)}{(H-z)^5}$$

$$= -120z + 770H + \sum_{i=0}^{3} \langle \frac{H^i}{-z-\psi} \rangle_{0,1,1}^{\mathscr{Q}} \frac{H^{3-i}}{5}$$
(2.3)

By construction, the coefficients of 1 and H agree on both sides. The remaining terms on the left side of (2.3) are

$$-\frac{575H^2}{z} - \frac{1150H^3}{z^2}.$$
 (2.4)

The moduli space $\overline{\mathcal{M}}_{g,n}(\mathcal{Q},\beta)$ is *n*-dimensional. A dimension count implies that the coefficients of H^2 and H^3 on the right side of (2.3) are equal to:

$$\langle \frac{\psi}{z^2} \rangle_{0,1,1}^{\mathscr{Q}} \frac{H^3}{5} + \langle \frac{H}{-z} \rangle_{0,1,1}^{\mathscr{Q}} \frac{H^2}{5}.$$
(2.5)
Using the dilaton and divisor equations, (2.5) simplifies to

$$\frac{-2}{z^2} \langle \rangle^{\mathscr{Q}}_{0,0,1} \frac{H^3}{5} + \frac{1}{-z} \langle \rangle^{\mathscr{Q}}_{0,0,1} \frac{H^2}{5}.$$

Setting this equal to (2.4) gives $\langle \rangle_{0,0,1}^{\mathscr{Q}} = 2875$.

We also calculate $\langle \rangle_{0,0,2}^{\mathscr{Q}}$, since the computation will show that there is some recursive structure in the Gromov-Witten invariants of \mathscr{Q} . Again setting t = 0, we compute that the Q^2 -coefficient of $\mathbf{t}(0,Q,z)$ is $\mathbf{t}_2(0,Q,z) := -113400z + 810225H$. Equating the coefficients of $\frac{Q^2H^2}{-5z}$ in (2.2) gives

$$-z\frac{\prod_{\gamma=1}^{10}(5H-\gamma z)}{\prod_{\gamma=1}^{2}(H-\gamma z)^{5}} = \frac{1}{2}\langle -120\psi + 770H, -120\psi + 770H, H\rangle_{0,3,0}^{\mathscr{Q}} + \langle -120\psi + 770H, H\rangle_{0,2,1}^{\mathscr{Q}} + \langle H\rangle_{0,1,2}^{\mathscr{Q}}$$

Applying the dilaton and divisor equations yields:

$$\frac{21040875}{4} = 296450 \langle H, H, H \rangle_{0,3,0}^{\mathscr{Q}} + 120 \langle \rangle_{0,0,1}^{\mathscr{Q}} + 770 \langle \rangle_{0,0,1}^{\mathscr{Q}} + 2 \langle \rangle_{0,0,2}^{\mathscr{Q}}$$

Plugging in the known values $\langle H, H, H \rangle_{0,3,0}^{\mathscr{Q}} = 5$ (a translation of that fact that $H^3 = 5[pt] \in H^6(\mathscr{Q}, \mathbb{Q})$) and $\langle \rangle_{0,0,1}^{\mathscr{Q}} = 2875$ then gives:

$$\langle \rangle_{0,0,2}^{\mathscr{Q}} = \frac{4876875}{8} = 609250 + \frac{2875}{8},$$

as claimed in Example 2.4.1.

There is also a notion of a *twisted Givental cone* \mathcal{L}_X^E . As above, let E be a T-equivariant vector bundle over X, where T acts trivially on X. A $\Lambda_{nov}^T[[x]]$ -valued point of \mathcal{L}_X^E is defined to be

$$-1z + \mathbf{t}(z) + \sum_{n=0}^{\infty} \sum_{\beta=0}^{\infty} \sum_{\phi} \frac{Q^{\beta}}{n!} \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{\gamma_{\phi}}{-z - \overline{\psi}} \right\rangle_{0, n+1, \beta}^{X, T, E} \gamma^{\phi},$$
(2.6)

for some $\mathbf{t}(z) \in \langle x, Q \rangle \subseteq \mathcal{H}_+[[x]].$

Remark 2.6.6. We may allow $\mathbf{t}(z)$ to be a power series in z, because $\overline{\psi}$ is nilpotent (as the T-action on X is trivial). We will need this to make sense of Condition (III) of Theorem 5.1.4. Also, here γ_{ϕ} and γ^{ϕ} are dual bases of $H_T^*(X, \mathbb{Q})$ under the *twisted* Poincaré pairing, see [18].

CHAPTER 3

Properties of $\operatorname{Sym}^d \mathbb{P}^r$

This chapter introduces symmetric product stacks/orbifolds $\operatorname{Sym}^d X$ and discusses geometric properties. Specifically, we describe the tangent bundle of $\operatorname{Sym}^d X$ (Lemma 3.1.1), and compute the Chen-Ruan cohomology of $\operatorname{Sym}^d X$ with rational coefficients (Section 3.1.3). We make this computation more explicit in the case $X = \mathbb{P}^r$ (Section 3.2.2).

3.1 Symmetric product stacks

3.1.1 Two definitions of $\operatorname{Sym}^d X$

Let X be a scheme. The dth power X^d has a natural action of S_d . The dth symmetric product $\operatorname{Sym}^d X$ is the orbifold $[X^d/S_d]$. That is, for S a scheme, an object $f: S \to \operatorname{Sym}^d X$ of $\operatorname{Sym}^d X$ over S is a principal S_d -bundle \tilde{S} over S, together with an S_d -equivariant map $\tilde{f}: \tilde{S} \to X^d$. A morphism $(f: S \to \operatorname{Sym}^d X) \to (g: T \to \operatorname{Sym}^d X)$ over $S \to T$ is a diagram

such that the square is cartesian and the triangle commutes. If X is smooth, then so is $Sym^d X$, since it has an étale cover by the smooth scheme X^d .

There is an equivalent characterization of $\operatorname{Sym}^d X$, which will allow us to sidestep some of the complications of working with stacks. We define a stack $\widetilde{\operatorname{Sym}}^d X$ that is naturally isomorphic to $\operatorname{Sym}^d X$. Roughly, rather than parametrizing "d ordered points of X up to reordering", $\widetilde{\operatorname{Sym}}^d X$ will parametrize "Maps $P : d(\bullet) \to X$," where $d(\bullet) = \bigcup_{j=1}^d \operatorname{Spec} \mathbb{C}$. Precisely, an object $f: S \to \widetilde{\operatorname{Sym}}^d X$ over S is an étale map $\rho: S' \to S$ of degree d (i.e. a bundle with fiber $d(\bullet)$), together with a map $f': S' \to X$. A morphism $(f: S \to \widetilde{\operatorname{Sym}}^d X) \to (g: T \to \widetilde{\operatorname{Sym}}^d X)$ over $S \to T$ is a diagram

$$S' \xrightarrow{f'} X$$

$$\downarrow \qquad \downarrow \qquad \downarrow$$

$$S \longrightarrow T$$

There is a map $\operatorname{Sym}^d X \to \widetilde{\operatorname{Sym}}^d X$ that sends:

$$\left(\begin{array}{cc} \tilde{S} \xrightarrow{\tilde{f}} X^d \\ \downarrow \\ S \end{array}\right) \mapsto \left(\begin{array}{cc} \tilde{S} \times_{S_d} \{1, \dots, d\} \longrightarrow X \\ \downarrow \\ S \end{array}\right),$$

where the S_d -action on $\{1, \ldots, d\}$ is the obvious one, and the map $\tilde{S} \times_{S_d} \{1, \ldots, d\} \to X$ sends $(\tilde{s}, i) \mapsto \operatorname{pr}_i \circ \tilde{f}(s)$, where pr_i denotes the *i*th projection $X^d \to X$. It is easy to check that this is S_d -equivariant, and defines a morphism of stacks.

In the other direction, we may send:

$$\left(\begin{array}{c}S' \xrightarrow{f'} X\\ \downarrow\\S\end{array}\right) \mapsto \left(\begin{array}{c}\mathcal{I}som_S(S', \{1, \dots, d\}) \longrightarrow X^d\\ \downarrow\\S\end{array}\right),$$

where $\mathcal{I}som_S(S', \{1, \ldots, d\})$ is the principal S_d -bundle given on small (étale) open sets $U \to S$ by the set of isomorphisms $S' \times_S U \to \{1, \ldots, d\} \times U$. Given such an isomorphism, f' determines a U-valued point of X^d . It is again straightforward to check that this defines a map of stacks $\widetilde{\operatorname{Sym}}^d X \to \operatorname{Sym}^d X$, and that it is an inverse to the previous map. For the rest of the thesis we will use the descriptions interchangeably and denote them both by $\operatorname{Sym}^d X$. It is useful to keep in mind the following diagram, where the cube is Cartesian and the left and right faces consist of étale maps:



(The composition $S' \to X^d \times_{S_d} \{1, \ldots, d\} \xrightarrow{P} X$ is f'.)

3.1.2 The tangent bundle to $\operatorname{Sym}^d X$

Now we assume X is a smooth scheme. The two definitions of $\operatorname{Sym}^d X$ in Section 3.1.1 give two descriptions of the tangent bundle $T \operatorname{Sym}^d X$. First, since $\operatorname{Sym}^d X$ is isomorphic to $[X^d/S_d]$, $T \operatorname{Sym}^d X$ is the vector bundle on $\operatorname{Sym}^d X$ corresponding to the S_d -equivariant vector bundle $T(X^d)$ on X^d , where S_d acts by the derivative. Consider the portion of Figure (3.1), with more maps named:



Lemma 3.1.1. There is a natural isomorphism $T \operatorname{Sym}^d X \cong \rho_*(P^*TX)$.

Proof. Since the square is cartesian and the maps are étale, we have

$$\operatorname{pr}^*(\rho_*(P^*TX)) \cong \tilde{\rho}_*((\operatorname{pr}')^*(P^*TX)) = \tilde{\rho}_*((\operatorname{pr}' \circ P^*TX)).$$

Recall that $pr' \circ P$ is simply the "universal coordinate map," so since $\tilde{\rho}$ is a trivial étale cover, there is a canonical isomorphism

$$\tilde{\rho}_*((\mathrm{pr}' \circ P^*TX)) \cong \bigoplus_{\ell=1}^d P_\ell^*TX \cong T(X^d).$$

Since $\tilde{\rho}$ is S_d -equivariant, there is an induced S_d -action on $T(X^d)$ which agrees with the usual one. Thus the isomorphism descends to give $\rho_*(P^*TX) \cong T \operatorname{Sym}^d X$.

3.1.3 The inertia stack and Chen-Ruan cohomology of $Sym^d X$

Again, let X be a smooth variety. We describe the inertia stack of $Sym^d X$, see Definition 2.1.4. (See also Section 3 of [2]).

To a map $d(\bullet) \xrightarrow{P} X$, we may assign a partition $\sigma \in Part(d)$, where parts correspond to points of Im(P). This gives a stratification of $Sym^d X$ into locally closed strata $(Sym^d X)_{\sigma}$ indexed by $\sigma \in Part(d)$. **Proposition 3.1.2.** For each $\sigma \in Part(d)$,

$$(\operatorname{Sym}^{d} X)_{\sigma} \cong \left(\prod_{i \ge 1} \operatorname{Sym}^{\operatorname{Mult}(\sigma, i)} X\right) \smallsetminus \Delta$$

up to generic stabilizer, where Δ is the union of all diagonals. Each point of $(\text{Sym}^d X)_{\sigma}$ has isotropy group isomorphic to $\prod_{n \in \sigma} S_{\sigma}$.

Proof. On $(\text{Sym}^d X)_{\sigma}$, there is a natural operation of contracting fibers of each map $S' \xrightarrow{P} X$, and remembering the number of points in each contracted fiber. This induces a well-defined map

$$(\operatorname{Sym}^{d} X)_{\sigma} \to \left(\prod_{i \ge 1} \operatorname{Sym}^{\operatorname{Mult}(\sigma,i)} X\right) \smallsetminus \Delta.$$

The "obvious" scheme-theoretic inverse fails to be a functor, but it is easy to check that the map above is a rigidification in the sense of [1]. The isotropy group at a point $d(\bullet) \xrightarrow{P} X$ is simply the group of automorphisms of $d(\bullet)$ commuting with P, which by definition of $(\text{Sym}^d X)_{\sigma}$ is $\prod_{n \in \sigma} S_{\sigma}$.

It follows from Proposition 3.1.2 that components of $I \operatorname{Sym}^d X$ are indexed by $\operatorname{Part}(d)$. We denote the component associated to $\sigma \in \operatorname{Part}(d)$ by $(I \operatorname{Sym}^d X)_{\sigma}$. We have, up to generic stabilizer,

$$I(\operatorname{Sym}^{d} X)_{\sigma} \cong \overline{(\operatorname{Sym}^{d} X)_{\sigma}} \cong \prod_{i \ge 1} \operatorname{Sym}^{\operatorname{Mult}(\sigma,i)} X_{i}$$

From Example 2.5 of [4], the generic stabilizer is isomorphic to the centralizer of (any representative of) σ in S_d . Explicitly, this is the group (see Section 1.7):

$$C(\sigma) = S_{\sigma} \ltimes \prod_{\eta \in \sigma} \mu_{\eta}.$$
(3.2)

The (nonorbifold) cohomology with rational coefficients of each component $I(\text{Sym}^d X)_{\sigma}$ may be computed explicitly from that of X, using the fact that

$$H^*(\operatorname{Sym}^d X, \mathbb{Q}) \cong H^*(X^d, \mathbb{Q})^{S_d} \cong (H^*(X, \mathbb{Q})^{\otimes d})^{S_d}.$$

In particular, there are isomorphisms:

$$H^2(\operatorname{Sym}^d X, \mathbb{Q}) \cong (H^2(X, \mathbb{Q})^{\oplus d})^{S_d} \cong H^2(X, \mathbb{Q}).$$

3.2 Symmetric products of projective space

In this section we expand on some of the above in the case $X = \mathbb{P}^r$, and discuss the natural torus action on Sym^d \mathbb{P}^r . First, we introduce some notation to be used throughout.

3.2.1 Notation for projective space

We denote a point of $\mathbb{P}^r = \mathbb{P}(\mathbb{C}^{r+1})$ by $[x_0 : x_1 : \cdots : x_r]$ with $x_i \in \mathbb{C}$. We denote the coordinate points of \mathbb{P}^r by P_0, P_1, \ldots, P_r , where P_i is the point where all coordinates except x_i vanish. We denote by $L_{\{i_1,i_2\}}$ (where $0 \le i_1, i_2 \le r$ and $i_1 \ne i_2$) the coordinate line passing through P_{i_1} and P_{i_2} . We write $P_{\{i_1,i_2\}}$ for the midpoint of this line, i.e. the point where $x_{i_1} = x_{i_2}$ and $x_i = 0$ for $i \ne i_1, i_2$. We also write H_i for the *i*th coordinate hyperplane $\{x_i = 0\} \subseteq \mathbb{P}^r$.

Throughout, we use the action of $T := (\mathbb{C}^*)^{r+1}$ on \mathbb{P}^r by scaling the coordinates. We have $H^*_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q}) = \mathbb{Q}[\alpha_0, \ldots, \alpha_r]$, and $H^*_{T, \operatorname{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q}) = \mathbb{Q}(\alpha_0, \ldots, \alpha_r)$, where $-\alpha_i$ is the weight of the character $T \to \mathbb{C}^*$ defined by $(\lambda_0, \ldots, \lambda_r) \mapsto \lambda_i$.

We denote by $[P_i]$, (resp. $[L_{\{i_1,i_2\}}]$, $[H_i]$) the class in $H_T^*(\mathbb{P}^r, \mathbb{Q})$ identified with P_i (resp. $L_{\{i_1,i_2\}}$, H_i) under the equivariant Poincaré duality isomorphism.

3.2.2 The Chen-Ruan cohomology of $Sym^d \mathbb{P}^r$

In Section 3.1.3, we saw that there exists an isomorphism $H^2_T(\operatorname{Sym}^d X, \mathbb{Q}) \cong H^2_T(X, \mathbb{Q})$. For convenience, we now fix such an isomorphism. If $\alpha \in H^2_T(X, \mathbb{Q})$, then we write $\alpha \in H^2_T(\operatorname{Sym}^d X, \mathbb{Q})$ for the class that pulls back to

$$\sum_{j=1}^{d} \operatorname{pr}_{j}^{*} \alpha \in H_{T}^{2}(X^{d}, \mathbb{Q}).$$

Using this isomorphism, we write $[H_i] \in H^2_T(\operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q})$, which gives a distinguished set of generators for $H^2_T(\operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q})$. (Via the Poincaré pairing, this defines a natural notion of the degree of a curve on $\operatorname{Sym}^d \mathbb{P}^r$.)

From Section 3.1.3, the components of $I \operatorname{Sym}^{d} \mathbb{P}^{r}$ are indexed by $\operatorname{Part}(d)$, and are given by

$$(I\operatorname{Sym}^{d} \mathbb{P}^{r})_{\sigma} \cong \prod_{i \ge 1} \operatorname{Sym}^{\operatorname{Mult}(\sigma,i)} \mathbb{P}^{r}.$$

We then have

$$H_T^*((I\operatorname{Sym}^d \mathbb{P}^r)_\sigma, \mathbb{Q}) \cong \bigoplus_{i\geq 1} H_T^*(\operatorname{Sym}^{\operatorname{Mult}(\sigma,i)} \mathbb{P}^r, \mathbb{Q}).$$

For $\eta \in \sigma$, there is a corresponding factor $H_T^*(\text{Sym}^{\text{Mult}(\sigma,i)} \mathbb{P}^r, \mathbb{Q})$, and we denote by $[H_{\sigma,\eta,i}]$ the class $[H_i]$ in this factor. We write $[H_{\sigma,i}]$ for $\sum_{\eta} [H_{\sigma,\eta,i}]$.

The Chen-Ruan cohomology is

$$H^*_{CR,T}(\operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q}) := \bigoplus_{\sigma \in \operatorname{Part}(d)} H^*_T((I\operatorname{Sym}^d \mathbb{P}^r)_\sigma, \mathbb{Q}),$$

with an appropriate shift in grading that we will not need.

3.2.3 The action of $(\mathbb{C}^*)^{r+1}$ on $\operatorname{Sym}^d \mathbb{P}^r$

The natural action of $T := (\mathbb{C}^*)^{r+1}$ on \mathbb{P}^r induces a diagonal T-action on $(\mathbb{P}^r)^d$. This commutes with the action of S_d , hence acts on $\operatorname{Sym}^d \mathbb{P}^r$. It is easy to check that this action agrees with that on $\widetilde{\operatorname{Sym}}^d \mathbb{P}^r$ defined by postcomposition of $f' : S \to \mathbb{P}^r$ with the action on \mathbb{P}^r . The T-action on $\operatorname{Sym}^d \mathbb{P}^r$ induces an action on $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ for all n, and β .

CHAPTER 4

Localization and virtual localization

In this chapter, we give a detailed description of $T := (\mathbb{C}^*)^{r+1}$ -equivariant geometry of $\operatorname{Sym}^d \mathbb{P}^r$ and $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. We focus on the aspects of this geometry required to compute Gromov-Witten invariants. In particular, we describe the *T*-fixed points and 1-dimensional *T*-orbits in $\operatorname{Sym}^d \mathbb{P}^r$. We use this to give a complete description (Theorem 4.5.23) of the *T*-fixed locus in $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$, and we compute the *T*-equivariant Euler class of the (virtual) normal bundle to the *T*-fixed locus (Proposition 4.6.2). We also give an example-heavy discussion (Section 4.2) of what can be said for general targets (smooth projective varieties).

4.1 The localization and virtual localization theorems

If $T = (\mathbb{C}^*)^r$ acts on a Deligne-Mumford stack X, the equivariant cohomology $H^*_T(X, \mathbb{Q})$ is a module over $H^*_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q}) \cong \mathbb{Q}[\alpha_0, \ldots, \alpha_r]$, where $-\alpha_i$ is the weight of the character $T \to \mathbb{C}^*$ defined by $(\lambda_0, \ldots, \lambda_r) \mapsto \lambda_i$. We write $H^*_{T, \operatorname{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})$ for the localization $\mathbb{Q}(\alpha_0, \ldots, \alpha_r)$, and more generally $H^*_{T, \operatorname{loc}}(X, \mathbb{Q}) := H^*_T(X, \mathbb{Q}) \otimes_{H^*_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q})} H^*_{T, \operatorname{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})$. We will often use the Atiyah-Bott *localization theorem*, as well as Graber-Pandharipande's generalization, the *virtual localization theorem*.

Theorem 4.1.1 (Atiyah-Bott [5], see Edidin-Graham [22] for statement in the Chow ring). Let T be a torus acting on a smooth compact manifold X, with fixed point set F. Then the map $(\iota_F)_* : H^*_{T,\text{loc}}(F, \mathbb{Q}) \to H^*_{T,\text{loc}}(X, \mathbb{Q})$ is an isomorphism, where $(\iota_F)_*$ is the Gysin map associated to the inclusion $F \hookrightarrow X$. The inverse map is $\iota_F^*/e_T(N_{F|X})$, where $e_T(N_F)$ is the equivariant Euler class of the normal bundle to F. In particular, for $\alpha \in H^*_{T,\text{loc}}(X, \mathbb{Q})$, we have

$$\int_X \alpha = \int_X (\iota_F)_* \left(\frac{\iota_F^* \alpha}{e_T(N_F)} \right) = \int_F \frac{\iota_F^* \alpha}{e_T(N_F)}.$$

Theorem 4.1.2 (Graber-Pandharipande [29]). Let X be a Deligne-Mumford stack with a T-action and a T-equivariant perfect obstruction theory E^{\bullet} . Again, let $\iota_F : F \hookrightarrow X$ denote the inclusion of the fixed locus. Let $[X]^{\text{vir}}$ denote the virtual fundamental class associated to E^{\bullet} . The T-fixed part of E^{\bullet} defines a perfect obstruction theory on F, with virtual fundamental class $[F]^{\text{vir}}$. The virtual normal bundle N_F^{vir} to F is the T-moving part of E^{\bullet} . Then

$$\int_{[X]^{\text{vir}}} \alpha = \int_{[F]^{\text{vir}}} \frac{\iota_F^* \alpha}{e_T(N_F^{\text{vir}})}.$$
(4.1)

Remark 4.1.3. The proof in [29] requires that X have an equivariant embedding into a smooth Deligne-Mumford stack, but this condition was removed in [11]. Also, it is usually convenient to write F as a union of connected components (or a union of open and closed substacks) F_j , in which case (4.1) becomes

$$\int_{[X]^{\mathrm{vir}}} \alpha = \sum_{j} \int_{[F_j]^{\mathrm{vir}}} \frac{\iota_{F_j}^* \alpha}{e_T(N_{F_j}^{\mathrm{vir}})}.$$

4.2 The *T*-fixed locus in $\overline{\mathcal{M}}_{0,n}(X,\beta)$, nonorbifold case

If X has a T-action, then postcomposition induces a T-action on $\overline{\mathcal{M}}_{0,n}(X,\beta)$ for all n and β . In this section, we describe the behavior of torus-fixed stable maps to X when X is a smooth projective variety. The main purpose of the section is to emphasize, via examples, that this behavior is often very complicated, and one needs a lot of information about X and its torus action in order to write down $\overline{\mathcal{M}}_{0,n}(X,\beta)$ explicitly. This section contains no original results.

Definition 4.2.1. Let X be a smooth projective variety with an action of a torus T. The T-graph $\Gamma_{X,T}$ of X is the graph with a vertex v for every connected component X_v^T of X^T , and an edge $e = \{v_1, v_2\}$ for every connected component of the union of all 1-dimensional T-orbits joining v_1 and v_2 .

As an example, the T-graphs for $\mathbb{P}^2 \times \mathbb{P}^2$ with two different torus actions are shown in Figure 4.1.

Remark 4.2.2. By Lemma 5 in [50], $\Gamma_{X,T}$ contains no self-edges. We do not know whether or not $\Gamma_{X,T}$ may contain multiple edges between two fixed vertices.

Example 4.2.3. In the case where T acts on X with isolated fixed points and 1-dimensional orbits, $\Gamma_{X,T}$ is identified (as a topological space) with the *1-skeleton* of X, i.e. the union of 0-dimensional and 1-dimensional T-orbits. This condition holds, for example, in any toric variety, e.g. Figure

When the *T*-graph of *X* is known, $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$ may be written as a union of combinatorially indexed pieces. If one is lucky, it may be clear how these pieces fit together into connected



Figure 4.1: The *T*-graph of $\mathbb{P}^2 \times \mathbb{P}^2$ with the action of its dense torus, and the diagonal action of $(\mathbb{C}^*)^3$, respectively.

components, or unions of connected components, of $\overline{\mathcal{M}}_{0,n}(X,\beta)$. (Computing these may still be difficult.) For $e \in E(\Gamma_{X,T})$ a union of 1-dimensional *T*-orbits, we denote by $H_2(X,\mathbb{Z})_e$ the subgroup of $H_2(X,\mathbb{Z})$ generated by the classes of these 1-dimensional orbits. (They need not all be linearly equivalent, as we see in Example 4.2.14.)

Definition 4.2.4. An *n*-marked (X, T)-decorated tree $\tilde{\Gamma} = (\Gamma, \text{Mark}, \text{f}, d, \text{vdeg})$ is

- A tree Γ ,
- A marking map Mark : $\{1, \ldots, n\} \rightarrow V(\Gamma)$,
- A map f of graphs $\Gamma \to \Gamma_{X,T}$,
- An edge degree map $\beta_{edge}: E(\Gamma) \to H_2(X, \mathbb{Z})_e$, and
- A vertex degree map vdeg : V(Γ) → H₂(𝔅(v), ℤ). (Recall that 𝔅(v) is a connected component of X^T.)

We will often refer to an *n*-marked (X, T)-decorated tree as simply a "decorated tree" when the the meaning is clear.

Definition 4.2.5. The *degree* $\beta(\tilde{\Gamma})$ of an *n*-marked (X, T)-decorated tree is

$$\beta(\tilde{\Gamma}) := \sum_{e \in E(\Gamma)} \beta_{edge}(e) + \sum_{v \in V(\Gamma)} \operatorname{vdeg}(v) \in H_2(X, \mathbb{Z}).$$

Notation 4.2.6. We denote the set of n-marked (X, T)-decorated trees of degree β by Trees_{0,n} (X, β) .

Proposition 4.2.7. There is a natural map

$$\Psi: (\overline{\mathcal{M}}_{0,n}(X,\beta))^T \to \operatorname{Trees}_{0,n}(X,\beta).$$

Proof. If $(f : C \to X) \in \overline{\mathcal{M}}_{0,n}(X,\beta)^T$, then on each irreducible component C, the restriction of f either

- 1. factors through X^T or
- 2. is a finite map to the closure of a 1-dimensional *T*-orbit.

We define a graph Γ_f whose vertex set $V(\Gamma_f)$ is the set of connected components of $f^{-1}(X^T)$ (which are unions of components of type 1), and whose edge set $E(\Gamma_f)$ is the set of irreducible components of type 2. Any irreducible component of type 2 contains exactly two distinct points that map to X^T , so $E(\Gamma_f)$ makes sense as a set of edges. The fact that Γ_f is a tree follows from the fact that C is genus zero.

There is a natural map f_f of graphs from Γ_f to $\Gamma_{X,T}$, which sends a connected component C_0 of $f^{-1}(X^T)$ to the component of X^T containing f(C), and sends an irreducible component $C_1 \in E(\Gamma_f)$ to the component of the union of 1-dimensional T-orbits that contains $f(C_1)$.

Since $(f : C \to X)$ is *T*-fixed, each marked point is mapped to X^T . We define a marking map $\operatorname{Mark}_f : \{1, \ldots, n\} \to V(\Gamma_f)$ that sends *i* to the connected component of $f^{-1}(X^T)$ containing b_i . We define a map $\beta_{edge,f} : E(\Gamma) \to \mathbb{N}$ that takes an edge *e* to $(f|_{C_e})_*[C_e]$, and a map $\operatorname{vdeg}_f : V(\Gamma) \to H_2(\mathbb{f}(v), \mathbb{Z})$ that sends a vertex *v* to $(f|_{C_v})_*[C_v]$. We then define $\Psi(f : C \to X) = (\Gamma_f, \mathbb{f}_f, \operatorname{Mark}_f, \beta_{edge,f}, \operatorname{vdeg}_f)$.

Notation 4.2.8. Let $\tilde{\Gamma}$ be a decorated tree, and let $(f : C \to X) \in \Psi^{-1}(\tilde{\Gamma})$. If $v \in V(\Gamma)$, then from Proposition 4.2.7, v corresponds to a union of irreducible components of C, or possibly a single point. We denote this by C_v . Similarly, an edge $e \in E(\Gamma)$ corresponds to an irreducible component of C, and we denote it by C_e . For $(v, e) \in F(\Gamma)$, we write $\xi(v, e)$ for the point $C_v \cap C_e \in C$. We say (v, e) is a *special flag* if $\xi(v, e)$ is a special point (mark or node), equivalently if val(v) > 1, vdeg(v) > 0, or $Mark^{-1}(v) \neq \emptyset$.

For any $(f : C \to X) \in \Psi^{-1}(\tilde{\Gamma})$, the tangent space to C_e at $\xi(v, e)$ is naturally a (fractional) character of T. We denote the weight of this character by $w(v, e) \in H^2_T(\operatorname{Spec}(\mathbb{C}), \mathbb{Q})$. It is straightforward to check that this does not depend on the choice of $(f : C \to X)$.

We adapt the following notation from [46], Definition 53:

$$\begin{split} V^{1}(\tilde{\Gamma}) &= \{ v \in V(\Gamma) | \operatorname{val}(v) = 1, \operatorname{vdeg}(v) = 0, \left| \operatorname{Mark}^{-1}(v) \right| = 0 \} \\ V^{1,1}(\tilde{\Gamma}) &= \{ v \in V(\Gamma) | \operatorname{val}(v) = 1, \operatorname{vdeg}(v) = 0, \left| \operatorname{Mark}^{-1}(v) \right| = 1 \} \\ V^{2}(\tilde{\Gamma}) &= \{ v \in V(\Gamma) | \operatorname{val}(v) = 2, \operatorname{vdeg}(v) = 0, \left| \operatorname{Mark}^{-1}(v) \right| = 0 \} \\ V^{S}(\tilde{\Gamma}) &= V(\Gamma) \smallsetminus (V^{1}(\tilde{\Gamma}) \cup V^{1,1}(\tilde{\Gamma}) \cup V^{2}(\tilde{\Gamma})). \end{split}$$

We call vertices in $V^S(\tilde{\Gamma})$ stable. A vertex v is stable if and only if C_v is 1-dimensional (a prestable curve, rather than a single point).

For $v \in V^1(\tilde{\Gamma}) \cup V^{1,1}(\tilde{\Gamma})$, we will usually write $E(\Gamma)_v = \{e_v = \{v, v'\}\}$. For $v \in V^2(\tilde{\Gamma})$, we will usually write $E(\Gamma)_v = \{e_v^1 = \{v, v_1\}, e_v^2 = \{v, v_2\}\}$.

Example 4.2.9. If T acts on X with isolated fixed points and 1-dimensional orbits, then the fibers $\mathcal{M}_{\tilde{\Gamma}} := \Psi^{-1}(\tilde{\Gamma})$ are easy to describe. In this case a decorated tree $\tilde{\Gamma}$ determines irreducible components C_e and maps $f|_{C_e}$. The only things that are not determined are the curves C_v , which are stable genus zero curves with marking sets $\operatorname{Mark}^{-1}(v) \cup E(\Gamma)_v$. (The map $f|_{C_v}$ is the constant map from C_v to the single point f(v).) Thus we have¹:

$$\mathcal{M}_{\tilde{\Gamma}} \cong \left(\prod_{v \in V(\Gamma)} \overline{\mathcal{M}}_{0, \operatorname{Mark}^{-1}(v) \cup E(\Gamma)_{v}} \right) / \operatorname{Aut}(\tilde{\Gamma})$$
(4.2)

(The automorphism group $\operatorname{Aut}(\tilde{\Gamma})$ consists of automorphisms of Γ that commute with the maps Mark, \mathfrak{f} , and β_{edge} .)

Since these fibers are proper, and there are finitely many of them for fixed n and β , they are in fact open and closed substacks of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$. (Since they are connected, they are exactly the connected components of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$.) In other words, if T acts on X with isolated fixed points and 1-dimensional orbits, then we have completely described the structure of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$: it is a disjoint union, over index set $\operatorname{Trees}_{0,n}(X,\beta)$, of spaces $\mathcal{M}_{\tilde{\Gamma}}$ as in (4.2).

Remark 4.2.10. It is important to note that this example is very special: in general, the fibers of Ψ^{-1} are only locally closed substacks of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$, as we see in the following example. Also, in general one should not expect $\mathcal{M}_{\tilde{\Gamma}}$ to be connected.

Example 4.2.11. Let \mathbb{C}^* act on $(\mathbb{P}^1)^2$ diagonally. The fixed points of the action are:

$$(0,0),(0,\infty),(\infty,0),(\infty,\infty).$$

(We identify \mathbb{P}^1 with $\mathbb{C} \cup \{\infty\}$ for now.) The complement of these points is a union of (1-dimensional) orbits. The *T*-graph is:

$$\Gamma_{(\mathbb{P}^1)^2,\mathbb{C}^*} =$$

The outer edges of the square correspond to unique orbits, whereas the diagonal edge corresponds

¹To be precise, one must further take the quotient by the trivial action of a group of automorphisms of the ramified cover $C_e \to f(C_e)$ for each edge e.



Figure 4.2: The torus orbits in $(\mathbb{P}^1)^2$

to the family of orbits of the form

$$\{(t,at): t \in \mathbb{C}^*\}.\tag{4.3}$$

The base of this family, given by the *a*-coordinate, is isomorphic to \mathbb{C}^* . (See Figure 4.2; this family is shown in red.)

If $(f : C \to (\mathbb{P}^1)^2)$ is \mathbb{C}^* -fixed, then an irreducible component of C is either contracted to one of the four fixed points, or maps as a ramified cover of a 1-dimensional orbit, ramified over two of the fixed points.

Consider a degree d cover by \mathbb{P}^1 of an orbit of the form (4.3), ramified over (0,0) and (∞,∞) . As a stable map, this lies in $\Psi^{-1}(\tilde{\Gamma}) \subseteq \overline{\mathcal{M}}_{0,0}((\mathbb{P}^1)^2, (d, d))$, where Γ is the tree $v \bullet \overset{e}{-} \bullet v'$, with

- f(v) = (0, 0),
- $f(v') = (\infty, \infty),$
- f(e) equal to the diagonal edge in Γ_{(ℙ¹)²,C*}
 ,
- $\beta_{edge}(e) = (d, d)$, and
- $\operatorname{vdeg}(v) = \operatorname{vdeg}(v') = 0.$

The family (4.3) of orbits induces a family $(f_a : \mathbb{P}^1 \to (\mathbb{P}^1)^2)$ of stable maps, corresponding to a map $\mathbb{C}^* \hookrightarrow \overline{\mathcal{M}}_{0,0}((\mathbb{P}^1)^2, (d, d))$. It is easy to compute that the limit as $a \to 0$ is the stable map $f_0 : C \to (\mathbb{P}^1)^2$, where C is isomorphic to two copies of \mathbb{P}^1 glued at a node, and f_0 sends the two components to the lines $\mathbb{P}^1 \times 0$ and $\infty \times \mathbb{P}^1$, each with degree d, each fully ramified over the two \mathbb{C}^* -fixed points.

The limit $(f_0: C \to (\mathbb{P}^1)^2)$ lies in the fiber $\Psi^{-1}(\tilde{\Gamma}_0 = (\Gamma_0, \mathfrak{f}_0, \beta_{edge,0}, \operatorname{vdeg}_0))$, where Γ_0 is the tree $v \bullet \underbrace{e_1 \bullet e_2}{v_1 \bullet v'}$, with

- $f_0(v) = (0,0),$
- $f_0(v_1) = (\infty, 0),$
- $f_0(v') = (\infty, \infty).$
- $f_0(e_1)$ equal to the bottom edge of the square in $\Gamma_{(\mathbb{P}^1)^2,\mathbb{C}^*}$,
- $f_0(e_2)$ equal to the rightmost edge of the square,
- $\beta_{edae.0}(e_1) = (d, 0),$
- $\beta_{edae,0}(e_2) = (0, d)$, and
- $\operatorname{vdeg}_0(v) = \operatorname{vdeg}_0(v_1) = \operatorname{vdeg}_0(v') = 0.$

This shows that $\Psi^{-1}(\tilde{\Gamma}_0) \subseteq \overline{\Psi^{-1}(\tilde{\Gamma})} \subseteq \overline{\mathcal{M}}_{0,0}((\mathbb{P}^1)^2, (d, d))$. (The former is a single point, with some automorphism group.)

Note that if we had taken the limit $a \to \infty$ instead, we would have gotten a similar limit f_{∞} , but the two components would map to the top and leftmost edges of the square.

Remark 4.2.12. In Example 4.2.11, $\tilde{\Gamma}$ is obtained from $\tilde{\Gamma}_0$ by contracting a chain of edges $e_1 \cup e_2$, with intermediate vertex v_1 , satisfying:

- $v_1 \in V^2(\tilde{\Gamma})$.
- $w(v_1, e_1) + w(v_1, e_2) = 0.$

This operation of "chain contraction" defines a partial ordering \leq on Trees_{0,n}($(\mathbb{P}^1)^2, \beta$), where $\tilde{\Gamma} \leq \tilde{\Gamma}'$ if $\tilde{\Gamma}$ may be reached from $\tilde{\Gamma}'$ via a sequence of chain contractions. Generalizing the example, one may check that $\tilde{\Gamma} \leq \tilde{\Gamma}'$ if and only if $\Psi^{-1}(\tilde{\Gamma}') \subseteq \overline{\Psi^{-1}(\tilde{\Gamma})}$. Furthermore, \leq has the property that if $\tilde{\Gamma}_1 \leq \tilde{\Gamma}'$ and $\tilde{\Gamma}_2 \leq \tilde{\Gamma}'$, then there exists $\tilde{\Gamma}$ satisfying $\tilde{\Gamma} \leq \tilde{\Gamma}_1$ and $\tilde{\Gamma} \leq \tilde{\Gamma}_2$.

It follows from these observations that the connected components of $\overline{\mathcal{M}}_{0,n}((\mathbb{P}^1)^2,\beta)$ are in bijection with decorated trees that are *minimal with respect to* \leq , namely those that admit no chain contractions, where the bijection identifies $\tilde{\Gamma}$ with $\overline{\mathcal{M}}_{\tilde{\Gamma}} := \Psi^{-1}(\tilde{\Gamma})$.

In fact, using Example 4.2.11, one may explicitly write down $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ as a stack, in terms of certain moduli spaces of marked curves. This is closely related to the computation in Section 4.5 for $X = \operatorname{Sym}^d \mathbb{P}^r.$



Figure 4.3: The graph $\Gamma_{\mathbb{P}^2,\mathbb{C}^*}$ and a picture of the 1-dimensional \mathbb{C}^* -orbits

Caution 4.2.13. Working from Example 4.2.11, one may be led to make all sorts of incorrect conjectures. The next few examples illustrate some of these.

Example 4.2.14. In Example 4.2.11, the ramification data of a *T*-fixed stable map was essentially constant in families. More specifically, when taking limits in $\overline{\mathcal{M}}_{0,n}((\mathbb{P}^1)^2, \beta)$, the only combinatorial difference between the special and general fibers is that irreducible components mapping as degree *d* fully ramified covers could be replaced with chains of rational curves, also mapping as degree *d* fully ramified covers. We see now that ramification data may change in a degeneration of *T*-fixed stable maps. Further, this change is not detected by our naive notion of decorated trees.

Consider $X = \mathbb{P}^2$, with \mathbb{C}^* acting by $t \cdot [x_0 : x_1 : x_2] = [x_0 : tx_1 : t^2x_2]$. The curves $x_1^2 = ax_0x_2$ are 1-dimensional orbit closures for $a \in \mathbb{C}^*$. The graph $\Gamma_{\mathbb{P}^2,\mathbb{C}^*}$ is shown in Figure 4.3, along with a pictorial representation of the orbits. We may identify this family of curves with a family of stable maps, namely the inclusions of the 1-dimensional orbit closures $C_a \hookrightarrow \mathbb{P}^2$. Taking the limit as $a \to \infty$ gives a similar answer to that in Example 4.2.11: a nodal source curve C_{∞} , embedding in \mathbb{P}^2 as the union of the lines $L_{\{1,2\}}$ and $L_{\{0,1\}}$. (Recall from Section 3.2.1 that $L_{\{1,2\}} := \{x_0 = 0\}$ and $L_{\{0,1\}} = \{x_2 = 0\}$.) However, taking the limit as $a \to 0$ gives a degree 2 map f_0 from $C_0 \cong \mathbb{P}^1$ to the line $L_{\{0,2\}}$, ramified over the points P_2 and P_0 .

In this example, the decorated tree is the same when $a \neq 0$ or a = 0, namely $\tilde{\Gamma} = v \bullet e \bullet v'$, with

- $f(v) = P_2$,
- $f(v) = P_0$,
- f(e) is the bottom edge of the triangle,
- $\beta_{edge}(e) = 2[L]$, and
- $\operatorname{vdeg}(v) = \operatorname{vdeg}(v') = 0.$

We note that the tangent weights at the points C_v and $C_{v'}$ are constant. To see this, note that the action on the tangent space at P_2 is equivariantly isomorphic to $t \cdot (x_0, x_1) = (t^{-2}x_0, t^{-1}x_1)$ on \mathbb{C}^2 . For every $a \neq 0$, $f(C_e)$ is tangent to the line $L_{\{1,2\}}$, so the tangent weight w(v, e) is $-\lambda$, where $\lambda \in H^2_{\mathbb{C}^*}(\operatorname{Spec}(\mathbb{C}), \mathbb{Q})$ is the canonical generator. When $a \to 0$, $f(C_e)$ is tangent to $L_{\{0,2\}}$, and is f is ramified at C_v with degree 2. Thus we have $w(v, e) = \frac{-2\lambda}{2} = -\lambda$.

The following example is modified from one pointed out to us by D. Speyer.

Example 4.2.15. In Example 4.2.11, we saw that if a graph $\tilde{\Gamma}'$ contained $v \bullet \frac{e_1 \bullet e_2}{v_1} \bullet v'$ as a subgraph, where

- $v_1 \in V^2(\tilde{\Gamma}')$ and
- $w(v_1, e_1) + w(v_1, e_2) = 0$,

then any $(f': C' \to (\mathbb{P}^1)^2) \in \Psi^{-1}(\tilde{\Gamma}')$ could be deformed to some $(f: C \to (\mathbb{P}^1)^2) \in \Psi^{-1}(\tilde{\Gamma})$, where $\tilde{\Gamma}$ is obtained from $\tilde{\Gamma}'$ by contracting the chain $e_1 \cup e_2$.

It is easy to see that the conditions $v_1 \in V^2(\tilde{\Gamma}')$ and $w(v_1, e_1) + w(v_1, e_2) = 0$ are necessary for such a deformation to exist. In general, a node of a curve may only be equivariantly smoothed if the tangent weights on the two branches add to zero. If $v_1 \notin V^2(\tilde{\Gamma}')$, then $v_1 \in V^S(\tilde{\Gamma}')$. Thus at the nodes $\xi(v_1, e_1)$ and $\xi(v_1, e_2)$, the tangent weight on one branch is zero, while the other is nonzero. Such nodes cannot deform.

However, we show that the two conditions are not sufficient. Consider \mathbb{P}^2 , with the \mathbb{C}^* -action $t \cdot [x_0 : x_1 : x_2] = [x_0 : tx_1 : t^2x_2]$. We blow up \mathbb{P}^2 at P_1 . Note that the \mathbb{C}^* -action lifts to this blowup, as pictured in Figure 4.4. It is easy to check that if we interpret the red curves as a family of stable maps and take the limit "upwards", the limit curve is a chain of three rational components, mapping to the three black lines. The middle component maps as a degree 2 ramified cover.

(One way to see this is by degree-counting. The generic map is degree 2H, where H is the pullback of the hyperplane class. In the limit, the degree is 2(H - E) + kE, where E is the exceptional divisor, H - E is the class of the proper transforms of the lines $L_{\{1,2\}}$ and $L_{\{0,1\}}$, and k is the degree of the map on the middle component. For the degree to be constant, we must have k = 2. We can also check that the tangent weights on the middle component are double those on the axes.)

Now consider the following stable map. Let C be isomorphic to two copies of \mathbb{P}^1 glued at a node, and let f map one copy to (the proper transform of) the line $L_{\{1,2\}}$ isomorphically, and the second copy to the exceptional divisor as a degree 2 ramified cover. It is easy to check (and follows from the previous paragraph) that the two tangent weights at the node are opposites. On the other hand, f may not deform. Indeed, it would need to deform to a cover of an irreducible 1-dimensional



Figure 4.4: The graph $\Gamma_{\mathcal{B}\ell_{P_1}\mathbb{P}^2,\mathbb{C}^*}$ and a picture of the 1-dimensional \mathbb{C}^* -orbits

orbit closure, but we have already classified all such orbits. We say a chain contraction is *realizable* if it corresponds to a smoothing of a node on the source curve of a stable map. This example then shows that not every chain contraction is realizable.

Note that the characterization of the limit of generic 1-dimensional orbit closures also shows that there are realizable chain contractions where the two branches of the node being smoothed have different ramification degrees.

Definition 4.2.16. In [50], and later in [13], there are introduced (slightly incompatible) notions of *breaking nodes* and *breaking subcurves*. We use the definition from [13]; namely, a *breaking node* of a stable map $(f : C \to X)$ is one where the tangent weights of the two branches do not add up to zero.

By definition, $\Psi(f)$ contains a flag (or perhaps two) for every node, except those nodes on the interior of a subcurve that maps to X^T . These nodes are not breaking, since both tangent weights are zero. Therefore the set of breaking nodes includes into $F(\Gamma)$; we call a flag a *breaking flag* if it is in the image of this map.

Remark 4.2.17. Any flag (v, e) with $v \in V^S(\tilde{\Gamma})$ is a breaking flag.

Example 4.2.18. As we saw above, breaking nodes may not be equivariantly deformed. We saw in Example 4.2.15 that it may also be impossible to deform a nonbreaking node. We illustrate one more case of this, where the node is on the interior of a subcurve that maps to X^T .

First, consider the blowup Y of \mathbb{P}^2 at the points [1:0:1], [2:0:1], and [3:0:1]. The inclusion of the proper transforms of the x_0 and x_1 axes is a stable map to X. If it were to deform to a smooth curve, if would need to deform to the proper transform of a smooth plane conic passing through the three given points. But a smooth plane conic cannot pass through three collinear points. Therefore, the node cannot be smoothed.

To relate this to nonbreaking nodes, we need to realize Y as a component of the fixed locus of some variety X. We may simply take $X = Y \times \mathbb{P}^1$, with \mathbb{C}^* acting on \mathbb{P}^1 .

Having seeing these pathologies, we must settle for a very weak proposition, with no hope for any kind of converse.

Proposition 4.2.19. Let $(f_0 : C_0 \to X)$ be in the closure of $\mathcal{M}_{\tilde{\Gamma}}$. Then $\tilde{\Gamma}$ can be obtained from $\Psi(f)$ by a sequence of chain contractions.

Proof. Let $(f_t : C_t \to X) \in \mathcal{M}_{\tilde{\Gamma}}$ be a family of stable maps whose limit as $t \to 0$ is $(f_0 : C_0 \to X)$. Locally at t = 0, C_t is a smoothing of some collection of nodes of C_0 . By equivariance, these nodes are nonbreaking, hence none of them connects an edge curve C_e to a vertex curve C_v . Thus smoothing these nodes corresponds to deleting vertices in $V^2(\tilde{\Gamma}_0)$, i.e. a sequence of chain contractions.

4.3 The *T*-fixed locus in $\overline{\mathcal{M}}_{0,n}(X,\beta)$, orbifold case

If X is an orbifold, the situation is much more subtle than that above, and we know very little about the general case. As above, we may use decorated trees to isolate certain open and closed substacks of $\overline{\mathcal{M}}_{0,n}(X,\beta)^T$. We give a refinement of the definition that captures some of the "orbifold data:"

Definition 4.3.1. An *n*-marked (X, T)-decorated tree $\tilde{\Gamma} = (\Gamma, \text{Mark}, \text{f}, d, \text{vdeg})$ is

- A tree Γ ,
- A marking map $Mark : \{1, \ldots, n\} \to V(\Gamma)$,
- A map f of graphs $\Gamma \to \Gamma_{X,T}$,
- An edge degree map $\beta_{edge}: E(\Gamma) \to H_2(X, \mathbb{Q})_e$, and
- A vertex degree map vdeg : $V(\Gamma) \to H_2(\mathbb{f}(v), \mathbb{Q})$.
- A "monodromy map" Mon that assigns to each i ∈ {1,...,n} a component of the inertia stack of f(Mark(i)), and to each flag (v, e) ∈ F(Γ) a conjugacy class in G_{ξ(v,e)}.

Remark 4.3.2. There are various conditions on these objects for the associated moduli space to be nonempty, such as: if $v \in V^2(\Gamma)$, then $Mon(v, e_v^1)$ and $Mon(v, e_v^2)$ must be in inverse conjugacy classes of $G_{\xi(v,e)}$.

In the nonorbifold case, there was a simple description of the T-fixed stable maps from an orbifold \mathbb{P}^1 to a specified 1-dimensional orbit closure; namely, a map is determined by its ramification degree at the two fixed points. In the orbifold case, it is much less clear what to do:

• When X is a toric stack, these maps may be computed exactly — this appears in [46], using methods from [36].



Figure 4.5: A point with one dimensional T-orbit in Sym¹⁷ \mathbb{P}^2 , with $i_1 = 1$ and $i_2 = 2$

- For $X = \operatorname{Sym}^{d} \mathbb{P}^{r}$, we are able to compute the maps, and we do so in the next two sections.
- More generally, if X is a quotient of a variety by a finite group G, then the maps may be described using principal G-bundles.
- If X (or more specifically the specified 1-dimensional orbit) is not toric, and is not a quotient of a variety by a finite group, then we do not know how to describe the maps.

4.4 *T*-Fixed points and 1-dimensional *T*-orbits in Sym^d \mathbb{P}^r

As in Section 3.1.1, we write $(d(\bullet) \xrightarrow{P} \mathbb{P}^r)$ to denote a point $\operatorname{Spec} \mathbb{C} \to \operatorname{Sym}^d \mathbb{P}^r$. We use notation from Section 3.2.1 for coordinate points and lines in \mathbb{P}^r .

Proposition 4.4.1. *1.* A point $(d(\bullet) \xrightarrow{P} \mathbb{P}^r) \in \text{Sym}^d \mathbb{P}^r$ is *T*-fixed if and only if $\text{Im}(P) \subseteq \{P_0, \ldots, P_r\}$.

2. $(d(\bullet) \xrightarrow{P} \mathbb{P}^r)$ is in a 1-dimensional T-orbit if and only if (it is not T-fixed and) $\operatorname{Im}(P) \subseteq \{P_0, \ldots, P_r\} \cup L_{\{i_1, i_2\}}$ for some $0 \le i_1, i_2 \le r$. (See Figure 4.5.)

Proof. 1 follows from the description of the *T*-action on $\widetilde{\text{Sym}}^d \mathbb{P}^r$, and the fact that $\{P_0, \ldots, P_r\}$ is the *T*-fixed locus of \mathbb{P}^r .

The *r*-dimensional subtorus defined by $t_{i_1} = t_{i_2}$ acts trivially on $\{P_0, \ldots, P_r\} \cup L_{\{i_1, i_2\}}$, proving the backwards direction of 4.4.1. If $\text{Im}(P) \not\subseteq \{P_0, \ldots, P_r\} \cup L_{\{i_1, i_2\}}$, the *T*-orbit is clearly at least 2-dimensional.

Remark 4.4.2. The *T*-fixed points of $\text{Sym}^d \mathbb{P}^r$ are in natural bijection with ZPart(d, r+1), where the *i*th part is the number of points of $d(\bullet)$ mapping to P_i . (See Section 1.7.) We will use this identification from now on.

By the second part of Proposition 4.4.1, for each 1-dimensional T-orbit there is the associated data:



Figure 4.6: The T-graph, and a picture of 1-dimensional orbits, for $\text{Sym}^3 \mathbb{P}^2$

- 1. Two (unordered) indices i_1 and i_2 ,
- 2. An element of ZPart(d'), where d' < d is the number of points of P mapping to $\{P_0, \ldots, P_r\}$, and
- 3. A (d d')-tuple of points of $L_{\{i_1, i_2\}}$, up to scaling.

In Figure 4.5, 1 refers to the indices 0 and 2, and 2 refers to the nonnegative ordered partition (2, 5, 3). (Of course, 3 depends on the relative coordinates of the 7 points on $L_{\{0,2\}}$.) Explicitly, this orbit is obtained from the point shown by simultaneously scaling the 7 points on the line $L_{\{0,2\}}$. As a consequence of Proposition 4.4.1, we obtain a description of $\Gamma_{\text{Sym}^d \mathbb{P}^r,T}$, pictured in Figure 4.6 for d = 3 and r = 2. (The frontispiece to this thesis is $\Gamma_{\text{Sym}^4 \mathbb{P}^2,T}$.)

4.5 The *T*-fixed locus in $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$

This section, which is in some sense the essential part of the proof of Theorem 6.1.2, draws heavily on the notations introduced in Sections 1.7, 3.1.1, 3.2.1, and 4.2. We will see that 1-dimensional orbits in $\operatorname{Sym}^d \mathbb{P}^r$ are reasonably well-behaved, in that the behaviors observed in Examples 4.2.14, 4.2.15, and 4.2.18 do not occur. As in Remark 4.2.12, we will define a partial ordering on decorated trees via chain contractions. The closure $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ of a *minimal* decorated tree $\mathcal{M}_{\tilde{\Gamma}}$ is an open and closed substack of $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)^T$. Theorem 4.5.23 exhibits each of these substacks as a quotient of a specific variety by a finite group. The variety in question is a toric compactification of the moduli space $\mathcal{M}_{0,n}$, and parametrizes chains of rational marked curves glued at nodes. It was introduced by Losev and Manin [47].

4.5.1 Sym^{*d*} \mathbb{P}^r -decorated trees

As above, for a *T*-fixed stable map $f : C \to \operatorname{Sym}^d \mathbb{P}^r$, each component C_{ν} of *C* maps into the fixed locus $(\operatorname{Sym}^d \mathbb{P}^r)^T$, or maps a 1-dimensional orbit closure, with special points (nodes and marks) and ramification points mapping to fixed points. Since *T* acts with isolated fixed points, we refer to the two types of components as *contracted* and *noncontracted*, since those of the first type map to a single point of $\operatorname{Sym}^d \mathbb{P}^r$.

Lemma 4.5.1. Let $(f : C \to \text{Sym}^d \mathbb{P}^r) \in \overline{\mathcal{M}}_{0,2}(\text{Sym}^d \mathbb{P}^r, \beta)$ be a *T*-fixed stable map of degree $\beta > 0$ with irreducible source curve. This is a map to a 1-dimensional orbit closure, which by Proposition 4.4.1 has associated indices i_1 and i_2 . Denote by b_1 and b_2 the two marked points of *C*. Then:

- The associated étale cover $\rho : C' \to C$ from Section 3.1.1 is a disjoint union of rational connected components,
- Under the associated map $f': C' \to \mathbb{P}^r$, each component of C' is either contracted to a T-fixed point of \mathbb{P}^r , or maps to the coordinate line $L_{\{i_1,i_2\}}$,
- On each component C'_{η} of the latter type, $\rho^{-1}(b_1)$ and $\rho^{-1}(b_2)$ are each a (single) fully ramified point, and
- If c_{η} is the degree of $\rho|_{C'_{\eta}} : C'_{\eta} \to C$ and β_{η} is the (coarse) degree of $f'|_{C'_{\eta}} : C'_{\eta} \to L_{\{i_1,i_2\}}$, then the ratio $q := \beta_{\eta}/c_{\eta}$ is independent of η (where η runs over noncontracted components of C').

Proof. The first three statements follow from the fact that C has exactly two orbifold points, and from Proposition 4.4.1. It is straightforward to check that the last statement is equivalent to the fact that the T-action is compatible with the map ρ , i.e. that the action of $\lambda \in T$ corresponds to changing coordinates on C.

Remark 4.5.2. The same statement and proof apply to $\overline{\mathcal{M}}_{0,1}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ and $\overline{\mathcal{M}}_{0,0}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ and in these cases we have a slightly stronger statement: since *C* has at most one orbifold point, it has no nontrivial étale cover, i.e. $C' \cong C \times \{1, \ldots, d\}$.

For $X = \text{Sym}^d \mathbb{P}^r$, we translate Definition 4.3.1 into the following equivalent, and more explicit, definition.

Definition 4.5.3. An *n*-marked (Sym^d \mathbb{P}^r , T)-decorated tree $\tilde{\Gamma} = (\Gamma, \text{Mark}, \text{VEval}, q, \text{Mon})$ is

• A tree Γ ,

- A marking map Mark : $\{1, \ldots, n\} \to V(\Gamma)$,
- A "vertex evaluation map" VEval = $(VEval_0, ..., VEval_r) : V(\Gamma) \rightarrow ZPart(d, r + 1),$ where ZPart(d, r + 1) is the set of nonnegative ordered partitions of d of length r + 1,
- An "edge degree ratio" map $q: E(\Gamma) \to \mathbb{Q}_{>0}$,
- A "monodromy map" Mon that assigns to each i ∈ {1,...,n} an element of the set MultiPart(VEval(Mark(i))) of multipartitions (see Section 1.7) of VEval(Mark(i)), and assigns to a flag (v, e) ∈ F(Γ) an element of MultiPart(VEval(v)),

subject to the conditions:

- If e is an edge of Γ connecting vertices v and v', then VEval_k(v) = VEval_k(v') for all but exactly two 0 ≤ k ≤ r, denoted i^{mov}(v, e) and i^{mov}(v', e), which are defined so that VEval_{i^{mov}(v,e)}(v) - VEval_{i^{mov}(v,e)}(v') > 0,,
- 2. If e is an edge connecting vertices v and v', then for $i \neq i^{\text{mov}}(v, e), i^{\text{mov}}(v', e), \text{Mon}_i(v, e)$ is equal² (as a partition of $\text{VEval}_i(v) = \text{VEval}_i(v')$) to $\text{Mon}_i(v', e)$. Furthermore, we have $\text{Mon}_{i^{\text{mov}}(v,e)}(v', e) \subseteq \text{Mon}_{i^{\text{mov}}(v,e)}(v, e), \text{Mon}_{i^{\text{mov}}(v',e)}(v, e) \subseteq \text{Mon}_{i^{\text{mov}}(v',e)}(v, e)$, and the relation between complements holds:

 $\operatorname{Mon}_{i^{\operatorname{mov}}(v,e)}(v,e) \smallsetminus \operatorname{Mon}_{i^{\operatorname{mov}}(v,e)}(v',e) = \operatorname{Mon}_{i^{\operatorname{mov}}(v',e)}(v',e) \smallsetminus \operatorname{Mon}_{i^{\operatorname{mov}}(v',e)}(v,e) =: \operatorname{Mov}(e),$

- 3. For $\eta \in Mov(e)$, we have $q(e)\eta \in \mathbb{Z}$,
- 4. If $v \in V^1(\Gamma)$ then $Mon(v, e_v)$ is the ones multipartition of VEval(v),
- 5. If $v \in V^{1,1}(\Gamma)$ with v = Mark(i), then $Mon(v, e_v) = Mon(i)$, and
- 6. If $v \in V^2(\Gamma)$, then $\operatorname{Mon}(v, e_v^1) = \operatorname{Mon}(v, e_v^2)$.

Definition 4.5.4. Let $\tilde{\Gamma}$ be an *n*-marked $(\operatorname{Sym}^d \mathbb{P}^r, T)$ -decorated tree $\tilde{\Gamma}$. If $e \in E(\Gamma)_{v,v'}$, then the degree of *e* is $\beta_{edge}(e) := q(e)(\operatorname{VEval}_{i^{\operatorname{mov}}(v,e)}(v) - \operatorname{VEval}_{i^{\operatorname{mov}}(v,e)}(v'))$. The degree of $\tilde{\Gamma}$ is $\beta(\tilde{\Gamma}) = \sum_{e \in E(\Gamma)} \beta_{edge}(e)$.

Remark 4.5.5. We explain why this is equivalent to Definition 4.3.1.

²Here Mon(v, e) is an r + 1-tuple of partitions, which we index by $\{0, \ldots, r\}$, so that $Mon_i(v, e)$ is the (i - 1)st partition.

- Because Γ_{Sym^d P^r,T} has a unique edge between any two adjacent vertices, the map f is determined by the induced map of vertex sets f_v : V(Γ) → V(Γ_{Sym^d P^r}). The map VEval above is exactly this map, since we saw in Section 4.4 that fixed points of Sym^d P^r are in natural bijection with ZPart(d, r + 1). Condition 1 in the definition guarantees that VEval comes from a map of graphs.
- By Lemma 4.5.1, the edge degree map in Definition 4.5.4 agrees with the one in Definition 4.3.1.
- Since T acts with isolated fixed points, vdeg is trivial.
- The sets MultiPart(VEval(Mark(i))) and MultiPart(VEval(v)) are in bijection with the sets of appropriate conjugacy classes, so the monodromy map defined here is the same as that in Definition 4.3.1.

The analog of Proposition 4.2.7 holds:

Proposition 4.5.6. There is a natural map

$$\Psi: (\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^{d} \mathbb{P}^{r},\beta))^{T} \to \operatorname{Trees}_{0,n}(\operatorname{Sym}^{d} \mathbb{P}^{r},\beta).$$

The proof is identical to that of Proposition 4.2.7, with the obvious definition of the map Mon. *Remark* 4.5.7. Conditions 2–6 on a decorated tree eliminate decorated trees for which $\mathcal{M}_{\tilde{\Gamma}}$ is trivially empty:

- Condition 2 is the condition that the monodromies at the two fixed points of a noncontracted component are compatible. (Unlike the others, this is specific to Sym^d P^r.)
- Condition 3 requires that the map $f'|_{C'_{\eta}}$ be a ramified cover with *integer* degree. (See Lemma 4.5.1)
- Condition 4 says that if a point of C mapping to a T-fixed point is not special, then it is not an orbifold point, hence has trivial monodromy.
- Condition 5 says that if a point of C mapping to a T-fixed point is marked, then the monodromy is the same whether this point is viewed as ξ(v, e) or the *i*th marking.
- Condition 6 says that if the monodromy at the two branches of a node must be inverses, when both branches are noncontracted.

These conditions are still not enough to guarantee that $\mathcal{M}_{\tilde{\Gamma}}$ is nonempty. Another, more complicated, condition is needed on the set of monodromies at special points on a contracted component. Namely, there must exist a ramified degree d cover of \mathbb{P}^1 whose ramification profile is given by that set of monodromies. This is equivalent to the nonemptiness of a Hurwitz space \mathcal{H}_v associated to each vertex v with $|Mark^{-1}(v)| + val(v) \geq 3$.

Alternatively, we could include in Definition 4.5.3 the additional data of a connected component of each Hurwitz space \mathcal{H}_v . (We could even define \mathcal{H}_v by convention when $|Mark^{-1}(v)| + val(v) < 3$, in which case Conditions 4– 6 of Definition 4.5.3 would no longer be necessary.) In this case, Theorem 4.5.23 would imply that $\mathcal{M}_{\tilde{\Gamma}}$ is nonempty and *connected*. However, we do not adopt this definition, because the combinatorial description of the connected components of a Hurwitz space is quite complicated.

Notation 4.5.8. Note that the isotropy group at $\xi(v, e)$ (resp. b_i) has order lcm(Mon(v, e)) (resp. lcm(Mon(i))). For brevity we denote this by r(v, e) (resp. r_i).

We now describe how the spaces $\mathcal{M}_{\tilde{\Gamma}}$ fit together via chain contractions (as defined in Remark 4.2.12). In particular, Theorem 4.5.15 tells us that any chain contraction gives rise to a limiting process of stable maps.

Definition 4.5.9. Let $\tilde{\Gamma} \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$, and let $e_1, e_2 \in E(\Gamma)$. We say e_1 and e_2 are *combinable*, and write $e_1 \parallel e_2$, if there exists $v \in V^2(\tilde{\Gamma})$ with $\{e_1, e_2\} = \{e_v^1, e_v^2\}$ and the following hold:

- $q(e_1) = q(e_2)$,
- $i^{\text{mov}}(v_1, e_1) = i^{\text{mov}}(v, e_2)$ and
- $i^{\text{mov}}(v, e_1) = i^{\text{mov}}(v_2, e_2).$

Denote by $\mathcal{P} \subseteq {\binom{E(\Gamma)}{2}}$ the set of pairs $\{\{e_1, e_2\} : e_1 \parallel e_2\}$.

Fix $(f: C \to \operatorname{Sym}^d \mathbb{P}^r) \in \mathcal{M}_{\tilde{\Gamma}}$. We compute that for a flag (v, e) with $e = \{v, v'\}$, the tangent weight at $\xi(v, e)$ along C_e is $w(v, e) = \frac{\alpha_{i^{\operatorname{mov}}(v, e)} - \alpha_{i^{\operatorname{mov}}(v', e)}}{q(e)}$. Thus two edges are combinable if and only if they correspond to components of C separated by a *nonbreaking* node.

Definition 4.5.10. Let $(v, e) \in F(\Gamma)$. We say (v, e) is a *steady flag* if either of the following holds:

- 1. $v \notin V^2(\Gamma)$, or
- 2. $v \in V^2(\Gamma)$ and $\{e_v^1, e_v^2\} \notin \mathcal{P}$.

Otherwise (v, e) is unsteady.



Figure 4.8: Combining two pairs of edges

Definition 4.5.11. Let $\tilde{\Gamma} \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$ and let $e_1 \parallel e_2$ be a pair of combinable edges. We may define a new decorated tree $\text{Comb}(\tilde{\Gamma}, e_1 \parallel e_2) \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$ by **combining** e_1 and e_2 . In other words, we delete the vertex v and the edges e_1 and e_2 , and add an edge $e_{12} = \{v_1, v_2\}$ with $q(e_{12}) = q(e_1) = q(e_2)$, $\text{Mon}(v_1, e_{12}) = \text{Mon}(v_1, e_1)$, and $\text{Mon}(v_2, e_{12}) = \text{Mon}(v_2, e_2)$. (See Figure 4.7.) It is easy to check that $\tilde{\Gamma}(e_1, e_2)$ satisfies the two conditions of a decorated tree, and that $\text{Mov}(e_{12}) = \text{Mov}(e_1) \cup \text{Mov}(e_2)$, and $\text{Mon}(e_{12}) = \text{Mon}(e_1) = \text{Mon}(e_2)$. There is a natural map $\phi_{\{e_1, e_2\}} : E(\Gamma) \to E(\text{Comb}(\tilde{\Gamma}, e_1 \parallel e_2))$ with $\phi_{\{e_1, e_2\}}(e_1) = \phi_{\{e_1, e_2\}}(e_2) = e_{12}$, and $\phi_{\{e_1, e_2\}}(e) = e$ for $e \in E(\Gamma) \setminus \{e_1, e_2\}$.

Proposition 4.5.12. Let $\tilde{\Gamma} \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$, and let $e_1 \parallel e_2$ and $e'_1 \parallel e'_2$ be two distinct pairs of combinable edges of Γ . Then $\phi_{\{e_1,e_2\}}(e'_1) \parallel \phi_{\{e_1,e_2\}}(e'_2)$ as edges of $\text{Comb}(\tilde{\Gamma}, e_1 \parallel e_2)$ and $\phi_{\{e'_1,e'_2\}}(e_1) \parallel \phi_{\{e'_1,e'_2\}}(e_2)$ as edges of $\text{Comb}(\tilde{\Gamma}, e'_1 \parallel e'_2)$. Also, combining pairs commutes, i.e.

$$\operatorname{Comb}(\operatorname{Comb}(\tilde{\Gamma}, e_1 \parallel e_2), e_1' \parallel e_2') \cong \operatorname{Comb}(\operatorname{Comb}(\tilde{\Gamma}, e_1' \parallel e_2'), e_1 \parallel e_2),$$

and this isomorphism identifies the maps $\phi_{\{e_1,e_2\}} \circ \phi_{\{e'_1,e'_2\}}$ and $\phi_{\{e'_1,e'_2\}} \circ \phi_{\{e_1,e_2\}}$.

Proof. There are two cases, pictured in the left side of Figure 4.5.1; either the pairs $e_1 \parallel e_2$ and $e'_1 \parallel e'_2$ share an edge, or they do not. Suppose we are in the first case, i.e. the top line of Figure 4.5.1. By definition of $\phi_{\{e_1,e_2\}}$, the edges $\phi_{\{e_1,e_2\}}(e'_1)$ and $\phi_{\{e_1,e_2\}}(e'_2)$ meet at v' (precisely, at the corresponding vertex in $\text{Comb}(\tilde{\Gamma}, e_1 \parallel e_2)$), and satisfy the three conditions of Definition 4.5.9. Thus $\phi_{\{e_1,e_2\}}(e'_1) \parallel \phi_{\{e_1,e_2\}}(e'_2)$. Similarly $\phi_{\{e'_1,e'_2\}}(e_1) \parallel \phi_{\{e'_1,e'_2\}}(e_2)$. To see that $\text{Comb}(\text{Comb}(\tilde{\Gamma}, e_1 \parallel e_2), e'_1 \parallel e'_2) \cong \text{Comb}(\text{Comb}(\tilde{\Gamma}, e'_1 \parallel e'_2), e_1 \parallel e_2)$, we note that both are obtained from the tree

in Figure 4.5.1 by replacing the three edges shown with a single edge e connecting v_1 to v'_2 . The decorations on this edge are:

- $q(e) := q(e_1) = q(e_2) = q(e'_2)$,
- $\operatorname{Mon}(e) := \operatorname{Mon}(e_1) = \operatorname{Mon}(e_2) = \operatorname{Mon}(e'_2),$

•
$$i^{\text{mov}}(v_1, e) := i^{\text{mov}}(v_1, e_1) = i^{\text{mov}}(v, e_2) = i^{\text{mov}}(v', e'_2)$$
, and

• $i^{\text{mov}}(v'_2, e) := i^{\text{mov}}(v_2, e'_2) = i^{\text{mov}}(v', e_2) = i^{\text{mov}}(v, e_1),$

where the equalities follow from $e_1 \parallel e_2$ and $e_2 \parallel e'_2$. The maps $\phi_{\{e_1,e_2\}} \circ \phi_{\{e'_1,e'_2\}}$ and $\phi_{\{e'_1,e'_2\}} \circ \phi_{\{e_1,e_2\}}$ both send all of e_1 , $e_2 = e'_1$, and e'_2 to e.

The second case (the bottom line of 4.5.1) is a special case of this argument, so we omit it. \Box

Corollary 4.5.13. Let $\tilde{\Gamma} \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$, and let \mathcal{E} be any subset of the set $\mathcal{P}(\tilde{\Gamma})$ of pairs of combinable edges in Γ . Then there is a well-defined tree $\text{Comb}(\tilde{\Gamma}, \mathcal{E}) \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$ obtained by combining all edge pairs in \mathcal{E} , in any order, and a well-defined associated map $\phi_{\mathcal{E}} : E(\Gamma) \to E(\text{Comb}(\tilde{\Gamma}, \mathcal{E}))$. Furthermore, \mathcal{E} is determined by the trees $\tilde{\Gamma}$ and $\text{Comb}(\tilde{\Gamma}, \mathcal{E})$, and the map $\phi_{\mathcal{E}}$.

Proof. The existence statement comes from repeatedly applying Proposition 4.5.12. The uniqueness statement amounts to the fact that if $e_1 \parallel e_2$ is a compatible pair of edges in $\tilde{\Gamma}$, then $\phi_{\mathcal{E}}(e_1) = \phi_{\mathcal{E}}(e_2)$ if and only if $(e_1, e_2) \in \mathcal{E}$. This follows from factoring $\phi_{\mathcal{E}}$ as a sequence of edge combination maps as in Definition 4.5.11.

Corollary 4.5.13 may be restated as follows. Definition 4.5.11 determines a partial order \leq on $\operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$, where $\tilde{\Gamma}' \leq \tilde{\Gamma}$ if $\tilde{\Gamma}'$ can be obtained from $\tilde{\Gamma}$ by combining edges. The Corollary then states that for $\tilde{\Gamma} \in \operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$, there is a natural order-reversing bijection between $\{\tilde{\Gamma}' : \tilde{\Gamma}' \leq \tilde{\Gamma}\}$ and {subsets of $\mathcal{P}(\tilde{\Gamma})$ }, where the latter is partially ordered by inclusion. In particular, associated to $\tilde{\Gamma}$ is a unique *minimal* decorated tree $\operatorname{Comb}(\tilde{\Gamma}, \mathcal{P}(\tilde{\Gamma}))$.

Notation 4.5.14. Denote by $\operatorname{Trees}_{0,n}^{\min}(\operatorname{Sym}^d \mathbb{P}^r, \beta) \subseteq \operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ the set of minimal elements with respect to \leq .

Theorem 4.5.15. Let $\tilde{\Gamma}_0 \in \operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. The closure of $\Psi^{-1}(\tilde{\Gamma}_0)$ is

$$\bigcup_{\substack{\tilde{\Gamma}\in \operatorname{Trees}_{0,n}(\operatorname{Sym}^{d}\mathbb{P}^{r},\beta)\\\tilde{\Gamma}_{0}<\tilde{\Gamma}}}\Psi^{-1}(\tilde{\Gamma})$$

where Ψ is the map from Proposition 4.2.7.

Lemma 4.5.16. Let $\tilde{\Gamma}_0 = v_1 \bullet e \bullet v_2$, where each of v_1 and v_2 contains a single marked point, b_1 and b_2 . Let $f: C \to \operatorname{Sym}^d \mathbb{P}^r$ be in the closure of $\Psi^{-1}(\tilde{\Gamma}_0)$, and let $\rho: C' \to C$ and $f': C' \to \mathbb{P}^r$ be the associated maps. Write C'_η for a noncontracted irreducible component of C', corresponding to $\eta \in \operatorname{Mov}(e) \subseteq \operatorname{Mon}(e)$, as described in Lemma 4.5.1. Denote by $L_e := L_{\{i^{\operatorname{mov}}(v_1,e),i^{\operatorname{mov}}(v_2,e)\}}$ the line in \mathbb{P}^r connecting $P_{i^{\operatorname{mov}}(v_1,e)}$ and $P_{i^{\operatorname{mov}}(v_2,e)}$. Then:

- 1. C and C'_n are nodal chains of rational curves,
- 2. $f'|_{C'_{\eta}}$ maps one irreducible component of C'_{η} to L_e with degree $\beta_{\eta}(e) = q(e) \cdot \eta$ (on coarse moduli spaces), and is fully ramified at the two special points of this component, and
- 3. $f'|_{C'_n}$ contracts all other irreducible components of C'_n to one of the endpoints of L_e .

Proof of Lemma. Let $f : \mathcal{C} \to \mathbb{P}^r$ be a family over S of stable maps whose generic fiber is in $\Psi^{-1}(\tilde{\Gamma}_0)$, and let $s \in S$ such that the fiber over s is the stable map $f : \mathcal{C} \to \operatorname{Sym}^d \mathbb{P}^r$. After an étale base change $\tilde{S} \to S$, \mathcal{C}' is a union of connected components \mathcal{C}'_{η} indexed by $\operatorname{Mon}(e)$, and the maps $\mathcal{C}'_{\eta} \to \mathcal{C}$ have degrees determined by $\operatorname{Mon}(e)$.

Consider the Stein factorization relative to S:

$$\mathcal{C}'_{\eta} \xrightarrow{\mathrm{sf}} \overline{\mathcal{C}'_{\eta}} \xrightarrow{\overline{f'}} \mathbb{P}^r$$

The pullbacks along $\overline{f'}$ of the divisors $P_{i^{\text{mov}}(v_1,e)}$ and $P_{i^{\text{mov}}(v_2,e)}$ on L_e are divisors on $\overline{C'_{\eta}}$, that by the definition of the Stein factorization do not contain a component of any fiber. On a generic fiber, these divisors are each supported on a single point, i.e. $\rho^{-1}(b_1)$ and $\rho^{-1}(b_2)$. Thus on the fiber $\overline{C'_{\eta}}$ over *s*, the divisors are still supported on single points, and $\rho^{-1}(b_1)$ and $\rho^{-1}(b_2)$ each lie above one of these points. (Also, the points are distinct since $\overline{f'}$ is well-defined.)

As any component of $\overline{C'_{\eta}}$ maps surjectively to L_e , the above implies that $\overline{C'_{\eta}}$ is irreducible. This proves claims (2) and (3).

Since f' is T-fixed, any irreducible components of C'_{η} that are contracted by f' map to either $P_{i^{\text{mov}}(v_1,e)}$ or $P_{i^{\text{mov}}(v_2,e)}$, i.e. they lie over either $(\overline{f'})^{-1}(P_{i^{\text{mov}}(v_1,e)})$ or $(\overline{f'})^{-1}(P_{i^{\text{mov}}(v_2,e)})$. Also, all nodes of C'_{η} lie over one of these two points. Since η was arbitrary, this shows that any irreducible component D of C' that is not contracted by f' has at most two special points, where a *special point* here means either a node or one of the points $\rho^{-1}(b_1)$ and $\rho^{-1}(b_2)$. Since $\rho^{-1}(b_1)$ and $\rho^{-1}(b_2)$ lie above distinct points of D, D has exactly two special points.

If C is not a chain, some component has only one special point. By stability there is a component D of $\rho^{-1}(D)$ that is not contracted by f', which contradicts the fact that D has two special points. Thus C is a chain, and it follows that each connected component C'_{η} is a chain. This proves claim (1).



Figure 4.9: A portion of a map in $\overline{\Psi^{-1}(\tilde{\Gamma}_0)}$, with $\eta = 1$ and q(e) = 3

Remark 4.5.17. In summary, the restriction to C'_{η} of a point in $\Psi^{-1}(\tilde{\Gamma}_0)$ may be represented as in Figure 4.5.1 (where despite appearances we mean for the map to L_e to have a single preimage point over each of $P_{i^{\text{mov}}(v_1,e)}$ and $P_{i^{\text{mov}}(v_2,e)}$).

Proof of Theorem 4.5.15. It is sufficient to consider the situation of Lemma 4.5.16. To see this, note that any $\tilde{\Gamma}_0 \in \text{Trees}_{0,n}(\text{Sym}^d \mathbb{P}^r, \beta)$ may be decomposed into subtrees of the form in the Lemma, together with single-vertex trees, glued at marked points. There is a corresponding decomposition of $\Psi^{-1}(\tilde{\Gamma}_0)$ as a product (up to a finite morphism), and this decomposition extends to the closure (see [2], Section 5.2, or [46], Section 9.2). Thus we may treat each factor of the product separately.

First, we show

$$\overline{\Psi^{-1}(\tilde{\Gamma}_0)} \subseteq \bigcup_{\tilde{\Gamma}, \tilde{\Gamma}_0 \leq \tilde{\Gamma}} \Psi^{-1}(\tilde{\Gamma}).$$

Let $(f: C \to \mathbb{P}^r) \in \overline{\Psi^{-1}(\tilde{\Gamma}_0)}$. It follows from Lemma 4.5.16 that $f^{-1}((\operatorname{Sym}^d \mathbb{P}^r)^T)$ is exactly the set of nodes of C, together with the two marked points. By stability, all irreducible components of C are noncontracted. Thus the tree $\Psi(f: C \to \operatorname{Sym}^d \mathbb{P}^r)$ is a chain with a vertex for each node and marked point, and an edge for each irreducible component.

Denote by v_1 and v_2 the leaves of C, such that $v_1 = \{b_1\}$ and $v_2 = \{b_2\}$. For $v \neq v_1, v_2$, we have $\operatorname{Mark}^{-1}(v) = \emptyset$. By claim 2 of Lemma 4.5.16, the degree ratios q(e) are equal for all edges e. By the description of the connected components of C', the partitions $\operatorname{Mon}(e)$ are equal for all e. Finally, deleting an edge e breaks C into two connected components, one containing v_1 and one containing v_2 . Let v be on the component with v_1 , and v' on the component with v_2 , such that $e = \{v, v'\}$. Then from the proof of Lemma 4.5.16, we have $i^{\mathrm{mov}}(v, e) = i^{\mathrm{mov}}(v_1, e_{12})$ and $i^{\mathrm{mov}}(v', e) = i^{\mathrm{mov}}(v_2, e_{12})$. Thus any pair of adjacent edges in $\Psi(f : C \to \operatorname{Sym}^d \mathbb{P}^r)$ is combinable. Combining them all yields $\tilde{\Gamma}_0$, i.e. $\tilde{\Gamma}_0 \leq \Psi(f : C \to \operatorname{Sym}^d \mathbb{P}^r)$.

For the converse, by induction on $\left|E(\tilde{\Gamma})\right| - \left|\tilde{\Gamma}_{0}\right| = \left|E(\tilde{\Gamma})\right| - 1$, it is sufficient to show that



Fix such a tree $\tilde{\Gamma} = v_1 \bullet \underbrace{e_1 \ v \ e_2}_{\bullet} \bullet v_2$, and fix $(f : C \to \operatorname{Sym}^d \mathbb{P}^r) \in \Psi^{-1}(\tilde{\Gamma})$. We will construct a family $f : \mathcal{C} \to \operatorname{Sym}^d \mathbb{P}^r$ over \mathbb{C} whose restriction to $0 \in \mathbb{C}$ is the map $f : C \to \operatorname{Sym}^d \mathbb{P}^r$.

By Lemma 4.5.1 and by representability of $f: C \to \operatorname{Sym}^d \mathbb{P}^r$, the orbifold points and nodes of C have order $\operatorname{lcm}(\operatorname{Mon}(e_1)) = \operatorname{lcm}(\operatorname{Mon}(e_2))$. Thus C is isomorphic to $V(xy) \subseteq [\mathbb{P}^2/\mu_{\operatorname{lcm}(\operatorname{Mon}(e_1))}]$, where \mathbb{P}^2 has coordinates x, y, z, and $\operatorname{lcm}(\operatorname{Mon}(e_1))$ acts by multiplication by inverse roots of unity on the first two coordinates. Define C so that $C_t = V(xy - tz^2)$ for $t \in \mathbb{C}$. Precisely, C is an open subset of $[\mathcal{B}\ell_{[1:0:0],[0:1:0]}\mathbb{P}^2/\mu_{\operatorname{lcm}(\operatorname{Mon}(e_1))}]$.

For $\eta \in Mon(e_1)$ a part, there is an étale quotient map $\tilde{\rho} : [\mathbb{P}^2/\mu_\eta] \to [\mathbb{P}^2/lcm(Mon(e_1))]$. As above, define $(\mathcal{C}'_\eta)_t = V(xy - tz^2) \subseteq [\mathbb{P}^2/\mu_\eta]$.

We must now define a map $\tilde{f}': C'_{\eta} \to \mathbb{P}^r$ for each $\eta \in Mon(e_1)$. As \mathbb{P}^r is a variety, it is enough to define this on coarse moduli spaces. We choose isomorphisms of the fibers $(C'_{\eta})_0$ and C_0 with C'_{η} and C respectively, such that the maps $\tilde{\rho}$ and ρ are identified. Then f' defines a map $\tilde{f}'_0: (C'_{\eta})_0 \to L_{e_1} = L_{e_2}$. (The case where C'_{η} is contracted is trivial, so we assume it is not contracted.) By Lemma 4.5.16, after equivariantly identifying $L_{e_1} \cong \mathbb{P}^1$, \tilde{f}'_0 is given (without loss of generality, on coarse moduli spaces) by

$$[x:0:z] \mapsto [0:1]$$

[0:y:z] $\mapsto [y^{\beta_{\eta}(e_1)}:z^{\beta_{\eta}(e_1)}].$

It remains to extend this to a map $\tilde{f}' : C'_{\eta} \to L_{e_1}$ that is fixed with respect to the *T*-action, i.e. fully ramified over the endpoints of L_{e_1} . We observe that the rational map

$$[x:y:z] \mapsto [y^{\beta_{\eta}(e_1)}:z^{\beta_{\eta}(e_1)}]$$

is regular after blowing up the point [1:0:0]. This defines a map \tilde{f}' as desired. Doing this for all η simultaneously shows that $f: C \to \operatorname{Sym}^d \mathbb{P}^r$ is in $\overline{\Psi^{-1}(\tilde{\Gamma}_0)}$ as desired.

Because $(\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta))^T = \bigcup_{\tilde{\Gamma}} \Psi^{-1}(\tilde{\Gamma})$, we have:

Corollary 4.5.18. Let $\tilde{\Gamma} \in \operatorname{Trees}_{0,n}^{\min}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. Then $\overline{\Psi^{-1}(\tilde{\Gamma})}$ is an open and closed substack of $(\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta))^T$. We denote it by $\overline{\mathcal{M}}_{\tilde{\Gamma}}$.

4.5.2 An explicit description of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$

In order to describe $\overline{\mathcal{M}}_{\tilde{\Gamma}}$, we recall certain moduli spaces of marked curves, studied originally by Losev and Manin [47].

Definition 4.5.19. Let $k \ge 1$, and fix a 2-element set $\{0, \infty\}$. An $0|k|\infty$ -marked Losev-Manin curve is a connected genus zero k + 2-marked nodal curve $(C, b_0, b_1, \ldots, b_k, b_\infty)$, satisfying:

- The irreducible components of C form a chain, with two leaves C_0 and C_{∞} ,
- The points $b_0, b_1, \ldots, b_k, b_\infty$ are smooth points of C, with $b_0 \in C_0$ and $b_\infty \in C_\infty$,
- $b_i \neq 0$ and $b_i \neq \infty$ for i = 1, ..., k (though it is possible that $b_i = b_j$ for $i \neq j$), and
- Each irreducible component of C contains at least one point of b_1, \ldots, b_k .

Theorem 4.5.20 (Losev-Manin [47], Theorems 2.2 and 2.6.3). The moduli space of $0|k|\infty$ -marked Losev-Manin curves $\overline{\mathcal{M}}_{0|k|\infty}$ is a smooth, proper variety, and the natural morphism $\varphi : \overline{\mathcal{M}}_{0,k+2} \to \overline{\mathcal{M}}_{0|k|\infty}$ is birational.

Remark 4.5.21. The spaces $\overline{\mathcal{M}}_{0|k|\infty}$ are special cases of moduli spaces of weighted stable curves, developed by Hassett [31], and Theorem 4.5.20 is a special case of Theorems 2.1 and 4.1 of [31]. Specifically, there is a natural isomorphism $\overline{\mathcal{M}}_{0|k|\infty} \to \overline{\mathcal{M}}_{0,\mathcal{A}}$, where \mathcal{A} is the weight datum $(1, \epsilon, \epsilon, \ldots, \epsilon, 1)$ of length k + 2, for $\epsilon \leq 1/k$.

Definition 4.5.22. Let $s \ge 1$ be an integer. An order *s* orbifold $0|k|\infty$ -marked Losev-Manin curve is a k + 2-marked twisted curve $(C, b_0, b_1, \dots, b_k, b_\infty)$ (in the sense of [51]) whose coarse moduli space is a *k*-marked Losev-Manin curve, such that *C* has orbifold structure only at b_0, b_∞ , and the nodes of *C*, all of which have order *s*.

By standard arguments about twisted curves, the moduli space $\overline{\mathcal{M}}_{0|k|\infty}^s$ of order *s* orbifold *k*-marked Losev-Manin curves has a natural isomorphism $\overline{\mathcal{M}}_{0|k|\infty}^s \to \overline{\mathcal{M}}_{0|k|\infty}$ that comes from taking coarse moduli spaces. The rest of this section proves the following:

Theorem 4.5.23. For a stable vertex v or edge $e = \{v_1, v_2\}$ of a minimal decorated tree $\tilde{\Gamma} = (\Gamma, \text{Mark}, \text{VEval}, q, \text{Mon}) \in \text{Trees}_{0,n}^{\min}(\text{Sym}^d \mathbb{P}^r, \beta)$, we define

$$\overline{\mathcal{M}}_v := \overline{\mathcal{M}}_{0, \overrightarrow{\mathrm{Mon}}(v)}(BS_{\mathrm{VEval}(v)}, 0)$$

$$\overline{\mathcal{M}}_e := \left[\overline{\mathcal{M}}_{v_1 | \text{mov}(e) | v_2}^{\text{lcm}(\text{Mon}(e))} / \left(\prod_{\eta \in \text{Mov}(e)} \mu_{\beta_\eta(e)} \text{ wr } S_e \right) \right],$$

where:

- $\overrightarrow{\mathrm{Mon}}(v)$ is the list of multipartitions $\{\mathrm{Mon}(i)\}_{i\in\mathrm{Mark}^{-1}(v)}\cup\{\mathrm{Mon}(v,e)\}_{(v,e)\in F(\Gamma)}$,
- $\overline{\mathcal{M}}_{v_1|\text{mov}(e)|v_2}^{\text{lcm}(\text{Mon}(e))}$ is the order lcm(Mon(e)) orbifold Losev-Manin space with mov(e) marked points $b_1, \ldots, b_{\text{mov}(e)}$ and labeling set $\{v_1, v_2\}$,
- S_e is the group $C_{\text{Stat}}(e) \times S_{\text{Mov}(e)}$, where $C_{\text{Stat}}(e)$ is the centralizer of any element of the conjugacy class Stat(e) in $\prod_{i=0}^r S_{|\text{Stat}(e)_i|}$, and acts trivially on the Losev-Manin space,
- A generator of $\mu_{\beta_n(e)}$ acts by translating the marked point b_η by $e^{2\pi i/q(e)}$, and
- wr denotes the wreath product.

Then the substack $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ associated to $\tilde{\Gamma}$ is isomorphic (see Remark 4.5.24) to

$$\left[\left(\prod_{v \in V^S(\Gamma)} \overline{\mathcal{M}}_v \times \prod_{e \in E(\Gamma)} \overline{\mathcal{M}}_e \right) \middle/ \operatorname{Aut}(\Gamma) \right].$$
(4.4)

Remark 4.5.24. More precisely, $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ has extra automorphisms coming from gluing at nodes, and is thus a gerbe over (4.4). Gluing of components is fibered over the *rigidified* inertia stack \overline{I} Sym^d \mathbb{P}^r (see [2] or [46]). In particular, for each steady flag (v, e) of $\tilde{\Gamma}$, we get an extra factor of $|C_{\text{VEval}(v)}(\text{Mon}(v, e))|/r(v, e)$ in the fundamental class of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$, where $C_{\text{VEval}}(\text{Mon}(v, e))$ is the centralizer of any element of the conjugacy class Mon(v, e) of $G_{\text{VEval}(v)}$. (We make the usual correction for double counting when $v \in V^2(\Gamma)$.)

Proof of 4.5.23. Using the gluing morphisms, we may write

$$\overline{\mathcal{M}}_{\widetilde{\Gamma}} \cong \left[\left(\prod_{v \in V(\Gamma)} \overline{\mathcal{M}}_{0, \overline{\mathrm{Mon}}(v)}(BS_{\mathrm{VEval}(v)}, 0) \times \prod_{e \in E(\Gamma)} \overline{\mathcal{M}}_{0, \{\mathrm{Mon}(e), \mathrm{Mon}(e)\}}(L_e, \beta(e))^T \right) \middle/ \mathrm{Aut}(\Gamma) \right],$$

We need to show that, for all $e = \{v_1, v_2\} \in E(\Gamma)$, we have

$$\overline{\mathcal{M}}_{0,\{\mathrm{Mon}(e),\mathrm{Mon}(e)\}}(L_e,\beta(e))^T \cong \left[\overline{\mathcal{M}}_{v_1|\mathrm{mov}(e)|v_2}^{\mathrm{lcm}(\mathrm{Mon}(e))} / \left(\prod_{\eta\in\mathrm{Mov}(e)}\mu_{\beta_\eta(e)}\operatorname{wr} S_e\right)\right].$$

(Note that the left hand side is isomorphic to $\overline{\mathcal{M}}_{\tilde{\Gamma}_e}$ for $\tilde{\Gamma}_e = v_1 \bullet e v_2$, where the decorations on $\tilde{\Gamma}_e$ are induced from $\tilde{\Gamma}$. (Here the two vertices are labeled, i.e. $\operatorname{Aut}(\tilde{\Gamma}_e) = 1$.) Write $P_e := P_{(i^{\operatorname{mov}}(v_1,e),i^{\operatorname{mov}}(v_2,e))}$ for the midpoint of L_e . For $(f: C \to \mathbb{P}^r) \in \overline{\mathcal{M}}_{\tilde{\Gamma}_e}$, consider the preimage of P_e under the associated map $f': C' \to \mathbb{P}^r$. By Lemma 4.5.16, C' is a union of connected components C'_{η} for $\eta \in \operatorname{Mon}(e)$, and if $\eta \in \operatorname{Mov}(e)$ then the preimage of P_e on C'_{η} consists of $\beta_{\eta}(e)$ points on the single noncontracted component of C'_{η} . These points are $\mu_{\beta_{\eta}(e)}$ -translates of each other, under the natural action that fixes the two special points.

After a principal $(\prod_{\eta \in Mov(e)} \mu_{\beta_{\eta}(e)} \operatorname{wr} S_e)$ -cover $\overline{\mathcal{M}}_{\Gamma_e} \to \overline{\mathcal{M}}_{\Gamma_e}$, we may fix a labeling of the connected components C'_{η} , and label a distinguished preimage of P_e on C'_{η} for $\eta \in Mov(e)$. (The S_e -cover removes all automorphisms of stable maps induced by automorphisms of the image curve that commute with the monodromy at b_{v_1} and b_{v_2} .) Remembering the images of these distinguished points under ρ yields a nodal chain of rational curves with mov(e) labeled marked points, none of which coincides with b_{v_1} or b_{v_2} . The stability condition for $\overline{\mathcal{M}}_{0,\{Mon(e),Mon(e)\}}(L_e,\beta(e))$ implies that this is a Losev-Manin curve, with orbifold points of order lcm(Mon(e)) at marked points and nodes. This construction works in families, so it defines a map $\overline{\widetilde{\mathcal{M}}}_{\Gamma_e} \to \overline{\mathcal{M}}_{v_1|mov(e)|v_2}^{lcm(Mon(e))}$, which is equivariant by definition with respect to the action of $\prod_{\eta \in Mov(e)} \mu_{\beta_{\eta}(e)}$ wr S_e . This gives a map

$$\Phi: \overline{\mathcal{M}}_{\tilde{\Gamma}_e} \to \left[\overline{\mathcal{M}}_{v_1 | \text{mov}(e) | v_2}^{\text{lcm}(\text{Mon}(e))} / \left(\prod_{\eta \in \text{Mov}(e)} \mu_{\beta_\eta(e)} \operatorname{wr} S_e \right) \right].$$

We now construct an inverse to this map. Let $(C, b_{v_1}, b_1, \ldots, b_{\text{mov}(e)}, b_{v_2}) \in \overline{\mathcal{M}}_{v_1|\text{mov}(e)|v_2}^{\text{lcm}(\text{Mon}(e))}$ be a Losev-Manin curve whose points are indexed by the multiset Mov(e). Fix a curve $C' = \bigcup_{\eta \in \text{Mon}(e)} C'_{\eta}$ with étale maps $\rho_{\eta} : C'_{\eta} \to C$ of degree η . This may be done uniquely up to isomorphism. Also, uniquely up to isomorphism (of C' commuting with $\rho : C' \to C$), for each $\eta \in \text{Mov}(e) \subseteq \text{Mon}(e)$ we may choose a preimage point $b'_{\eta} \in C'_{\eta}$ of the corresponding marked point $b_{\eta} \in C$. Finally, there is a unique map $f' : C' \to \mathbb{P}^r$ that sends:

- C'_{η} to a *T*-fixed point, for $\eta \notin Mov(e)$,
- C'_{η} to L_e with degree $\beta_{\eta}(e)$, with b'_{η} mapping to P_e , $\rho^{-1}(b_{v_1})$ mapping to $P_{i^{\text{mov}}(v_1,e)}$ and $\rho^{-1}(b_{v_2})$ mapping to $P_{i^{\text{mov}}(v_2,e)}$, for $\eta \in \text{Mov}(e)$.

Again, this works in families, and defines a map $\tilde{\Theta} : \overline{\mathcal{M}}_{v_1|\text{mov}(e)|v_2}^{\text{lcm}(\text{Mon}(e))} \to \overline{\mathcal{M}}_{\tilde{\Gamma}_e}$, which we claim is invariant under the action of $\prod_{\eta \in \text{Mov}(e)} \mu_{\beta_\eta(e)} \text{ wr } S_e$. Indeed, acting by $e^{2\pi i/q(e)}$ on b_η translates the preimage b'_{η} by some power of $e^{2\pi i/\beta_{\eta}(e)}$, and commutes with f'. Thus $\tilde{\Theta}$ descends to a map

$$\Theta: \left[\overline{\mathcal{M}}_{v_1 | \mathrm{mov}(e) | v_2}^{\mathrm{lcm}(\mathrm{Mon}(e))} / \left(\prod_{\eta \in \mathrm{Mov}(e)} \mu_{\beta_{\eta}(e)} \operatorname{wr} S_e \right) \right] \to \overline{\mathcal{M}}_{\tilde{\Gamma}_e},$$

which is by construction an inverse to Φ .

Corollary 4.5.25. The $(\prod_{\eta \in Mov(e)} \mu_{\beta_{\eta}(e)} \operatorname{wr} S_e)$ -action on $\overline{\mathcal{M}}_{v_1|mov(e)|v_2}^{\operatorname{lcm}(Mon(e))}$ extends to the universal curve, so we have a universal curve on $\overline{\mathcal{M}}_e$, and by gluing, a universal curve on the left side of (4.4). The isomorphism of 4.5.23 naturally identifies this with the universal curve on $\overline{\mathcal{M}}_{\tilde{\Gamma}}$.

Proof. The first statement is by definition of the action, and the second is immediate from the proof of Theorem 4.5.23. \Box

Remark 4.5.26. Theorem 4.5.23 shows in particular that $\overline{\mathcal{M}}_{\tilde{\Gamma}_e}$ is irreducible, so connected components of $(\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta))^T$ are indexed by minimal decorated trees with the additional data of a connected component of $\overline{\mathcal{M}}_{\operatorname{\overline{Mon}}(v)}(BS_{\operatorname{VEval}(v)}, 0)$ for each v.

Notation 4.5.27. For a special flag $(v, e) \in F(\Gamma)$, we denote by $\psi_v^{\overline{\mathcal{M}}_e}$ the ψ -class on $\overline{\mathcal{M}}_e$ at the point labeled by v. If $v \in V^S(\tilde{\Gamma})$, we denote by $\psi_e^{\overline{\mathcal{M}}_v}$ the ψ -class on $\overline{\mathcal{M}}_v$ at the marked point $\xi(v, e)$. We use the same notation for the $\overline{\psi}$ -classes.

4.6 The virtual normal bundle and virtual fundamental class of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$

In this section we compute the Euler class of the virtual normal bundle to $\overline{\mathcal{M}}_{\tilde{\Gamma}}$, and show that the virtual fundamental class of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ is equal to its fundamental class. Many of the arguments are "classical," and we refer the reader to [46] for these.

In this section we fix $\tilde{\Gamma} \in \operatorname{Trees}_{0,n}^{\min}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. Let $\pi : \mathcal{C} \to \overline{\mathcal{M}}_{\tilde{\Gamma}}$ and $\rho : \mathcal{C}' \to \mathcal{C}$ denote the universal curve and universal étale cover, respectively:



By a standard argument (see [46], Section 9.3), we have an exact sequence of *T*-equivariant sheaves on $\overline{\mathcal{M}}_{0,n+1}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ giving the perfect obstruction theory³

$$0 \to \operatorname{Aut}(\mathcal{C}) \to R^0 \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r) \to \operatorname{Def}(\mathcal{C}, f) \to$$

$$\to \operatorname{Def}(\mathcal{C}) \to R^1 \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r) \to \operatorname{Obs}(\mathcal{C}, f) \to 0,$$
(4.5)

where $\operatorname{Aut}(\mathcal{C})$ (resp. $\operatorname{Def}(\mathcal{C})$) is the sheaf on $\overline{\mathcal{M}}_{0,n+1}(\operatorname{Sym}^d \mathbb{P}^r)$ of infinitesimal automorphisms (resp. deformations) of the marked source curve \mathcal{C} . (See [46] for rigorous definitions.) For

³We will always use the notation in (4.5) for higher direct image sheaves, writing e.g. $R^i \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ instead of $R^i \pi_* f^*T \operatorname{Sym}^d \mathbb{P}^r$. This is because we will restrict π to various substacks of \mathcal{C} , and wish to avoid renaming maps.

 $(f: C \to \operatorname{Sym}^d \mathbb{P}^r) \in \overline{\mathcal{M}}_{\tilde{\Gamma}}$, we also have a normalization exact sequence computing the fibers of the middle terms:

$$0 \to H^{0}(C, f^{*}T \operatorname{Sym}^{d} \mathbb{P}^{r}) \to \bigoplus_{\nu} H^{0}(C_{\nu}, f^{*}T \operatorname{Sym}^{d} \mathbb{P}^{r}) \to \bigoplus_{\xi} H^{0}(\xi, f^{*}T \operatorname{Sym}^{d} \mathbb{P}^{r}) \to (4.6)$$
$$\to H^{1}(C, f^{*}T \operatorname{Sym}^{d} \mathbb{P}^{r}) \to \bigoplus_{\nu} H^{1}(C_{\nu}, f^{*}T \operatorname{Sym}^{d} \mathbb{P}^{r}) \to 0,$$

where ν runs over the set of irreducible components of C, and ξ runs over nodes of C. The sequences (4.5) and (4.6) each split as direct sums of two exact sequences: the *T*-fixed part and the *T*-moving part. We use the notations $\operatorname{Aut}(\mathcal{C})^{\operatorname{fix}}$ and $\operatorname{Aut}(\mathcal{C})^{\operatorname{mov}}$ (and similar) to denote the *T*-fixed subsheaf or subspace and its *T*-invariant complement. By definition (see [29]), the *Euler class of the virtual normal bundle* $e_T(N_{\tilde{\Gamma}}^{\operatorname{vir}})$ is

$$\frac{e_T(\operatorname{Def}(\mathcal{C}, f)^{\operatorname{mov}})}{e_T(\operatorname{Obs}(\mathcal{C}, f)^{\operatorname{mov}})} = \frac{e_T(\operatorname{Def}(\mathcal{C})^{\operatorname{mov}})e_T(R^0\pi_*(\mathcal{C}, f^*T\operatorname{Sym}^d\mathbb{P}^r)^{\operatorname{mov}})}{e_T(\operatorname{Aut}(\mathcal{C})^{\operatorname{mov}})e_T(R^1\pi_*(\mathcal{C}, f^*T\operatorname{Sym}^d\mathbb{P}^r)^{\operatorname{mov}})} \in H_T^*(\overline{\mathcal{M}}_{\tilde{\Gamma}}, \mathbb{Q}), \quad (4.7)$$

and the *virtual fundamental class* $[\overline{\mathcal{M}}_{\tilde{\Gamma}}]^{\text{vir}}$ of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ is $e_T(\text{Obs}(\mathcal{C}, f)^{\text{fix}})$. We compute the various terms of (4.5) and (4.6) one by one. It is convenient to compute by pulling back to the canonical $\text{Aut}(\Gamma)$ -cover $\overline{\mathcal{M}}_{\tilde{\Gamma}}^{\text{rig}}$ of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$, so that the correspondence between C and $\tilde{\Gamma}$ is more concrete.

The sheaves $Aut(\mathcal{C})$ and $Def(\mathcal{C})$. In the toric case, from [46] Section 9.3.1, we have

$$e_T(\operatorname{Aut}(\mathcal{C})^{\operatorname{mov}}) = \prod_{v \in V^1(\tilde{\Gamma})} e_T(T_{\xi(v,e_v)}C) = \prod_{v \in V^1(\tilde{\Gamma})} \psi_v^{\overline{\mathcal{M}}_{e_v}}.$$
(4.8)

The same argument and answer apply here, using (Theorem 4.5.15 and) the observation that combining edges gives a natural identification of $V^1(\tilde{\Gamma})$. Briefly, moving automorphisms come from noncontracted components with only one special point, and correspond to vector fields on such a component that are nonvanishing at the nonspecial *T*-fixed point.

Similarly, in the toric case [46] Section 9.3.2 gives

$$e_{T}(\operatorname{Def}(\mathcal{C})) = \left(\prod_{\substack{v \in V^{2}(\tilde{\Gamma})\\(v, e_{v}^{1}) \text{ steady}}} (-\psi_{v}^{\overline{\mathcal{M}}_{e_{v}^{1}}} - \psi_{v}^{\overline{\mathcal{M}}_{e_{v}^{2}}})\right) \left(\prod_{\substack{(v, e) \in F(\Gamma)\\v \in V^{S}(\tilde{\Gamma})}} (-\psi_{e}^{\overline{\mathcal{M}}_{v}} - \psi_{v}^{\overline{\mathcal{M}}_{e}})\right).$$
(4.9)

This is again correct in our case. The factors in (4.9) come from smoothing nodes. (Classically, the deformation space of a node is the tensor product of the tangents spaces to the two branches.) Therefore the observation we need is that breaking nodes correspond to steady flags.

The bundles $R^0 \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ and $R^1 \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$. We use the sequence (4.6). The computation is very similar to the original one by Kontsevich [41] (and the orbifold computations of Johnson [36] and Liu [46]), but requires some care due to the edge moduli spaces.

Because normalization does not commute with base change, (4.6) only computes fibers of $R^i \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$. However, normalization of breaking nodes does commute with base change on $\overline{\mathcal{M}}_{\tilde{\Gamma}}^{\operatorname{rig}}$, because the set of breaking nodes is canonically identified for any two points of $\overline{\mathcal{M}}_{\tilde{\Gamma}}^{\operatorname{rig}}$. Thus we have the sequence

$$0 \to R^{0}\pi_{*}(\mathcal{C}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \to \bigoplus_{\underline{\nu}} R^{0}\pi_{*}(\mathcal{C}_{\underline{\nu}}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \to \bigoplus_{\xi} R^{0}\pi_{*}(\xi, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \to R^{1}\pi_{*}(\mathcal{C}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \to \bigoplus_{\underline{\nu}} R^{1}\pi_{*}(\mathcal{C}_{\underline{\nu}}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \to 0,$$

$$(4.10)$$

where $\underline{\nu}$ runs over maximal subcurves of C containing only nonbreaking nodes, and ξ runs over breaking nodes. ($C_{\underline{\nu}}$ may contain a single branch of a breaking node, but not both branches.) Observe that either $C_{\underline{\nu}}$ is contracted by f, or each fiber $C_{\underline{\nu}}$ of $C_{\underline{\nu}}$ contains only noncontracted components.

By Section 3.1.2, we have

$$R^{i}\pi_{*}(\mathcal{C}_{\underline{\nu}}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) = R^{i}\pi_{*}(\mathcal{C}_{\underline{\nu}}, \rho_{*}(f')^{*}T\mathbb{P}^{r}) = R^{i}(\pi \circ \rho)_{*}(\mathcal{C}_{\underline{\nu}}', (f')^{*}T\mathbb{P}^{r}).$$

(The second equality follows from the fact that ρ is étale, hence ρ_* is exact.) After an étale base change, we may distinguish the connected components of fibers of $\mathcal{C}'_{\underline{\nu}} \to \overline{\mathcal{M}}_{\tilde{\Gamma}}^{\mathrm{rig}}$. In other words, we may write

$$\mathcal{C}'_{\underline{\nu}} = \bigsqcup_{\eta} \mathcal{C}'_{\underline{\nu},\eta},$$

where $C'_{\nu,\eta}$ has connected fibers. Then

$$R^{i}\pi_{*}(\mathcal{C}_{\underline{\nu}}',(f')^{*}T\mathbb{P}^{r}) = \bigoplus_{\eta} R^{i}(\pi \circ \rho)_{*}(\mathcal{C}_{\underline{\nu},\eta}',(f')^{*}T\mathbb{P}^{r}).$$
(4.11)

If $\mathcal{C}_{\underline{\nu}} = \mathcal{C}_v$ is contracted, then $(f')^* T \mathbb{P}^r$ is trivial on $\mathcal{C}'_{\nu,\eta}$. Thus we have

$$R^{i}(\pi \circ \rho)_{*}(\mathcal{C}'_{\underline{\nu},\eta}, (f')^{*}T\mathbb{P}^{r}) \cong R^{i}(\pi \circ \rho)_{*}(\mathcal{C}'_{\underline{\nu},\eta}, \mathcal{O}_{\mathcal{C}'_{\underline{\nu},\eta}}) \otimes T_{P_{i(\eta)}}\mathbb{P}^{r},$$

where as usual we write $P_{i(\eta)}$ for $f'(\mathcal{C}'_{\nu,\eta})$. In particular,

$$R^{0}\pi_{*}(\mathcal{C}_{v}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})^{\operatorname{fix}} = R^{1}\pi_{*}(\mathcal{C}_{\underline{\nu}}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})^{\operatorname{fix}} = 0.$$
(4.12)

The bundle $R^1\pi_*(\mathcal{C}_v, f^*T\operatorname{Sym}^d \mathbb{P}^r)^{\mathrm{mov}}$ is nontrivial, and is isomorphic to a Hurwitz-Hodge bundle (see [46], Section 7.5). However, note that $e_T(R\pi_*(\mathcal{C}_v, f^*T\operatorname{Sym}^d \mathbb{P}^r))$ is the inverse of the twisting class from (2.6). We will use this fact in Section 5.1 in our characterization of $\mathcal{L}_{\mathrm{Sym}^d \mathbb{P}^r}$, and in Section 6.2.3 to apply the orbifold quantum Riemann-Roch theorem.

Similarly for a breaking node $\xi(v, e)$, we have

$$R^{0}\pi_{*}(\xi(v,e), f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})^{\operatorname{fix}} = 0$$

$$R^{0}\pi_{*}(\xi(v,e), f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})^{\operatorname{mov}} = T_{(\operatorname{VEval}(v),\operatorname{Mon}(v,e))}I\operatorname{Sym}^{d}\mathbb{P}^{r} = \bigoplus_{\eta\in\operatorname{Mon}(v,e)}T_{P_{i(\eta)}}\mathbb{P}^{r}.$$
(4.13)

Suppose $C_{\underline{\nu}}$ is not contracted. The components $C'_{\underline{\nu},\eta}$ are in bijection with Mon(e), where e is the edge of $\tilde{\Gamma}$ corresponding to $C_{\underline{\nu}}$.) First, we argue that $R^1(\pi \circ \rho)_*(C'_{\underline{\nu},\eta}, (f')^*T\mathbb{P}^r)$ vanishes for all η . The normalization exact sequence for a fiber $C'_{\underline{\nu},\eta}$ reads:

$$\begin{split} 0 &\to H^0(C'_{\underline{\nu},\eta}, (f')^*T\mathbb{P}^r) \to \bigoplus_{\nu \in \underline{\nu}} H^0(C'_{\nu,\eta}, (f')^*T\mathbb{P}^r) \to \bigoplus_{\xi} H^0(\xi, (f')^*T\mathbb{P}^r) \to \\ &\to H^1(C'_{\underline{\nu},\eta}, (f')^*T\mathbb{P}^r) \to \bigoplus_{\nu \in \underline{\nu}} H^1(C'_{\nu,\eta}, (f')^*T\mathbb{P}^r) \to 0, \end{split}$$

where we also denote by $\underline{\nu}$ the set indexing irreducible components C_{ν} of $C_{\underline{\nu}}$ (equivalently, irreducible components $C'_{\nu,\eta}$ of $C_{\underline{\nu},\eta}$). For each $\nu \in \underline{\nu}$, we have

$$H^{1}(C_{\nu}, (f')^{*}T\mathbb{P}^{r}) = 0$$
(4.14)

by convexity of \mathbb{P}^r . We claim that the map

$$\bigoplus_{\nu \in \underline{\nu}} H^0(C'_{\nu,\eta}, (f')^* T\mathbb{P}^r) \to \bigoplus_{\xi} H^0(\xi, (f')^* T\mathbb{P}^r)$$

is surjective, so that $H^1(C'_{\underline{\nu},\eta}, (f')^*T\mathbb{P}^r) = 0$. (The map takes the difference of the sections on the two branches of a node.) If $C'_{\underline{\nu},\eta}$ has a component $C'_{\nu_0,\eta}$ not contracted by f', there is at most one, by Lemma 4.5.16. On any other component $C'_{\nu,\eta}$, we have $(f')^*T\mathbb{P}^r \cong \mathcal{O}_{C'_{\nu,\eta}} \otimes T\mathbb{P}^r$, i.e. $H^0(C'_{\nu,\eta}, \mathcal{O}_{C'_{\nu,\eta}} \otimes T\mathbb{P}^r) \cong T\mathbb{P}^r$. Fix an arbitrary section $s \in H^0(C'_{\nu_0,\eta}, (f')^*T\mathbb{P}^r)$. Then "working outward" from $C'_{\nu_0,\theta}$ shows that the map is surjective. The case where f' contracts $C'_{\underline{\nu},\eta}$ is similar and simpler.

Next, we compute $R^0(\pi \circ \rho)_*(\mathcal{C}'_{\nu,\eta}, (f')^*T\mathbb{P}^r)$. If $\mathcal{C}'_{\nu,\eta}$ is contracted, $(f')^*T\mathbb{P}^r$ is trivial and we
have

$$R^{0}(\pi \circ \rho)_{*}(\mathcal{C}'_{\underline{\nu},\eta}, (f')^{*}T\mathbb{P}^{r}) \cong T\mathbb{P}^{r} \otimes \mathcal{O}_{\overline{\mathcal{M}}_{\overline{\Gamma}}^{\mathrm{rig}}}$$

by properness of $\pi \circ \rho$. Suppose $\mathcal{C}'_{\underline{\nu},\eta}$ is not contracted. Consider the Stein factorization of $f'|_{\mathcal{C}'_{\underline{\nu},\eta}}$ relative to $\pi \circ \rho$:



If $(f : C \to \operatorname{Sym}^d \mathbb{P}^r)$ is in the dense open substack $\Psi^{-1}(\tilde{\Gamma}) \subseteq \overline{\mathcal{M}}_{\tilde{\Gamma}}^{\operatorname{rig}}$, then $C_{\underline{\nu}}$ is irreducible, hence so is $C'_{\underline{\nu},\eta}$. This, with the fact that $\mathcal{C}'_{\underline{\nu},\eta}$ is not contracted, implies that sf is birational. By the projection formula for coherent sheaves,

$$(\pi \circ \rho)_* (f')^* T \mathbb{P}^r = (\pi \circ \rho)_* \mathrm{sf}^* (f'')^* T \mathbb{P}^r$$

= $(\overline{\pi \circ \rho})_* \mathrm{sf}_* \mathrm{sf}^* (f'')^* T \mathbb{P}^r$
= $(\overline{\pi \circ \rho})_* ((f'')^* T \mathbb{P}^r \otimes \mathrm{sf}_* \mathcal{O}_{C'_{\underline{\nu},\eta}})$
= $(\overline{\pi \circ \rho})_* (f'')^* T \mathbb{P}^r.$

After an étale base change on $\overline{\mathcal{M}}_{\tilde{\Gamma}}^{\text{rig}}$, the map f'' trivializes the family $\overline{\mathcal{C}'_{\underline{\nu},\eta}}$. Thus the vector bundle $R^0(\overline{\pi \circ \rho})_*(\overline{\mathcal{C}'_{\nu,\eta}}, (f'')^*T\mathbb{P}^r)$ is trivial.

Calculation of the *T*-weights of this vector bundle is identical to Kontsevich's calculation in Section 3.3.4 of [41], which uses the Euler sequence on \mathbb{P}^r . The weights are

$$\frac{A}{\beta_{\eta}(e)}\alpha_{i^{\mathrm{mov}}(v_{1},e)} + \frac{B}{\beta_{\eta}(e)}\alpha_{i^{\mathrm{mov}}(v_{2},e)} - \alpha_{i}$$

where $0 \leq A, B \leq \beta_{\eta}(e), A + B = \beta_{\eta}(e)$, and $i \in \{0, \ldots, r\}$. Note that this is zero exactly when A = 0 and $i = i^{\text{mov}}(v_2, e)$, or B = 0 and $i = i^{\text{mov}}(v_1, e)$. (These factors contribute to $e_T(R^0(\overline{\pi \circ \rho})_*(\overline{\mathcal{C}'_{\underline{\nu},\eta}}, (f'')^*T\mathbb{P}^r)^{\text{fix}})$.) The Euler class $e_T(R^0(\overline{\pi \circ \rho})_*(\overline{\mathcal{C}'_{\underline{\nu}}}, (f'')^*T\mathbb{P}^r)^{\text{mov}})$ for $\underline{\nu}$ noncontracted is thus

$$\left(\prod_{\substack{\eta \in \text{Stat}(e)\\i \neq i(\eta)}} (\alpha_{i(\eta)} - \alpha_i)\right) \left(\prod_{\substack{\eta \in \text{Mov}(e)\\0 \leq i \leq r\\(A,i) \neq (0,i^{\text{mov}}(v_2,e))\\(B,i) \neq (0,i^{\text{mov}}(v_1,e))}} \left(\frac{A}{\beta_{\eta}(e)} \alpha_{i^{\text{mov}}(v_1,e)} + \frac{B}{\beta_{\eta}(e)} \alpha_{i^{\text{mov}}(v_2,e)} - \alpha_i\right)\right)$$

$$(4.15)$$

Summary. We collect the arguments of this section in the following two statements.

Proposition 4.6.1. For any minimal decorated tree $\tilde{\Gamma}$, $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ is smooth, and the virtual fundamental class is equal to the fundamental class.

Proposition 4.6.2. The equivariant Euler class $e_T(N_{\overline{\mathcal{M}}_{\tilde{\Gamma}}}^{\text{vir}})$ of the virtual normal bundle to $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ is

$$\begin{split} &\left(\frac{\prod_{v\in V^{2}(\tilde{\Gamma})}(-\psi_{v}^{\overline{\mathcal{M}}_{e_{v}^{1}}}-\psi_{v}^{\overline{\mathcal{M}}_{e_{v}^{2}}})\prod_{(v,e)\in F(\Gamma)}(-\psi_{e}^{\overline{\mathcal{M}}_{v}}-\psi_{v}^{\overline{\mathcal{M}}_{e}})}{v\in V^{S}(\tilde{\Gamma})}\right) \\ & \cdot \prod_{e\in E(\Gamma)} \left(\left(\prod_{\eta\in \text{Stat}(e)}(\alpha_{i(\eta)}-\alpha_{i})\right) \\ & \cdot \left(\prod_{\eta\in \text{Mov}(e)}\prod_{\substack{A+B=\beta\eta(e)\\0\leq t\leq r\\(A,i)\neq (0,i^{\min v}(v_{2},e))\\(B,i)\neq (0,i^{\min v}(v_{1},e))}}\left(\frac{A}{\beta\eta(e)}\alpha_{i^{\min v}(v_{2},e)}+\frac{B}{\beta\eta(e)}\alpha_{i^{\min v}(v_{2},e)}-\alpha_{i}\right)\right)\right) \\ \\ & \cdot \frac{\prod_{v\in V^{1}(\Gamma)\cup V^{1,1}(\Gamma)\cup V^{2}(\Gamma)}e_{T}(T_{(\text{VEval}(v),\text{Mon}(v))I}\operatorname{Sym}^{d}\mathbb{P}^{r})}{\prod_{(v,e)\in F(\Gamma)}e_{T}(T_{(\text{VEval}(v),\text{Mon}(v))I}\operatorname{Sym}^{d}\mathbb{P}^{r})} \\ & \cdot \prod_{v\in V^{S}(\tilde{\Gamma})}e_{T}(R\pi_{*}(C_{v},f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})). \end{split}$$

Proof of Proposition 4.6.1. Recall from Theorem 4.1.2 that the virtual fundamental class of $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ is

obtained from the fixed part of the perfect obstruction theory on $\overline{\mathcal{M}}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. By (4.13), the fixed part of $\bigoplus_{\xi} R^0 \pi_*(\xi, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ is zero. Thus by (4.10), we have

$$R^{1}\pi_{*}(\mathcal{C}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}) \cong \bigoplus_{\underline{\nu}} R^{1}\pi_{*}(\mathcal{C}_{\underline{\nu}}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r}).$$

But we showed, in (4.12) and (4.14), that $\bigoplus_{\underline{\nu}} R^1 \pi_*(\mathcal{C}_{\underline{\nu}}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ has no fixed part. Thus $R^1 \pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ has no fixed part. By Proposition 5.5 of [6], the Proposition follows. (Smoothness already followed easily from Theorem 4.5.23.)

Proof of Proposition 4.6.2. The first line is the contribution from $Def(\mathcal{C})^{mov}$ and $Aut(\mathcal{C})^{mov}$, from (4.8) and (4.9). The second line is the contribution to $R\pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ from noncontracted components $\mathcal{C}_{\underline{\nu}}$, as in (4.15) and (4.14). The third line is the contribution of breaking nodes to $R\pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$, from (4.13). (The numerator corrects for the fact that $F(\Gamma)$ overcounts the breaking nodes.) The last line is the contribution to $R\pi_*(\mathcal{C}, f^*T \operatorname{Sym}^d \mathbb{P}^r)$ from contracted components \mathcal{C}_{ν} , by definition.

CHAPTER 5

The Givental cone is characterized by recursion relations

5.1 Characterization of the Givental cone $\mathcal{L}_{Svm^d \mathbb{P}^r}$

In this section, we apply the results of Sections 4.5 and 4.6 to give a criterion (Theorem 5.1.4) that exactly determines whether a given power series lies on the Givental cone $\mathcal{L}_{Svm^d \mathbb{P}^r}$.

Definition 5.1.1. Fix $(\mu, \sigma) \in (I \operatorname{Sym}^d \mathbb{P}^r)^T$. Let $\Upsilon(\mu, \sigma) \subseteq \operatorname{Trees}_{0,2}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ be the set of 1-edge decorated trees $\tilde{\kappa} = v_1 \bullet \underbrace{e} \bullet v_2$, with marking set $\{n + 1, \bullet\}$, with $\operatorname{Mark}(n + 1) = v_1$ and $\operatorname{Mark}(\bullet) = v_2$, such that $\mu = \operatorname{VEval}(v_1)$ and $\sigma = \operatorname{Mon}(v_1, e)$.

Notation 5.1.2. For $\tilde{\kappa} \in \Upsilon(\mu, \sigma)$ as in Definition 5.1.1, we write:

- $q(\tilde{\kappa}) := q(e)$,
- $\operatorname{Mov}(\tilde{\kappa}) := \operatorname{Mov}(e),$
- $\operatorname{mov}(\tilde{\kappa}) := \operatorname{mov}(e),$
- $\operatorname{Stat}(\tilde{\kappa}) := \operatorname{Stat}(e),$
- $i_1^{\text{mov}}(\tilde{\kappa}) := i^{\text{mov}}(v_1, e),$
- $i_2^{\mathrm{mov}}(\tilde{\kappa}) := i^{\mathrm{mov}}(v_2, e),$
- $\mu'(\tilde{\kappa}) := \operatorname{VEval}(v_2),$
- $\sigma'(\tilde{\kappa}) := \operatorname{Mon}(v_2, e)$, and
- $r(\tilde{\kappa}) := r(v_1, e) = r(v_2, e) = r_{n+1}$.
- $w(\tilde{\kappa}) := w(v_1, e).$

We compute:

$$w(v_1, e) = \frac{\alpha_{i_1^{\text{mov}}(\tilde{\kappa})} - \alpha_{i_2^{\text{mov}}(\tilde{\kappa})}}{q(\tilde{\kappa})} \in H^2_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q}).$$

Similarly to the notations ψ and $\overline{\psi}$, we write $\overline{w} = r(\tilde{\kappa})w$.

Definition 5.1.3. Let $\tilde{\kappa} \in \Upsilon(\mu, \sigma)$ and let $a \in \mathbb{Z}_{>0}$ We define the recursion coefficient as

$$\mathbf{RC}(\tilde{\kappa}, a) = \frac{(-1)^{\mathrm{mov}(\tilde{\kappa})-a}}{q(\tilde{\kappa})^{\mathrm{mov}(\tilde{\kappa})}} \binom{\sigma_{i_{1}^{\mathrm{mov}}(\tilde{\kappa})}}{\mathrm{Mov}(\tilde{\kappa})} \binom{\mathrm{mov}(\tilde{\kappa})-1}{a-1} \cdot \frac{1}{\prod_{\eta \in \mathrm{Mov}(\tilde{\kappa})} \prod_{\substack{1 \le B \le \beta_{\eta}(e) \\ 0 \le i \le r \\ (B,i) \ne (\beta_{\eta}(e), i_{2}^{\mathrm{mov}}(\tilde{\kappa}))}} \frac{\left(\frac{\beta_{\eta}(e)-B}{\beta_{\eta}(e)}\alpha_{i_{1}^{\mathrm{mov}}(\tilde{\kappa})} + \frac{B}{\beta_{\eta}(e)}\alpha_{i_{2}^{\mathrm{mov}}(\tilde{\kappa})} - \alpha_{i}\right)},$$

where $\binom{\sigma_{i_1}^{\mathrm{mov}(\tilde{\kappa})}}{\mathrm{Mov}(\tilde{\kappa})}$ is the number of ways of choosing $\mathrm{Mov}(\tilde{\kappa})$ as a subpartition of $\sigma_{i_1}^{\mathrm{mov}(\tilde{\kappa})}$ with specified parts.

The following theorem and its proof are adapted from Theorem 41 of [18], which in turn is adapted from Theorem 2 of [8].

Theorem 5.1.4. Let \mathbf{f} be an element of $\mathcal{H}[[x]]$ such that $\mathbf{f}|_{Q=x=0} = -1z$, where 1 denotes the fundamental class of $\operatorname{Sym}^d \mathbb{P}^r \subseteq I \operatorname{Sym}^d \mathbb{P}^r$. Then \mathbf{f} is a $\Lambda^T_{\operatorname{nov}}[[x]]$ -valued point of $\mathcal{L}_{\operatorname{Sym}^d \mathbb{P}^r}$ if and only if for each T-fixed point $(\mu, \sigma) \in I \operatorname{Sym}^d \mathbb{P}^r$, the following three conditions hold:

(I) The restriction $\mathbf{f}_{(\mu,\sigma)}$ along $\iota_{(\mu,\sigma)} : (\mu,\sigma) \hookrightarrow I \operatorname{Sym}^d \mathbb{P}^r$ is a power series in Q and x, such that each coefficient of this power series is an element of $H^*_{T,\operatorname{loc}}(\bullet, \mathbb{Q})(z)$. Each coefficient is regular in z except for possible poles at $z = 0, z = \infty$, and

$$z \in \{\overline{w}(\tilde{\kappa}) : \tilde{\kappa} \in \Upsilon(\mu, \sigma)\}.$$

(II) The Laurent coefficients of $\mathbf{f}_{(\mu,\sigma)}$ at the poles (other than z = 0 and $z = \infty$) satisfy the recursion relation:

$$\operatorname{Laur}(\mathbf{f}_{\mu,\sigma}, (\overline{w} - z)^{-a}) = \sum_{\substack{\tilde{\kappa} \in \Upsilon_{(\mu,\sigma)} \\ \overline{w}(\tilde{\kappa}) = \overline{w} \\ \operatorname{Mov}(\tilde{\kappa}) \ge a}} Q^{\beta(\tilde{\kappa})} \operatorname{\mathbf{RC}}(\tilde{\kappa}, a) \operatorname{Laur}(\mathbf{f}_{(\mu'(\tilde{\kappa}), \sigma'(\tilde{\kappa}))}, (\overline{w} - z)^{\operatorname{mov}(\tilde{\kappa}) - a})$$
(5.1)

for a > 0, and

(III) The restriction \mathbf{f}_{μ} along $\iota_{\mu} : I\mu \hookrightarrow I \operatorname{Sym}^{d} \mathbb{P}^{r}$ is a $\Lambda^{T}_{nov}[[x]]$ -valued point of $\mathcal{L}^{T \operatorname{Sym}^{d} \mathbb{P}^{r}}_{\mu}$, the twisted Givental cone from Section 2.6.

Remark 5.1.5. In (III), Λ_{nov}^T is the equivariant Novikov ring associated to $\operatorname{Sym}^d \mathbb{P}^r$, not μ . In other words, $\Lambda_{nov}^T[[x]] = H^*_{CR,T,\operatorname{loc}}(\mu, \mathbb{Q})[[x, Q]].$

Remark 5.1.6. The major difference between Theorem 5.1.4 and the corresponding theorems in [18] and [8] is that condition (II) gives a recursive relation for *all* negative-exponent Laurent coefficients at $z = w(\tilde{\kappa})$, in terms of positive-exponent ones. In [18] and [8], only stacks with isolated 1-dimensional *T*-orbits are considered. Thus the poles at $z = w(\tilde{\kappa})$ are simple, and a recursive relation is given for their residues.

Proof. For convenience, we abbreviate $\overline{\mathcal{M}}_{0,n+1,\beta}$: $\overline{\mathcal{M}}_{0,n+1}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$. Let **f** be a $\Lambda_{\operatorname{nov}}^T[[x]]$ -valued point of $\mathcal{L}_{\operatorname{Sym}^d \mathbb{P}^r}$. By definition, we can write

$$\begin{aligned} \mathbf{f} &= -1z + \mathbf{t}(z) + \sum_{\substack{n,\beta \ge 0\\\beta > 0 \text{ or } n \ge 2}} \sum_{\phi} \frac{Q^{\beta}}{n!} \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{\gamma_{\phi}}{-z - \overline{\psi}} \right\rangle_{0,n+1,\beta}^{\operatorname{Sym}^{d} \mathbb{P}^{r}, T} \gamma^{\phi} \\ &= -1z + \mathbf{t}(z) + \sum_{\substack{n,\beta \ge 0\\\beta > 0 \text{ or } n \ge 2}} \frac{Q^{\beta}}{n!} (\operatorname{ev}_{n+1})_{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}) \cup \frac{1}{-z - \overline{\psi}} \cap [\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}} \right) \end{aligned}$$

for $\mathbf{t}(z) \in \mathcal{H}_+[[x]]$ with $\mathbf{t}|_{Q=x=0} = 0$. The restriction $\mathbf{f}_{(\mu,\sigma)}$ is then

$$-\delta_{\sigma=\{1,\dots,1\}}z + \iota_{(\mu,\sigma)}^{*}\mathbf{t}(z) + \sum_{\substack{n,\beta\geq 0\\\beta>0 \text{ or }n\geq 2}} \frac{Q^{\beta}}{n!}\iota_{(\mu,\sigma)}^{*}\left((\mathrm{ev}_{n+1})_{*}\left(\prod_{j=1}^{n}\mathrm{ev}_{j}^{*}\,\mathbf{t}(\overline{\psi})\cup\frac{1}{-z-\overline{\psi}}\cap[\overline{\mathcal{M}}_{0,n+1,\beta}]^{\mathrm{vir}}\right)\right).$$

Using the projection formula, we write

$$\begin{split} \iota_{(\mu,\sigma)}^{*} \left((\operatorname{ev}_{n+1})_{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}_{j}) \cup \frac{1}{-z - \overline{\psi}_{n+1}} \cap [\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}} \right) \right) \\ &= |C_{\mu}(\sigma)| \int_{\operatorname{Sym}^{d} \mathbb{P}^{r}} (\iota_{(\mu,\sigma)})_{*} \iota_{(\mu,\sigma)}^{*} \left((\operatorname{ev}_{n+1})_{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}_{j}) \cup \frac{1}{-z - \overline{\psi}_{n+1}} \cap [\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}} \right) \right) \\ &= |C_{\mu}(\sigma)| \int_{\operatorname{Sym}^{d} \mathbb{P}^{r}} \left[(\mu, \sigma) \right] \cup \left((\operatorname{ev}_{n+1})_{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}_{j}) \cup \frac{1}{-z - \overline{\psi}_{n+1}} \cap [\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}} \right) \right) \\ &= |C_{\mu}(\sigma)| \int_{\operatorname{Sym}^{d} \mathbb{P}^{r}} \left((\operatorname{ev}_{n+1})_{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}_{j}) \cup \frac{\operatorname{ev}^{*}([(\mu, \sigma)])}{-z - \overline{\psi}_{n+1}} \cap [\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}} \right) \right) \\ &= |C_{\mu}(\sigma)| \int_{[\overline{\mathcal{M}}_{0,n+1,\beta}]^{\operatorname{vir}}} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}_{j}) \cup \frac{\operatorname{ev}^{*}([(\mu, \sigma)])}{-z - \overline{\psi}_{n+1}} \right) \\ &= |C_{\mu}(\sigma)| \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{[(\mu, \sigma)]}{-z - \overline{\psi}} \right\rangle^{\operatorname{Sym}^{d} \mathbb{P}^{r}, T}_{0,n+1,\beta}. \end{split}$$

The first equality uses the identification of $\int_{\operatorname{Sym}^d \mathbb{P}^r} \circ \iota_{(\mu,\sigma)}$ with the identity map $\operatorname{Spec} \mathbb{C} \to \operatorname{Spec} \mathbb{C}$ on coarse moduli spaces, and the factor $|C_{\mu}(\sigma)|$ corrects for the isotropy at $(\mu, \sigma) \in I \operatorname{Sym}^d \mathbb{P}^r$. (Recall that $C_{\mu}(\sigma)$ denotes the centralizer of any element of σ in G_{μ} .) In summary,

$$\mathbf{f}_{(\mu,\sigma)} = -\delta_{\sigma=\{1,\dots,1\}} z + \mathbf{t}_{(\mu,\sigma)}(z)$$

$$+ \sum_{\substack{n,\beta \ge 0\\\beta > 0 \text{ or } n \ge 2}} \frac{|C_{\mu}(\sigma)| Q^{\beta}}{n!} \left\langle \mathbf{t}(\overline{\psi}),\dots,\mathbf{t}(\overline{\psi}), \frac{[(\mu,\sigma)]}{-z - \overline{\psi}} \right\rangle_{0,n+1,\beta}^{\operatorname{Sym}^{d} \mathbb{P}^{r},T},$$
(5.2)

where $\mathbf{t}_{(\mu,\sigma)}(z) := \iota_{(\mu,\sigma)}^* \mathbf{t}(z)$. Now we calculate (5.2) by virtual torus localization (see Theorem 4.1.2). Namely, we may write

$$|C_{\mu}(\sigma)| \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{[(\mu, \sigma)]}{-z - \overline{\psi}} \right\rangle_{0, n+1, \beta}^{\operatorname{Sym}^{d} \mathbb{P}^{r}, T} = \sum_{\tilde{\Gamma} \in \operatorname{Trees}_{0, n+1}^{\min}(\operatorname{Sym}^{d} \mathbb{P}^{r}, \beta)} \operatorname{Contr}_{(\mu, \sigma)}(\tilde{\Gamma}).$$
(5.3)

We can partition $\operatorname{Trees}_{0,n+1}^{\min}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ into three subsets:

- (i) $\tilde{\Gamma}$ such that $(\operatorname{VEval}(\operatorname{Mark}(b_{n+1})), \operatorname{Mon}(b_{n+1})) \neq (\mu, \sigma),$
- (ii) $\tilde{\Gamma}$ such that $(\text{VEval}(\text{Mark}(b_{n+1})), \text{Mon}(b_{n+1})) = (\mu, \sigma)$ and $\text{Mark}(b_{n+1}) \in V^{1,1}(\Gamma)$, and
- (iii) $\tilde{\Gamma}$ such that $(\operatorname{VEval}(\operatorname{Mark}(b_{n+1})), \operatorname{Mon}(b_{n+1})) = (\mu, \sigma)$ and $\operatorname{Mark}(b_{n+1}) \in V^S(\Gamma)$.

In some literature, e.g. [13], decorated trees of type (ii) are called *recursion type* and those of type (iii) are called *initial type*. Let $v_1 := Mark(b_{n+1})$ be the vertex containing the point b_{n+1} . (We will see below, however, that in our setup both types are used recursively.)

For a tree $\tilde{\Gamma}$ of type (i), the restriction $\operatorname{ev}_{n+1}^*([(\mu, \sigma)])$ vanishes, hence $\operatorname{Contr}_{(\mu,\sigma)}(\tilde{\Gamma}) = 0$. For this reason, we may simplify our notation, and write $\operatorname{Contr}(\tilde{\Gamma}) := \operatorname{Contr}_{(\mu,\sigma)}(\tilde{\Gamma})$, where $\mu = \operatorname{VEval}(\operatorname{Mark}(b_{n+1}))$ and $\sigma = \operatorname{Mon}(\operatorname{Mark}(b_{n+1}))$.

If $\tilde{\Gamma}$ is a tree of type (iii), then by Theorem 4.5.23 and Corollary 4.5.25, $\overline{\psi}_{n+1}$ is pulled back from $\overline{\mathcal{M}}_{0,\overline{\mathrm{Mon}}(v_1)}(BG_{\mu},0)$, where G_{μ} is the isotropy group of μ . Since this stack parametrizes maps that factor through the fixed point μ , the action of T is trivial, hence

$$H^*_{T,\mathrm{loc}}(\overline{\mathcal{M}}_{0,\overline{\mathrm{Mon}}(v_1)}(BG_{\mu},0),\mathbb{Q}) \cong H^*(\overline{\mathcal{M}}_{0,\overline{\mathrm{Mon}}(v_1)}(BG_{\mu},0),\mathbb{Q}) \otimes H^*_{T,\mathrm{loc}}(\bullet,\mathbb{Q}).$$

In particular, $\overline{\psi}_{n+1}$ is nilpotent. It follows that $\operatorname{Contr}(\tilde{\Gamma})$ is a polynomial in z^{-1} , hence has a pole only at z = 0.

Finally, let $\tilde{\Gamma}$ be a tree of type (ii). By (4.1), we have

$$\operatorname{Contr}(\tilde{\Gamma}) = |C_{\mu}(\sigma)| \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}}]'} \frac{1}{e_T(N_{\tilde{\Gamma}}^{\operatorname{vir}})} \iota_{\tilde{\Gamma}}^* \left(\prod_{j=1}^n \operatorname{ev}_j^* \mathbf{t}(\overline{\psi}) \cup \frac{\operatorname{ev}_{n+1}^*[(\mu,\sigma)]}{-z - \overline{\psi}_{n+1}} \right),$$
(5.4)

where $\iota_{\tilde{\Gamma}}$ is the inclusion $\overline{\mathcal{M}}_{\tilde{\Gamma}} \hookrightarrow \overline{\mathcal{M}}_{0,n+1}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$, and $[\overline{\mathcal{M}}_{\tilde{\Gamma}}]'$ denotes the fundamental class, weighted by factors from Remark 4.5.24. Note that $\operatorname{ev}_{n+1} \circ \iota_{\tilde{\Gamma}}$ factors through (μ, σ) , hence $\iota_{\tilde{\Gamma}}^* \operatorname{ev}_{n+1}^*[(\mu, \sigma)]$ is the weight $e_T(T_{(\mu,\sigma)}I\operatorname{Sym}^d \mathbb{P}^r)$.

Then $\tilde{\Gamma}$ has a decorated subtree $\tilde{\kappa} \in \Upsilon(\mu, \sigma)$, obtained by removing all edges except for $e := e_{v_1}$ (and necessary vertices), and all marked points except b_{n+1} . Let $\tilde{\Gamma} \smallsetminus \tilde{\kappa}$ denote the tree obtained by *pruning* $\tilde{\kappa}$. That is, $\tilde{\Gamma} \smallsetminus \tilde{\kappa} \in \operatorname{Trees}_{0,n+1}^{\min}(\operatorname{Sym}^d \mathbb{P}^r, \beta - \beta(\tilde{\kappa}))$ is defined by $V(\Gamma \smallsetminus \kappa) = V(\Gamma) \smallsetminus \{v_1\}, E(\Gamma \smallsetminus \kappa) = E(\Gamma) \smallsetminus e$, and decorations Mark, VEval, q, and Mon are unchanged, except $\operatorname{Mark}(b_{n+1}) := v_2$, where v_2 is the common vertex of $\tilde{\kappa}$ and $\tilde{\Gamma} \smallsetminus \tilde{\kappa}$. Observe that an automorphism of Γ fixes b_{n+1} , and therefore fixes e, so we have $\operatorname{Aut}(\tilde{\Gamma}) = \operatorname{Aut}(\tilde{\Gamma} \smallsetminus \tilde{\kappa})$ and may write

$$\overline{\mathcal{M}}_{\tilde{\Gamma}} \cong \overline{\mathcal{M}}_e \times \overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}$$

by Theorem 4.5.23, up to a gerbe from Remark 4.5.24. We factor the *T*-equivariant map $\overline{\mathcal{M}}_{\tilde{\Gamma}} \rightarrow$

Spec \mathbb{C} through the second projection, i.e. we integrate over $\overline{\mathcal{M}}_e$:

$$\operatorname{Contr}(\widetilde{\Gamma}) = \frac{|C_{\mu}(\sigma)| \left| C_{\mu'(\widetilde{\kappa})}(\sigma'(\widetilde{\kappa})) \right|}{r(\widetilde{\kappa})} \\ \cdot \int_{[\overline{\mathcal{M}}_{\widetilde{\Gamma} \smallsetminus \widetilde{\kappa}}]'} \left(\int_{\overline{\mathcal{M}}_{e}} \frac{e_{T}(T_{(\mu,\sigma)}I\operatorname{Sym}^{d}\mathbb{P}^{r})}{e_{T}(N_{\widetilde{\Gamma}}^{\operatorname{vir}})} \iota_{\widetilde{\Gamma}}^{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi}) \cup \frac{1}{-z - \overline{\psi}_{n+1}} \right) \right).$$

The factor $|C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa}))|/r(\tilde{\kappa})$ is the correction from Remark 4.5.24. From Proposition 4.6.2, we may write

$$\frac{e_T(T_{(\mu,\sigma)}I\operatorname{Sym}^d\mathbb{P}^r)}{e_T(N_{\tilde{\Gamma}}^{\operatorname{vir}})} = \frac{1}{W} \cdot \frac{e_T(T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}I\operatorname{Sym}^d\mathbb{P}^r)}{e(N_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}^{\operatorname{vir}})(-\psi_e^{\overline{\mathcal{M}}_{v_2}} - \psi_{v_2}^{\overline{\mathcal{M}}_e})},$$
(5.5)

where

$$W = \frac{\prod_{\substack{i \neq i(\eta) \\ e_T(T_{(\mu,\sigma)}I \operatorname{Sym}^d \mathbb{P}^r)}}{\sum_{\substack{i \in \operatorname{Mov}(\tilde{\kappa}) \\ 0 \leq i \leq r \\ (A,i) \neq (0,i^{\operatorname{mov}}(v_2,e)) \\ (B,i) \neq (0,i^{\operatorname{mov}}(v_1,e))}} \prod_{\substack{A+B=\beta_\eta(\tilde{\kappa}) \\ 0 \leq i \leq r \\ (A,i) \neq (0,i^{\operatorname{mov}}(v_2,e)) \\ (B,i) \neq (0,i^{\operatorname{mov}}(v_1,e))}}} \left(\frac{A}{\beta_\eta(\tilde{\kappa})} \alpha_{i^{\operatorname{mov}}(v_1,e)} + \frac{B}{\beta_\eta(\tilde{\kappa})} \alpha_{i^{\operatorname{mov}}(v_2,e)} - \alpha_i}\right)$$
$$= \prod_{\substack{\eta \in \operatorname{Mov}(\tilde{\kappa}) \\ 1 \leq B \leq \beta_\eta(\tilde{\kappa}) \\ 0 \leq i \leq r \\ (B,i) \neq (\beta_\eta(\tilde{\kappa}),i^{\operatorname{mov}}(v_2,e))}}} \left(\frac{\beta_\eta(\tilde{\kappa}) - B}{\beta_\eta(\tilde{\kappa})} \alpha_{i^{\operatorname{mov}}(v_1,e)} + \frac{B}{\beta_\eta(\tilde{\kappa})} \alpha_{i^{\operatorname{mov}}(v_2,e)} - \alpha_i}\right) \in H^*_{T,\operatorname{loc}}(\operatorname{Spec} \mathbb{C}, \mathbb{Q})$$

Note that the cancellation in the last step removes the terms with $B \neq 0$ in the product, and that

1/W is the product appearing $\mathbf{RC}(\tilde{\kappa}, a)$. To avoid confusion, we write $\overline{\psi}_{n+1}^{\tilde{\Gamma}}$ (resp. $\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}$) for the $\overline{\psi}$ -class at the (n + 1)st marked point on $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ (resp. $\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}$), recalling that on $\tilde{\Gamma} \smallsetminus \tilde{\kappa}$ we defined $\operatorname{Mark}(b_{n+1}) = v_2$. We also have $\iota_{\tilde{\Gamma}}^* \overline{\psi}_{n+1}^{\tilde{\Gamma}} = \overline{\psi}_{v_1}^{\overline{\mathcal{M}}_e}$. The *T*-weight on $\overline{\psi}_{v_1}^{\overline{\mathcal{M}}_e}$ is $-\overline{w}(\tilde{\kappa})$ (see Notation 5.1.2), so we have

$$\overline{\psi}_{v_1}^{\overline{\mathcal{M}}_e} = \overline{\psi}_{v_1}^{\mathrm{ne}} - \overline{w}(\tilde{\kappa}) \in H_T^*(\overline{\mathcal{M}}_{\tilde{\Gamma}}, \mathbb{Q}) \cong H^*(\overline{\mathcal{M}}_{\tilde{\Gamma}}, \mathbb{Q}) \otimes H_T^*(\operatorname{Spec} \mathbb{C}, \mathbb{Q}),$$
(5.6)

where $\overline{\psi}_{v_1}^{ne}$ denotes the nonequivariant $\overline{\psi}$ -class. Similarly $\overline{\psi}_{v_2}^{\overline{\mathcal{M}}_e} = \overline{\psi}_{v_2}^{ne} + \overline{w}(\tilde{\kappa})$. Then since $\iota_{\tilde{\Gamma}}^* \operatorname{ev}_j^* \mathbf{t}(\overline{\psi})$

is pulled back from $\overline{\mathcal{M}}_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}$,

$$\operatorname{Contr}(\tilde{\Gamma}) = \frac{|C_{\mu}(\sigma)| \left| C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa})) \right|}{r(\tilde{\kappa})} \frac{e_T(T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}I\operatorname{Sym}^d \mathbb{P}^r)}{W} \\ \cdot \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}]'} \left(\frac{\iota_{\tilde{\Gamma}}^*\left(\prod_{j=1}^n \operatorname{ev}_j^* \mathbf{t}(\overline{\psi})\right)}{e_T(N_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}^{\operatorname{vir}})} \int_{\overline{\mathcal{M}}_e} \frac{1}{(-\psi_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}} - \psi_{v_2}^{\operatorname{ne}} - w(\tilde{\kappa}))} \frac{1}{(-z - \overline{\psi}_{v_1}^{\operatorname{ne}} + \overline{w}(\tilde{\kappa}))} \right).$$

We compute the last integral using the fact that $w(\tilde{\kappa})$ is invertible, the following lemma, which follows immediately from Lemma 2.3 of [48]:

Lemma 5.1.7. Let $\psi_{0,LM}$ and $\psi_{\infty,LM}$ denote the tautological cotangent classes at b_0 and b_{∞} on $\overline{\mathcal{M}}_{0|k|\infty}$. The pullbacks $\varphi^*\psi_{0,LM}$ and $\varphi^*\psi_{\infty,LM}$ along the reduction morphism $\overline{\mathcal{M}}_{0,k+2} \to \overline{\mathcal{M}}_{0|k|\infty}$ are the cotangent classes ψ_0 and ψ_{∞} , respectively.

This says that we may integrate on $\overline{\mathcal{M}}_{k+2}$ (with an appropriate constant factor) instead of $\overline{\mathcal{M}}_e$. We use $r(\tilde{\kappa})(-\psi_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}-\psi_{v_2}^{\mathrm{ne}}-w(\tilde{\kappa})) = \overline{\psi}_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}-\overline{\psi}_{v_2}^{\mathrm{ne}}-\overline{w}(\tilde{\kappa})$. It is well-known (see e.g. [40], Lemma 1.5.1) that

$$\int_{\overline{\mathcal{M}}_{0,k}} \psi_1^m \psi_2^{k-3-m} = \binom{k-3}{m}.$$

By Lemma 5.1.7, this identity holds on $\overline{\mathcal{M}}_{0|k|\infty}$ also. Thus:

$$\int_{\overline{\mathcal{M}}_{e}} \frac{1}{\left(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - \overline{\psi}_{v_{2}}^{\mathrm{ne}} - \overline{w}(\tilde{\kappa})\right)} \frac{1}{\left(-z - \overline{\psi}_{v_{1}}^{\mathrm{ne}} + \overline{w}(\tilde{\kappa})\right)} \\
= \frac{1}{|S_{e}| \prod_{\eta \in \mathrm{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \int_{\overline{\mathcal{M}}_{v_{1}|\mathrm{mov}(\tilde{\kappa})|v_{2}}} \left(\sum_{m_{1},m_{2}=0}^{\infty} \frac{(\overline{\psi}_{v_{2}})^{m_{1}}}{\left(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - \overline{w}(\tilde{\kappa})\right)^{m_{1}+1}} \frac{(\overline{\psi}_{v_{1}})^{m_{2}}}{(-z + \overline{w}(\tilde{\kappa}))^{m_{2}+1}} \right) \\
= \frac{1}{|S_{e}| \prod_{\eta \in \mathrm{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \sum_{m_{1}+m_{2}=\mathrm{mov}(\tilde{\kappa})-1} \frac{\binom{\mathrm{mov}(\tilde{\kappa})-1}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - \overline{w}(\tilde{\kappa}))^{m_{1}+1}(-z + \overline{w}(\tilde{\kappa}))^{m_{2}+1}} (5.7) \\
= \frac{1}{|S_{e}| \prod_{\eta \in \mathrm{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \frac{(-z - \overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}})^{\mathrm{mov}(\tilde{\kappa})-1}}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \setminus \tilde{\kappa}} - \overline{w}(\tilde{\kappa}))^{\mathrm{mov}(\tilde{\kappa})}}.$$

(The last inequality is gotten in the backwards direction by writing the numerator as $((-z + \overline{w}(\tilde{\kappa})) +$

 $(-\overline{\psi}_{n+1}^{\widetilde{\Gamma}\smallsetminus\widetilde{\kappa}}-\overline{w}(\widetilde{\kappa})))^{\mathrm{mov}(\widetilde{\kappa})-1}$ and expanding.) We have

$$\operatorname{Contr}(\tilde{\Gamma}) = \frac{|C_{\mu}(\sigma)| |C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa}))|}{|S_{e}| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \frac{e_{T}(T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}I\operatorname{Sym}^{d}\mathbb{P}^{r})}{W(-z+\overline{w}(\tilde{\kappa}))^{\operatorname{mov}(\tilde{\kappa})}} \qquad (5.8)$$
$$\cdot \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}]'} \left(\frac{\iota_{\tilde{\Gamma}}^{*}\left(\prod_{j=1}^{n}\operatorname{ev}_{j}^{*}\mathbf{t}(\overline{\psi})\right)}{e_{T}(N_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}^{\operatorname{vir}})} \frac{(-z-\overline{\psi}_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}})^{\operatorname{mov}(\tilde{\kappa})-1}}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}-\overline{w}(\tilde{\kappa}))^{\operatorname{mov}(\tilde{\kappa})}} \right).$$

For fixed β_0 , and n_0 , from (5.3), the coefficient of $Q^{\beta_0}x^{n_0}$ in $\mathbf{f}_{(\mu,\sigma)}$ only has contributions from $\tilde{\Gamma} \in \operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ for $\beta + n \leq \beta_0 + n_0$. This is because $\mathbf{t}(z) \in \langle x, Q \rangle$, so if $\mathcal{H}[[x]]$ is graded by giving Q and x degree 1, then the (n, β) term in (5.2) has degree at least $n + \beta$. In particular, $\bigcup_{\beta+n\leq\beta_0+n_0}\operatorname{Trees}_{0,n}(\operatorname{Sym}^d \mathbb{P}^r, \beta)$ is a finite set. Thus (5.3) and (5.8) realize the contribution to such a coefficient from trees of type (ii) as a finite sum of rational functions with poles at the weights $\tilde{\kappa}$. Together with the analysis above for types (i) and (iii), this proves that $\mathbf{f}_{(\mu,\sigma)}$ satisfies condition (I) of the Theorem.

We consider the Laurent coefficient Laur(Contr($\tilde{\Gamma}$), $(\overline{w}-z)^{-a}$). By (5.8), Laur(Contr($\tilde{\Gamma}$), $(\overline{w}-z)^{-a}$) is zero if $\overline{w} \neq \overline{w}(\tilde{\kappa})$, or if mov($\tilde{\kappa}$) < a. Otherwise,

$$\begin{aligned} \operatorname{Laur}(\operatorname{Contr}(\tilde{\Gamma}),(\overline{w}-z)^{-a}) &= \frac{1}{(\operatorname{mov}(\tilde{\kappa})-a)!} \left(\frac{d^{\operatorname{mov}(\tilde{\kappa})-a}}{d(\overline{w}(\tilde{\kappa})-z)^{\operatorname{mov}(\tilde{\kappa})-a}} (\overline{w}(\tilde{\kappa})-z)^{\operatorname{mov}(\tilde{\kappa})} \operatorname{Contr}(\tilde{\Gamma}) \right) \Big|_{z \mapsto \overline{w}(\tilde{\kappa})} \\ &= \frac{(-1)^{\operatorname{mov}(\tilde{\kappa})-a} |C_{\mu}(\sigma)| \left| C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa})) \right| \binom{\operatorname{mov}(\tilde{\kappa})-1}{a-1}}{W |S_{e}| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \\ &\cdot \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}]'} \left(\frac{\iota_{\tilde{\Gamma}}^{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\overline{\psi})\right)}{e_{T}(N_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}^{\operatorname{vir}})} \frac{e_{T}(T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}I\operatorname{Sym}^{d} \mathbb{P}^{r})}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - \overline{w}(\tilde{\kappa}))^{\operatorname{mov}(\tilde{\kappa})-a+1}} \right). \end{aligned}$$

Now, summing over all $\tilde{\Gamma}$ of type (ii) with associated subtree $\tilde{\kappa}$ yields

$$\frac{(-1)^{\operatorname{mov}(\tilde{\kappa})-a} |C_{\mu}(\sigma)| |C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa}))| \binom{\operatorname{mov}(\tilde{\kappa})-1}{a-1}}{W |S_{e}| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} \left\langle \mathbf{t}(\overline{\psi}), \dots, \mathbf{t}(\overline{\psi}), \frac{[(\mu'(\tilde{\kappa}), \sigma'(\tilde{\kappa}))]}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - \overline{w}(\tilde{\kappa}))^{\operatorname{mov}(\tilde{\kappa})-a+1}} \right\rangle_{0,n+1,\beta-\beta(\tilde{\kappa})}^{\operatorname{Sym}^{d} \mathbb{P}^{r}, T} .$$
(5.9)

On the other hand, the coefficient $\operatorname{Laur}(\mathbf{f}_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))},(\overline{w}(\tilde{\kappa})-z)^{\operatorname{mov}(\tilde{\kappa})-a})$ is

$$\sum_{\substack{n,\beta\geq 0\\\beta>0 \text{ or } n\geq 2}} \frac{\left|C_{\mu'(\tilde{\kappa})}(\sigma'(\tilde{\kappa}))\right| Q^{\beta}}{n!} \left\langle \mathbf{t}(\overline{\psi}),\dots,\mathbf{t}(\overline{\psi}), \frac{\left[(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))\right]}{(-\overline{\psi}_{n+1}^{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}-\overline{w}(\tilde{\kappa}))^{\operatorname{mov}(\tilde{\kappa})-a+1}} \right\rangle_{0,n+1,\beta}^{\operatorname{Sym}^{d}\mathbb{P}^{r},T}$$
(5.10)

We compute $\frac{|C_{\mu}(\sigma)|}{|S_e|\prod_{\eta\in Mov(\tilde{\kappa})}\beta_{\eta}(\tilde{\kappa})}$ explicitly:

$$\begin{aligned} |C_{\mu}(\sigma)| &= |S_{\sigma}| \prod_{\eta \in \sigma} \eta \\ |S_{e}| &= |C_{\operatorname{Stat}}(\tilde{\kappa})| \left| S_{\operatorname{Mov}(\tilde{\kappa})} \right| = \left| S_{\operatorname{Stat}(\tilde{\kappa})} \right| \left| S_{\operatorname{Mov}(\tilde{\kappa})} \right| \prod_{\eta \in \operatorname{Stat}(\tilde{\kappa})} \eta \\ \frac{|C_{\mu}(\sigma)|}{|S_{e}| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} &= \frac{|S_{\sigma}| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \eta}{|S_{\operatorname{Stat}(\tilde{\kappa})}| \left| S_{\operatorname{Mov}(\tilde{\kappa})} \right| \prod_{\eta \in \operatorname{Mov}(\tilde{\kappa})} \beta_{\eta}(\tilde{\kappa})} = \frac{1}{q(\tilde{\kappa})^{\operatorname{mov}(\tilde{\kappa})}} \binom{\sigma}{\operatorname{Mov}(\tilde{\kappa})} \end{aligned}$$

With (5.9) and (5.10), this proves (II). Note that the contribution from all graphs of type (ii) (and the term $\mathbf{t}_{(\mu,\sigma)}(z)$) is

$$\boldsymbol{\tau}_{(\mu,\sigma)}(z) := \mathbf{t}_{(\mu,\sigma)}(z) + \sum_{\substack{\tilde{\kappa} \in \Upsilon(\mu,\sigma) \\ a \le \operatorname{mov}(\tilde{\kappa})}} \frac{Q^{\beta(\tilde{\kappa})} \operatorname{\mathbf{RC}}(\tilde{\kappa},a)}{(\overline{w}(\tilde{\kappa}) - z)^a} \operatorname{Laur}(\mathbf{f}_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}, (\overline{w}(\tilde{\kappa}) - z)^{\operatorname{mov}(\tilde{\kappa}) - a}).$$
(5.11)

The proof of condition (III) is identical to that of condition (C3) in [18], and we reproduce the argument here for convenience.

Consider a decorated tree $\tilde{\Gamma}$ of type (iii). We write $v := \operatorname{Mark}(b_{n+1}) \in V^S(\Gamma)$. The marked points of $\overline{\mathcal{M}}_v$ correspond to (1) elements of $\operatorname{Mark}^{-1}(v)$, and (2) edges $e \in E(\Gamma)_v$. To e is associated a maximal subtree Γ_e containing v, with $E(\Gamma_e, v) = e$. We decorate Γ_e so that $\operatorname{Mark}^{-1}(v) = b$, and the rest of the decorations inherited from $\tilde{\Gamma}$. We will then write $\operatorname{Contr}(\tilde{\Gamma})$ in terms of $\operatorname{Contr}(\tilde{\Gamma}_e)$ for $e \in E(\Gamma)_v$, and integrals over the vertex moduli space $\overline{\mathcal{M}}_v$.

We apply (5.4) again. After an étale base change $\overline{\mathcal{M}}_{\tilde{\Gamma}} \to \overline{\mathcal{M}}_{\tilde{\Gamma}}$, we may label the subtrees $\tilde{\Gamma}_e$. (Write M for the degree of this base change.) We then write $\widetilde{\overline{\mathcal{M}}}_{\tilde{\Gamma}} \cong \overline{\mathcal{M}}_v \times \prod_{e \in E(\Gamma)_v} \overline{\mathcal{M}}_{\tilde{\Gamma}_e}$. Now we again apply Proposition 4.6.2, to see that

$$\frac{1}{e_T(N_{\tilde{\Gamma}}^{\mathrm{vir}})} = e_T^{-1}(R\pi_*(C_v, f^*T\operatorname{Sym}^d \mathbb{P}^r)) \prod_{e \in E(\Gamma)_v} \frac{r(v, e)e_T(T_{(\mu, \operatorname{Mon}(v, e))}I\operatorname{Sym}^d \mathbb{P}^r)}{(-\overline{\psi}_e^{\overline{\mathcal{M}}_v} - \overline{\psi}_v^{\overline{\mathcal{M}}_e})e_T(N_{\tilde{\Gamma}_e}^{\mathrm{vir}})}$$

Observe that $\frac{e_T(T_{(\mu,\operatorname{Mon}(v,e))}I\operatorname{Sym}^d\mathbb{P}^r)}{(-\overline{\psi}_e^{\overline{\mathcal{M}}_v} - \overline{\psi}_v^{\overline{\mathcal{M}}_v})}$ is the insertion at b in $\operatorname{Contr}(\widetilde{\Gamma}_e)|_{z\mapsto \overline{\psi}_e^{\overline{\mathcal{M}}_v}}$. Thus

$$\operatorname{Contr}(\widetilde{\Gamma}) = \frac{1}{M} \int_{\overline{\mathcal{M}}_{v}} \left(\prod_{e \in E(\Gamma)_{v}} \left| C_{\mu}(\sigma) \right| Q^{\beta(\widetilde{\Gamma}_{e})} \operatorname{Contr}(\widetilde{\Gamma}_{e}) \right|_{z \mapsto \overline{\psi}_{e}^{\overline{\mathcal{M}}_{v}}} \right) \cup \left(\prod_{b_{i} \in \operatorname{Mark}^{-1}(v)} \mathbf{t}(\overline{\psi}) \right)$$
$$\cup \frac{e_{T}(T_{(\mu,\sigma)}I\operatorname{Sym}^{d}\mathbb{P}^{r})}{-z - \overline{\psi}_{n+1}} \cup e_{T}^{-1}(R\pi_{*}(C_{v}, f^{*}T\operatorname{Sym}^{d}\mathbb{P}^{r})).$$

This is almost a twisted Gromov-Witten invariant of VEval(v), but not quite, since there are restrictions on the monodromies at the marked points. Summing over $\tilde{\Gamma}_e$ for a single e, with everything else fixed, gives the insertion $\tau_{(\mu,Mon(v,e))}(\psi)$, where the initial term comes from replacing $\tilde{\Gamma}_e$ with a marked point. Thus summing over all σ , and over all $\tilde{\Gamma}$ of type (iii), gives

$$\sum_{m=2}^{\infty} \sum_{\sigma} \frac{1}{m!} \left\langle \boldsymbol{\tau}_{\mu}(\overline{\psi}), \dots, \boldsymbol{\tau}_{\mu}(\overline{\psi}), \frac{[(\mu, \sigma)]}{-z - \overline{\psi}_{n+1}} \right\rangle_{0, m+1, 0}^{\operatorname{VEval}(v), T, T \operatorname{Sym}^{d} \mathbb{P}^{r}} \mathbf{1}_{(\mu, \sigma)} \in H_{T, \operatorname{loc}}^{*}(I\mu, \mathbb{Q}),$$

where $1_{(\mu,\sigma)}$ is the fundamental class of $(\mu, \sigma) \in I\mu$, and $\tau_{\mu}(z) = \sum_{\sigma' \in \text{MultiPart}(\mu)} \tau_{(\mu,\sigma')}(z) 1_{(\mu,\sigma)}$. Adding in the contributions from type (ii) graphs, summing (5.2) over σ yields:

$$\mathbf{f}_{\mu} = \sum_{\sigma} \mathbf{f}_{(\mu,\sigma)} \mathbf{1}_{\mu,\sigma}$$
$$= -\mathbf{1}_{\mu} z + \boldsymbol{\tau}_{\mu}(z) + \sum_{m=2}^{\infty} \sum_{\sigma} \frac{1}{m!} \left\langle \boldsymbol{\tau}_{\mu}(\overline{\psi}), \dots, \boldsymbol{\tau}_{\mu}(\overline{\psi}), \frac{[(\mu,\sigma)]}{-z - \overline{\psi}_{n+1}} \right\rangle_{0,m+1,0}^{\operatorname{VEval}(v), T, T \operatorname{Sym}^{d} \mathbb{P}^{r}} \mathbf{1}_{(\mu,\sigma)},$$

where 1_{μ} is the untwisted fundamental class on $I\mu$. This shows that \mathbf{f}_{μ} is a $\Lambda_{\text{nov}}^{T}[[x]]$ -valued point of $\mathcal{L}_{\mu}^{T \operatorname{Sym}^{d} \mathbb{P}^{r}}$. (Here we are using both comments in Remark 2.6.6.)

The converse also requires no modification from [18]. Suppose f satisfies the conditions of theorem. By conditions (I) and (II), we may uniquely write

$$\mathbf{f}_{\mu} = -\mathbf{1}_{\mu}z + \sum_{\sigma \in \operatorname{MultiPart}(\mu)} \boldsymbol{\tau}_{(\mu,\sigma)}(z)\mathbf{1}_{(\mu,\sigma)} + O(z^{-1}),$$

where $\tau_{(\mu,\sigma)}(z)$ is the expression in (5.11), for some $\mathbf{t}_{(\mu,\sigma)}(z) \in \iota_{\mu}^{*}(\mathcal{H}_{+})[[x]]$. We claim that the set $\{\mathbf{t}_{(\mu,\sigma)}(z)\}$ for all fixed points (μ,σ) determines **f**. By the localization isomorphism, if suffices to show that it determines $\mathbf{f}_{(\mu,\sigma)}$ for all (μ,σ) . We induct on the degree $\beta + k$, where k is the exponent of x. The base case $\beta = k = 0$ is taken care of by the assumption $\mathbf{f}|_{Q=x=0} = -1z$. Assume the coefficients of $\mathbf{f}_{(\mu,\sigma)}$ up to degree $\beta + k$ are determined by $\{\mathbf{t}_{(\mu,\sigma)}\}$. Consider the coefficients of degree $\beta + k+1$. Some of these appear in $\mathbf{t}(z)$, but these are given. Some of them appear in $\tau_{(\mu,\sigma)}(z)$,

but these are determined since they are of the form: $Q^{\beta(\tilde{\kappa})}$ multiplied by a factor determined by the inductive hypothesis. The sum of all of these terms is in $H^*_{CR,T,\text{loc}}(\mu, \mathbb{Q})[[x, Q]][[z]]$.

Finally, some of them appear in $O(z^{-1})$. However, condition (III) and (2.6) show that these are determined by terms of $-1z + \tau_{(\mu,\sigma)}(z)$ of degree at most $\beta + k + 1$. Since all such terms are determined by $\mathbf{t}_{(\mu,\sigma)}$ and induction, the degree $\beta + k + 1$ coefficients of $\mathbf{f}_{(\mu,\sigma)}$ are determined. Thus in fact **f** is determined by $\{\mathbf{t}_{(\mu,\sigma)}(z)\}$.

Again by the localization isomorphism, the set $\{\mathbf{t}_{(\mu,\sigma)}(z)\}$ corresponds uniquely to an element $\mathbf{t}(z) \in \mathcal{H}_+[[x]]$ that restricts to each $\mathbf{t}_{(\mu,\sigma)}(z)$. This in turn corresponds uniquely to a $\Lambda_{\text{nov}}^T[[x]]$ -valued point \mathbf{f}_{GW} of \mathcal{L}_x . By the uniqueness argument above we have $\mathbf{f} = \mathbf{f}_{\text{GW}}$.

Remark 5.1.8. No modifications are required to replace x with a tuple (x_1, \ldots, x_m) .

5.2 Characterization of the Givental cone \mathcal{L}_X , general (nonorbifold) case

In this section, we discuss the extent to which Theorem 5.1.4 can be extended to a general variety X with action of a torus T. Here are a few of the major points:

- In Condition (II) of Theorem 5.1.4, the terms of the recursion are elements of H*((μ, σ), Q) for various (μ, σ), and the equality is via the canonical isomorphism between these groups. For more general (X, T), these terms a priori lie in the homology groups of various components of X^T. Thus (5.1) must have a somewhat different form when X does not have isolated fixed points.
- 2. As in the proof of Theorem 5.1.4, for each component μ of X^T , one may compute \mathbf{f}_{μ} by localization, and decompose \mathbf{f}_{μ} as a sum of two terms, the contributions from trees of initial and recursion type. As before, these correspond to poles at z = 0 or ∞ , or poles at $z = w(\tilde{\kappa})$, respectively.
- 3. Essentially the same argument as in the last section shows that Condition (III) of Theorem 5.1.4 holds in general, with $\mathcal{L}_{\mu}^{T \operatorname{Sym}^d \mathbb{P}^r}$ replaced with the Givental cone $\mathcal{L}_{\mu}^{N_{\mu}|X}$ of μ twisted by the normal bundle to μ in X. (This coincides with TX when μ is a point.)
- 4. In [24] Theorem 3.5, Fan and Lee show the existence of a general recursion for points on the Givental cone, via a similar method. Our recursion (5.1) is, in appearance, much simpler than theirs. In particular, the left side of ours is a single Laurent coefficient of f_μ, and the right side is expressed in terms of Laurent coefficients of f_{μ'} for various μ'. On the other hand, the left side of their recursion is the entire principal part Prin_{z=w} f_μ, and the right side is expressed in

terms of *nonprincipal* parts of certain derivatives of $\mathbf{f}_{\mu'}$ for various μ' . Theorem 5.1.4 makes us hope that a simpler form such as this is possible for arbitrary X, and we suspect that it is. For $X = \text{Sym}^d \mathbb{P}^r$, the refined form was enabled by the "miracle" computation (5.7).

Though we are unable to prove a sufficiently refined recursion, we go through some of the steps to demonstrate what is needed and what is difficult.

Definition 5.2.1. (CHANGE THIS) Analogously to 5.1.1, for $\mu \in V(\Gamma_{X,T})$ a connected component of X^T , let $\Upsilon(\mu) \subseteq \operatorname{Trees}_{0,2}(X,\beta)$ be the set of X-decorated trees

$$\tilde{\kappa} = v_1 \bullet \underbrace{e_1}_{v_2} \cdots \underbrace{e_k}_{v_k} \bullet v_{k+1}$$

with $Mark(1) = v_1$, $Mark(2) = v_2$, $\mu = f(v_1)$, $vdeg(v_1) = \cdots = vdeg(v_{k+1}) = 0$, and (v_i, e_{i-1}) unsteady for $2 \le i \le k$.

Notation 5.2.2. For $\tilde{\kappa} \in \Upsilon(\mu)$ as in Definition 5.1.1, we write $\mu'(\tilde{\kappa}) := \mathfrak{f}(v_{k+1})$ and $w(\tilde{\kappa}) := w(v_1, e_1) = w(v_i, e_i)$.

Let **f** be a $\Lambda_{nov}^T[[x]]$ -valued point of \mathcal{L}_X . Using the projection formula, we can write

$$\mathbf{f}_{\mu} = -1_{\mu}z + \mathbf{t}_{\mu}(z) + \sum_{\substack{n,\beta \ge 0\\\beta > 0 \text{ or } n \ge 2}} \frac{Q^{\beta}}{n!} \sum_{\varphi} \left\langle \mathbf{t}(\psi), \dots, \mathbf{t}(\psi), \frac{(\iota_{\mu})_{*}(\gamma_{\varphi}^{\mu})}{-z - \psi} \right\rangle_{0, n+1, \beta}^{X, T} \gamma_{\mu}^{\varphi}.$$
 (5.12)

where $\mathbf{t}_{\mu}(z) := \iota_{\mu}^* \mathbf{t}(z)$. Ignoring the subtleties of Section 4.2, we define a partial order on decorated trees using (not necessarily realizable) chain contractions, as in Section 4.5.1. We consider minimal trees $\tilde{\Gamma}$, and write $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ for the locus of *T*-fixed stable maps $(f : C \to X)$ satisfying $\Psi(f) \geq \tilde{\Gamma}$. Then we may write

$$\sum_{\beta,n} \sum_{\varphi} \left\langle \mathbf{t}(\psi), \dots, \mathbf{t}(\psi), \frac{(\iota_{\mu})_{*}(\gamma_{\varphi}^{\mu})}{-z - \psi} \right\rangle_{0,n+1,\beta}^{X,T} \gamma_{\mu}^{\varphi} = \sum_{\tilde{\Gamma} \in \operatorname{Trees}_{0,n+1}^{\min}(X,\beta)} \operatorname{Contr}_{\mu}(\tilde{\Gamma}),$$
(5.13)

and we may decompose the right side as two sums, one over trees of initial type, and the other over trees of recursion type.

Let $\tilde{\Gamma}$ be a tree of recursion type. By (4.1), we have

$$\operatorname{Contr}(\tilde{\Gamma}) = \sum_{\varphi} \gamma_{\mu}^{\varphi} \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}}]^{vir}} \frac{1}{e_T(N_{\tilde{\Gamma}}^{vir})} \iota_{\tilde{\Gamma}}^* \left(\prod_{j=1}^n \operatorname{ev}_j^* \mathbf{t}(\psi) \cup \frac{\operatorname{ev}_{n+1}^*(\iota_{\mu})_*(\gamma_{\varphi}^{\mu})}{-z - \psi_{n+1}} \right).$$
(5.14)

Since $ev_{n+1} \circ \iota_{\tilde{\Gamma}}$ factors through μ , we have

$$\iota_{\widetilde{\Gamma}}^* \operatorname{ev}_{n+1}^*(\iota_{\mu})_*(\gamma_{\varphi}^{\mu}) = \iota_{\widetilde{\Gamma}}^* \operatorname{ev}_{n+1}^*(\gamma_{\varphi}^{\mu} \cup e_T(N_{\mu|X})).$$

From $\tilde{\Gamma}$, via "edge pruning" we again obtain two decorated trees $\tilde{\kappa} \in \Upsilon(\sigma)$ and $\tilde{\Gamma} \smallsetminus \tilde{\kappa} \in \operatorname{Trees}_{0,n+1}(X, \beta - \beta(\tilde{\kappa}))$. This realizes $\overline{\mathcal{M}}_{\tilde{\Gamma}}$ as the fiber product of moduli spaces $\overline{\mathcal{M}}_e$ and $\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}$. Note that $\overline{\mathcal{M}}_e$ may be quite complicated — for example, the generic source curve may be nodal. (These are the moduli spaces $M_{\omega,2}(X)$ in [49], which works in the case $T = \mathbb{C}^*$, and in the additional generality of weighted stable maps.)

We factor the *T*-equivariant map $\overline{\mathcal{M}}_{\tilde{\Gamma}} \to \operatorname{Spec} \mathbb{C}$ through the second projection, i.e. we integrate over $\overline{\mathcal{M}}_e$:

$$\operatorname{Contr}(\tilde{\Gamma}) = \sum_{\varphi} \gamma_{\mu}^{\varphi} \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}]^{vir}} \int_{[\overline{\mathcal{M}}_{e}]^{vir}} \frac{\iota_{\tilde{\Gamma}}^{*} \left(\operatorname{ev}_{n+1}^{*} (\gamma_{\varphi}^{\mu} \cup e_{T}(N_{\mu|X}) \cup \prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\psi) \cup \frac{1}{-z - \psi_{n+1}} \right)}{e_{T}(N_{\tilde{\Gamma}}^{vir})}.$$
(5.15)

By the proof of Lemma 7.5.2 of [13] (pp. 441–442), we may generalize (5.5) to:

$$\frac{1}{e_T(N_{\widetilde{\Gamma}}^{\operatorname{vir}})} = \frac{e_T(\operatorname{ev}^*(N_{\mu'(\widetilde{\kappa})|X}))}{e_T(N_{\overline{\mathcal{M}}_e}^{\operatorname{vir}})e_T(N_{\widetilde{\Gamma}\smallsetminus\widetilde{\kappa}}^{\operatorname{vir}})(-\psi_e^{\overline{\mathcal{M}}_{v_2}} - \psi_{v_2}^{\overline{\mathcal{M}}_e})}.$$

Then (5.15) becomes

$$\operatorname{Contr}(\tilde{\Gamma}) = \sum_{\varphi} \gamma_{\mu}^{\varphi} \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}]^{vir}} \frac{\iota_{\tilde{\Gamma}}^{*} \left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\psi)\right)}{e_{T}(N_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}^{vir})} \\ \left(\int_{[\overline{\mathcal{M}}_{e}]^{vir}} \frac{e_{T}(\operatorname{ev}_{\bullet}^{*}(N_{\mu'(\tilde{\kappa})|X})) \cup \operatorname{ev}_{n+1}^{*}(\gamma_{\varphi}^{\mu} \cup e_{T}(N_{\mu|X}))}{e_{T}(N_{\overline{\mathcal{M}}_{e}}^{vir})(-\psi_{e}^{\overline{\mathcal{M}}_{v_{2}}} - \psi_{v_{2}}^{\overline{\mathcal{M}}_{e}})(-z - \psi_{n+1})}\right).$$
(5.16)

As in (5.6), we write $\psi_{v_1}^{\overline{\mathcal{M}}_e} = \psi_{v_1}^{\mathrm{ne}} - \overline{w}(\tilde{\kappa})$ and $\psi_{v_2}^{\overline{\mathcal{M}}_e} = \psi_{v_2}^{\mathrm{ne}} + \overline{w}(\tilde{\kappa})$. Expanding the denominator in

(5.16),

$$\operatorname{Contr}(\tilde{\Gamma}) = \sum_{\varphi} \gamma_{\mu}^{\varphi} \int_{[\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}]^{vir}} \left(\frac{e_{T}(\operatorname{ev}_{\bullet}^{*}(N_{\mu'(\tilde{\kappa})})) \cup \iota_{\tilde{\Gamma}}^{*}\left(\prod_{j=1}^{n} \operatorname{ev}_{j}^{*} \mathbf{t}(\psi)\right)}{e_{T}(N_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}^{vir})} \cdot \left(\int_{[\overline{\mathcal{M}}_{e}]^{vir}} \frac{\operatorname{ev}_{n+1}^{*}(\gamma_{\varphi}^{\mu} \cup e_{T}(N_{\mu|X}))}{e_{T}(N_{\overline{\mathcal{M}}_{e}}^{vir})} \sum_{m_{1},m_{2} \ge 0} \frac{(\psi_{v_{2}}^{\mathrm{ne}})^{m_{1}}(\psi_{v_{1}}^{\mathrm{ne}})^{m_{2}}}{(-\psi_{n+1}^{\tilde{\Gamma} \smallsetminus \tilde{\kappa}} - w(\tilde{\kappa}))^{m_{1}+1}(-z + w(\tilde{\kappa}))^{m_{2}+1}} \right) \right)$$

$$=\sum_{\varphi}\gamma_{\mu}^{\varphi}\sum_{m_{1},m_{2}\geq0}\int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}]^{vir}}\left(\frac{e_{T}(\mathrm{ev}_{\bullet}^{*}(N_{\mu'(\tilde{\kappa})}))\cup\iota_{\tilde{\Gamma}}^{*}\left(\prod_{j=1}^{n}\mathrm{ev}_{j}^{*}\mathbf{t}(\psi)\right)}{e_{T}(N_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}^{\mathrm{vir}})(-\psi_{n+1}^{\Gamma\check{\varsigma}\tilde{\kappa}}-w(\tilde{\kappa}))^{m_{1}+1}(-z+w(\tilde{\kappa}))^{m_{2}+1}}\right)$$
$$\cdot\left(\int_{[\overline{\mathcal{M}}_{e}]^{\mathrm{vir}}}\frac{\mathrm{ev}_{n+1}^{*}(\gamma_{\varphi}^{\mu}\cup e_{T}(N_{\mu|X}))(\psi_{v_{2}}^{\mathrm{ne}})^{m_{1}}(\psi_{v_{1}}^{\mathrm{ne}})^{m_{2}}}{e_{T}(N_{\overline{\mathcal{M}}_{e}}^{\mathrm{vir}})}\right)$$
(5.17)

Note that the sums over m_1 and m_2 are finite, since $\psi_{v_1}^{\text{ne}}$ and $\psi_{v_2}^{\text{ne}}$ are nonequivariant, hence nilpotent. Also, the second line of (5.17) is now valued in $H^*_T(\operatorname{Spec} \mathbb{C}, \mathbb{Q})$ instead of $H^*_T(\overline{\mathcal{M}}_{\tilde{\Gamma} \smallsetminus \tilde{\kappa}}, \mathbb{Q})$. Applying the projection formula, (5.17) is equal to

$$\sum_{m_1,m_2\geq 0} \frac{1}{(-z+w(\tilde{\kappa}))^{m_2+1}} \left(\int_{[\overline{\mathcal{M}}_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}]^{vir}} \frac{e_T(\operatorname{ev}^*_{\bullet}(N_{\mu'(\tilde{\kappa})})) \cup \iota_{\tilde{\Gamma}}^*\left(\prod_{j=1}^n \operatorname{ev}_j^* \mathbf{t}(\psi)\right)}{e_T(N_{\tilde{\Gamma}\smallsetminus\tilde{\kappa}}^{vir})(-\psi_{n+1}^{\Gamma\tilde{\varsigma}\tilde{\kappa}} - w(\tilde{\kappa}))^{m_1+1}} \right) \\ \cdot (\operatorname{ev}_{n+1})_* \left((\operatorname{ev}^*_{n+1} e_T(N_{\mu|X})) \cup \frac{(\psi_{v_2}^{\operatorname{ne}})^{m_1}(\psi_{v_1}^{\operatorname{ne}})^{m_2}}{e_T(N_{\overline{\mathcal{M}}_e}^{vir})} \cap [\overline{\mathcal{M}}_e]^{\operatorname{vir}} \right)$$
(5.18)

The integral in this expression is, formally, the contribution of $\tilde{\Gamma} \smallsetminus \tilde{\kappa}$ to

$$\frac{1}{m!} \int_{\mu'(\tilde{\kappa})} \left. \frac{d^{m+1} \mathbf{f}_{\mu'(\tilde{\kappa})}}{dz^{m+1}} \right|_{z \mapsto w(\tilde{\kappa})}.$$

However, this expression is not well-defined, since $\frac{d^{m+1}\mathbf{f}_{\mu'(\tilde{\kappa})}}{dz^{m+1}}$ has, in general, a pole at $z = w(\tilde{\kappa})$. Summing over all trees of recursion type gives instead

$$\sum_{\tilde{\kappa}} \sum_{m_1,m_2 \ge 0} \frac{1}{(-z+w(\tilde{\kappa}))^{m_2+1}} \frac{1}{m!} \int_{\mu'(\tilde{\kappa})} \left(\frac{d^{m+1} \mathbf{f}_{\mu'(\tilde{\kappa})}}{dz^{m+1}} - \operatorname{Prin}_{z=w(\tilde{\kappa})} \frac{d^{m+1} \mathbf{f}_{\mu'(\tilde{\kappa})}}{dz^{m+1}} \right) \Big|_{z\mapsto w(\tilde{\kappa})} \cdot (\operatorname{ev}_{n+1})_* \left((\operatorname{ev}_{n+1}^* e_T(N_{\mu|X})) \cup \frac{(\psi_{v_2}^{\operatorname{ne}})^{m_1}(\psi_{v_1}^{\operatorname{ne}})^{m_2}}{e_T(N_{\overline{\mathcal{M}}_e})} \cap [\overline{\mathcal{M}}_e]^{\operatorname{vir}} \right),$$
(5.19)

where Prin denotes the principal part of the Laurent expansion.

Remark 5.2.3. As mentioned, this recursion (which is the one appearing in [24]) does not contain individual Laurent coefficients of the other restrictions $\mathbf{f}_{\mu'(\tilde{\kappa})}$. We have hope that summing over the index m_1 should give a more elegant expression similar to that in Theorem 5.1.4. If true, this would give us information about the geometry of the mysterious spaces $\overline{\mathcal{M}}_e$.

CHAPTER 6

The *I*-function and mirror theorem for $Sym^d \mathbb{P}^r$

6.1 Statement of the main theorem

In this section we introduce the function $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, -z)$. We will then show that it is a $\Lambda^T_{\text{nov}}[[t, \mathbf{x}]]$ -valued point of $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$, where $\mathbf{x} = \{x_{\varpi}\}_{\varpi \in \text{Part}(d)}$ are formal variables.

Definition 6.1.1. The (extended) *I*-function is

$$I_{\text{Sym}^{d} \mathbb{P}^{r}}(t, \mathbf{x}, Q, z) = z \sum_{\sigma \in \text{Part}(d)} 1_{\sigma} \sum_{\beta \ge 0} \exp\left(\sum_{i=0}^{r} t_{i}([H_{\sigma,i}]/z + \beta)\right) Q^{\beta} \sum_{\substack{\mathbb{Z}_{>0}\text{-labels } L = (L_{\eta})\\\text{of } \sigma \text{ with sum } \beta}} \left(\sum_{\substack{\mathbf{k} = (k_{\varpi})_{\varpi} \in \text{Part}(d)\\k_{\varpi} \ge 0}} \frac{\mathbf{x}^{\mathbf{k}} H(\sigma \prod_{\varpi} \varpi^{k_{\varpi}})}{\mathbf{k}! z^{\mathbf{k}}}\right) \left(\frac{|S_{\sigma}|}{|S_{\sigma,L}|}\right) \left(\prod_{\eta \in \sigma} \frac{1}{\prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(H_{\sigma,\eta,i} + \frac{\gamma}{\eta}z\right)}\right)$$
(6.1)

where:

- $1_{\sigma} \in H^*_{CR,T}(\operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q})$ is the fundamental class of the twisted sector corresponding to σ ,
- $[H_{\sigma,i}]$ and $[H_{\sigma,\eta,i}]$ are defined in Section 3.1.3,
- $\mathbf{x}^{\mathbf{k}} := \prod_{\varpi} x_{\varpi}^{k_{\varpi}},$
- $\mathbf{k}! := \prod_{\varpi} k_{\varpi}!,$
- $z^{\mathbf{k}} := \prod_{\varpi} z^{k_{\varpi}}$, and
- H(σ Π_ω ω^{k_w}) is the number of ways of factoring 1 ∈ S_d as a product a₁ ··· a_{1+Σk_w}, where the conjugacy classes (i.e. partitions) of the permutations a_j are given by the list (σ Π_ω ω^{k_w}).

Note that (6.1) uses the normal cup product on $H^*(I \operatorname{Sym}^d \mathbb{P}^r, \mathbb{Q})$, not the Chen-Ruan product. As mentioned, we prove:

Theorem 6.1.2. $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, -z)$ is a $\Lambda_{\text{nov}}^T[[t, \mathbf{x}]]$ -valued point of $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$.

6.2 **Proof of Theorem 6.1.2**

Instead, we will show that a different series $\underline{I}_{\text{Sym}^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ is a $\Lambda_{\text{nov}}^T[[\mathbf{x}]]$ -valued point of $\mathcal{L}_{\text{Sym}^d \mathbb{P}^r}$, where $\underline{I}_{\text{Sym}^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ is obtained from $I_{\text{Sym}^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ by removing the exponential factor. The divisor equation in Gromov-Witten theory then implies Theorem 6.1.2.

It is immediate that $\underline{I}_{\text{Sym}^d \mathbb{P}^r}(0, 0, 0, -z) = -1z$. Per Theorem 5.1.4, it now suffices to prove conditions (I), (II), and (III), which we do in Sections 6.2.1, 6.2.2, and 6.2.3. We write $\underline{I}_{(\mu,\sigma)}$ for the restriction of $\underline{I}_{\text{Sym}^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ to a *T*-fixed point $(\mu, \sigma) \in \underline{I} \text{Sym}^d \mathbb{P}^r$. We write $r_{\sigma} := \text{lcm}(\sigma)$.

6.2.1 $\underline{I}_{Svm^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ satisfies condition (I)

From (6.1),

$$\underline{I}_{(\mu,\sigma)} = -z \sum_{\beta \ge 0} Q^{\beta} \sum_{\substack{\text{labels } L = (L_{\eta}) \\ \text{of } \sigma \text{ with sum } \beta}} \left(\left(\sum_{\substack{\mathbf{k} = (k_{\varpi})_{\varpi \in \text{Part}(d)} \\ k_{\varpi} \ge 0}} \frac{\mathbf{x}^{\mathbf{k}} H(\sigma \prod_{\varpi} \varpi^{k_{\varpi}})}{\mathbf{k}! (-z)^{\mathbf{k}}} \right) \left(\frac{|S_{\sigma}|}{|S_{\sigma,L}|} \right) \left(\frac{1}{\prod_{\eta \in \sigma} \prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(r_{\sigma}(\alpha_{i(\eta)} - \alpha_{i}) - \frac{\gamma}{\eta} z \right)} \right).$$
(6.2)

It is clear that the coefficient of a single power x and Q is a rational function in z. The poles of such a coefficient are (at worst) z = 0, $z = \infty$, and $z = \frac{r_{\sigma}(\alpha_{i_1} - \alpha_{i_2})}{q}$, where $i_1 = i(\eta)$ for some $\eta \in \sigma$, and $q \in \frac{1}{n}\mathbb{Z}$. This is exactly the set of values arising as $\overline{w}(\tilde{\kappa})$ for $\tilde{\kappa} \in \Upsilon(\mu, \sigma)$. This proves (I).

6.2.2 $\underline{I}_{Svm^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ satisfies condition (II)

We work with the left and right sides of (5.1). Fix a and $\overline{w} = \frac{r_{\sigma}(\alpha_{i_1} - \alpha_{i_2})}{q}$. When applied to $\mathbf{f} = \underline{I}_{\text{Sym}^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ the factor

$$\Omega := -z \left(\sum_{\substack{\mathbf{k} = (k_{\varpi})_{\varpi} \in \operatorname{Part}(d) \\ k_{\varpi} \ge 0}} \frac{\mathbf{x}^{\mathbf{k}} H_{\sigma}(\sigma \prod_{\varpi} \varpi^{k_{\varpi}})}{\mathbf{k}! (-z)^{\mathbf{k}}} \right) \left(\frac{1}{\prod_{\substack{\eta \in \sigma \\ i(\eta) \neq i_{1} \text{ or } \\ L_{\eta} < q\eta}} \prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(r_{\sigma}(\alpha_{i(\eta)} - \alpha_{i}) - \frac{\gamma}{\eta} z \right)} \right)$$

appears identically on both sides, so we may prove (5.1) instead for $\underline{I}_{(\mu,\sigma)}/\Omega$. We break up $\underline{I}_{(\mu,\sigma)}/\Omega$, the left-hand side of (5.1), into terms by the label L (with sum β):

$$T_{(\mu,\sigma),L}(Q,z) := \frac{|S_{\sigma}|}{|S_{\sigma,L}|} \left(\frac{Q^{\beta}}{\prod_{\substack{\eta \in \sigma \\ i(\eta)=i_1 \\ L_{\eta} \ge q\eta}} \prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(r_{\sigma}(\alpha_{i(\eta)} - \alpha_{i}) - \frac{\gamma}{\eta} z \right)} \right).$$
(6.3)

Write σ_T for the multiset $\{\eta \in \sigma | i(\eta) = i_1, L_\eta \ge q\eta\}$. This consists of parts of σ that are in $Mov(\tilde{\kappa})$ for some $\tilde{\kappa}$ with weight \overline{w} based at (μ, σ) . We compute

$$\operatorname{Laur}(T_{(\mu,\sigma),L}(Q,z),(\overline{w}-z)^{-a}) = \frac{1}{(|\sigma_{T}|-a)!} \left(\frac{d^{|\sigma_{T}|-a}}{d(\overline{w}-z)^{|\sigma_{T}|-a}} (\overline{w}-z)^{|\sigma_{T}|} T_{(\mu,\sigma),L}(Q,z) \right)_{z\mapsto\overline{w}}$$
$$= \frac{|S_{\sigma}| / |S_{\sigma,L}|}{(|\sigma_{T}|-a)!} \left(\sum_{A} \prod_{(\eta,\gamma,i)\in A} \frac{\gamma/\eta}{\left(r_{\sigma}(\alpha_{i_{1}}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right)}\right) \right)$$
$$\left(\frac{Q^{\beta}}{q^{|\sigma_{T}|} \prod_{\eta\in\sigma_{T}} \prod_{i=0}^{r} \prod_{\substack{1\leq\gamma\leq L_{\eta}\\(\gamma,i)\neq(q\eta,i_{2})}} \left(r_{\sigma}(\alpha_{i_{1}}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right)}\right), \quad (6.4)$$

where A ranges over $(|\sigma_T| - a)$ -tuples of factors in the denominator of $T_{(\mu,\sigma),L}(Q, z)$, i.e. over unordered tuples of triples (η, γ, i) , with $\eta \in \sigma_T$, $1 \le \gamma \le L_{\eta}$, $0 \le i \le r$, and $(\gamma, i) \ne (q\eta, i_2)$. Observe that

$$r_{\sigma}(\alpha_{i_1} - \alpha_i) - \frac{\gamma}{\eta}\overline{w} = r_{\sigma}\left(\frac{q\eta - \gamma}{q\eta}\alpha_{i_1} + \frac{\gamma}{q\eta}\alpha_{i_2} - \alpha_i\right).$$

For the right-hand side of (5.1), let $\tilde{\kappa} \in \Upsilon(\mu, \sigma)$ with $\overline{w}(\tilde{\kappa}) = \overline{w}$. Such an edge corresponds to a subset $Mov(\tilde{\kappa}) \subseteq \sigma_T$. Let $L'(\tilde{\kappa})$ be the label of $\sigma'(\tilde{\kappa})$ obtained by decreasing L_η by $q\eta$ for $\eta \in$ $Mov(\tilde{\kappa})$, using an identification of σ and $\sigma'(\tilde{\kappa})$ as partitions. (There is a factor of $|S_{\sigma,L'(\tilde{\kappa})}| / |S_{\sigma,L}|$ from different choices of identification.) As above we write $T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa})),L'(\tilde{\kappa})}(Q, z)$ for the factors of $\underline{I}_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa}))}$ coming from σ_T . (The meaning of the multiset $\sigma_T \subseteq \sigma$ has not changed.) Then as before

$$\operatorname{Laur}(T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa})),L'(\tilde{\kappa})}(Q,z),(\overline{w}-z)^{\operatorname{mov}(\tilde{\kappa})-a}) = \frac{1}{(|\sigma_T|-a)!} \left(\frac{d^{|\sigma_T|-a}}{d(\overline{w}-z)^{|\sigma_T|-a}}(\overline{w}-z)^{|\sigma_T|-\operatorname{mov}(\tilde{\kappa})}T_{(\mu'(\tilde{\kappa}),\sigma'(\tilde{\kappa})),L'(\tilde{\kappa})}\right)$$

$$= \frac{|S_{\sigma}| / |S_{\sigma,L}| Q^{\beta-\beta(\tilde{\kappa})}}{(|\sigma_{T}|-a)! q^{|\sigma_{T}|-\operatorname{mov}(\tilde{\kappa})}} \left(\sum_{B_{\tilde{\kappa}}} \prod_{(\eta,\gamma,i)\in B_{\tilde{\kappa}}} \frac{\gamma/\eta}{\left(r_{\sigma}(\alpha_{i(\eta)}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right)} \right)$$
(6.5)
$$\cdot \left(\frac{1}{\prod_{\eta\in\operatorname{Mov}(\tilde{\kappa})} \prod_{i=0}^{r} \prod_{\gamma=1}^{L_{\eta}-q\eta} \left(r_{\sigma}(\alpha_{i_{2}}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right)}\right)}{\left(\frac{1}{\prod_{\eta\in\sigma_{T}\smallsetminus\operatorname{Mov}(\tilde{\kappa})} \prod_{i=0}^{r} \prod_{\gamma=1}^{L_{\eta}-q\eta} \prod_{\substack{1\leq\gamma\leq L_{\eta}\\(\gamma,i)\neq(q\eta,i_{2})}} \left(r_{\sigma}(\alpha_{i_{1}}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right)}\right),$$

where $B_{\tilde{\kappa}}$ runs over $(|\sigma_T| - a)$ -tuples of factors in the denominator. The product in the denominator over $\eta \in \sigma_T \setminus Mov(\tilde{\kappa})$ appears identically in (6.4), and the product over $\eta \in Mov(\tilde{\kappa})$ of the factors

$$\left(r_{\sigma}(\alpha_{i_{2}}-\alpha_{i})-\frac{\gamma}{\eta}\overline{w}\right) = \left(r_{\sigma}(\alpha_{i_{1}}-\alpha_{i})-\left(q+\frac{\gamma}{\eta}\right)\overline{w}\right)$$

appears in (6.4) via the substitution $\gamma \mapsto \gamma - q\eta$. Together with the denominator of $\mathbf{RC}(\tilde{\kappa}, a)$, this makes up entire denominator of (6.4), excluding the sum over A. The factor Q^{β} also appears on both sides, so it remains to prove:

$$\sum_{A} \prod_{(\eta,\gamma,i)\in A} \gamma/\eta = \sum_{\tilde{\kappa}} \sum_{B_{\tilde{\kappa}}} (-1)^{\operatorname{mov}(\tilde{\kappa})-a} {\sigma_{i_{1}^{\operatorname{mov}}(\tilde{\kappa})} \choose \operatorname{Mov}(\tilde{\kappa})} {\operatorname{mov}(\tilde{\kappa})-1 \choose a-1} \prod_{(\eta,\gamma,i)\in B_{\tilde{\kappa}}} \gamma/\eta.$$
(6.6)

We switch the order of summation on the right-hand side, and identify each tuple $B_{\tilde{\kappa}}$ with one of the tuples A via the substitution $\gamma/\eta \mapsto \gamma/\eta - q$ for $\eta \in Mov(\tilde{\kappa})$. We now want to prove:

$$\sum_{A} \prod_{(\eta,\gamma,i)\in A} \gamma/\eta = \sum_{A} \sum_{\substack{\operatorname{Mov}\subseteq\sigma_T\\|\operatorname{Mov}|\geq a}} (-1)^{|\operatorname{Mov}|-a} \binom{\sigma_{i_1}}{\operatorname{Mov}} \binom{|\operatorname{Mov}|-1}{a-1} \prod_{\substack{(\eta,\gamma,i)\in A\\\eta\in\operatorname{Mov}}} (\gamma/\eta-q) \prod_{\substack{(\eta,\gamma,i)\in A\\\eta\notin\operatorname{Mov}}} \gamma/\eta.$$

We break up the right side further by fixing the set $A' := \{(\eta, \gamma, i) \in A | \eta \in Mov\}$:

$$\sum_{A} \sum_{A' \subseteq A} \prod_{(\eta,\gamma,i) \in A'} (\gamma/\eta - q) \prod_{(\eta,\gamma,i) \in A \smallsetminus A'} \gamma/\eta \sum_{\substack{\text{Mov} \subseteq \sigma_T \\ |\text{Mov}| \ge a \\ \eta \notin \text{Mov for } (\eta,\gamma,i) \in A' \\ \eta \notin \text{Mov for } (\eta,\gamma,i) \in A \smallsetminus A'}} (-1)^{|\text{Mov}| - a} {\sigma_{i_1} \\ \text{Mov}} {\left(|\text{Mov}| - 1 \\ a - 1 \right)}.$$
(6.7)

The factor $\binom{\sigma_{i_1}}{Mov}$ turns the second summation on the right into a sum over *labeled* submultisets $Mov \subseteq \sigma_T$. We then use the straightforward combinatorial identity:

$$\sum_{\substack{|\mathrm{Mov}| \ge a \\ \eta \in \mathrm{Mov \ for }(\eta, \gamma, i) \in A' \\ \eta \notin \mathrm{Mov \ for }(\eta, \gamma, i) \in A \smallsetminus A'}} (-1)^{|\mathrm{Mov}| - a} \binom{|\mathrm{Mov}| - 1}{a - 1} = \begin{cases} 0 & A' \neq \emptyset \\ 1 & A' = \emptyset. \end{cases}$$

Thus (6.7) is equal to $\sum_{A} \prod_{(\eta,\gamma,i) \in A} \gamma/\eta$, proving (6.6) and (II).

6.2.3 $\underline{I}_{Sym^d \mathbb{P}^r}(\mathbf{x}, Q, -z)$ satisfies condition (III)

Finally, we prove (III) using Tseng's orbifold quantum Riemann-Roch operator. From Proposition 3.4 of [34], Proposition 3.4, we compute that $J_{\mu}(\mathbf{x}, Q, -z)$ is a $\Lambda_{\text{nov}}^{T}[[\mathbf{x}]]$ -valued point of the *untwisted* Givental cone \mathcal{L}_{μ} , where¹

$$J_{\mu}(\mathbf{x}, Q, z) := z \sum_{\sigma} \sum_{\mathbf{k}, \beta} \frac{1_{\sigma} Q^{\beta} \mathbf{x}^{\mathbf{k}} H_{\sigma}(\sigma, \mathbf{k})}{\beta! z^{\beta} \mathbf{k}! z^{\mathbf{k}}}$$

Here σ runs over conjugacy classes in G_{μ} , and $\mathbf{t}(\mathbf{x}, Q, z) = Q \cdot 1 + \sum_{\varpi} x_{\varpi} \mathbf{1}_{\varpi}$. We introduce variables $Q_{\sigma,\eta}$ indexed by a multipartition σ and part η , and define

$$J^{Q}_{\mu}(\mathbf{x},Q,-z) := -z \sum_{\sigma} \sum_{\mathbf{k},\beta,L} \frac{1_{\sigma} \mathbf{x}^{\mathbf{k}} H_{\sigma}(\sigma,\mathbf{k})}{\mathbf{k}!(-z)^{\mathbf{k}}} \left(\frac{|S_{\sigma}|}{|S_{\sigma,L}|} \prod_{\eta \in \sigma} \frac{Q^{L_{\eta}}_{\sigma,\eta} \eta^{L_{\eta}}}{(-z)^{L_{\eta}} L_{\eta}!} \right)$$
$$\underline{I}^{Q}_{\mu} := -z \sum_{\sigma,\beta,\mathbf{k},L} \frac{1_{\sigma} \mathbf{x}^{\mathbf{k}} H_{\sigma}(\sigma \prod_{\varpi} \varpi^{k_{\varpi}})}{\mathbf{k}!(-z)^{\mathbf{k}}} \left(\frac{|S_{\sigma}|}{|S_{\sigma,L}|} \prod_{\eta \in \sigma} \frac{Q^{L_{\eta}}_{\sigma,\eta} \eta^{L_{\eta}}}{(-z)^{L_{\eta}} L_{\eta}! \prod_{\gamma=1}^{L_{\eta}} \prod_{i=0}^{r} \left(\alpha_{i(\eta)} - \alpha_{i} - \frac{\gamma}{\eta} z \right) \right)$$

Using combinatorics we may check that specializing $Q_{\sigma,\eta} = Q$ for all σ, η recovers $J_{\mu}(\mathbf{x}, Q, z)$ and \underline{I}_{μ} . This is the analog of the change of variables after Equation 36 of [18].

¹Note that the coefficients of $J_{\mu}(\mathbf{x}, Q, z)$ differ from the Gromov-Witten invariants of μ (as calculated in [34]) by a factor of |G|; this is because on an orbifold point, the Poincaré dual to the fundamental class 1 is $|G| \cdot 1$.

From here, (III) essentially follows from the proof in Section 7.3 of [18], despite the fact that $T_{\mu} \operatorname{Sym}^{d} \mathbb{P}^{r}$ is not a direct sum of line bundles. We give an outline.

As in [18], we work with a general multiplicative characteristic class c_s . Denote by Δ_s the orbifold quantum Riemann-Roch operator, which by [54] maps \mathcal{L}_{μ} to $\mathcal{L}_{\mu}^{T \operatorname{Sym}^d \mathbb{P}^r}$. Explicitly,

$$\Delta_{\mathbf{s}} := \bigoplus_{\sigma} \exp\left(\sum_{\substack{\eta \in \sigma \\ 0 \le i \le r \\ i \ne i(\eta)}} \sum_{\substack{0 \le \ell \le \eta - 1 \\ m \ge 0}} s_{m-1} R(\sigma, \ell) \frac{B_m(\ell/\eta)}{m!} z^{m-1}\right),$$

where B_m is the *m*th Bernoulli polynomial, and $R(\sigma, \ell)$ is the rank of the eigenbundle of $T_\mu \operatorname{Sym}^d \mathbb{P}^r$ on which elements of $\sigma \subseteq G_\mu$ act with eigenvalue $e^{2\pi i \ell/\eta}$. Note that the values

$$s_{k} = \begin{cases} -\log(\alpha_{i(\eta)} - \alpha_{i}) & k = 0\\ (-1)^{k} \frac{(k-1)!}{(\alpha_{i(\eta)} - \alpha_{i})^{k}} & k > 0 \end{cases}$$

recover the *T*-equivariant Euler class. Using the functional equation of the Bernoulli polynomials, we may check that $\underline{I}^Q_{\mu,\sigma} = \Delta_{\mathbf{s}}(J^{\text{mod}}_{\mu}(\mathbf{x}, Q, -z))$, where

$$J_{\mu}^{\text{mod}}(\mathbf{x}, Q, -z) := -z \sum_{\sigma} \sum_{\mathbf{k}, \beta, L} \frac{1_{\sigma} \mathbf{x}^{\mathbf{k}} H_{\sigma}(\sigma \prod_{\varpi} \varpi^{k_{\varpi}})}{\mathbf{k}! (-z)^{\mathbf{k}}} \frac{|S_{\sigma}|}{|S_{\sigma,L}|} \prod_{\eta \in \sigma} \left(\frac{Q_{\sigma, \eta}^{L_{\eta}} \eta^{L_{\eta}}}{(-z)^{L_{\eta}} L_{\eta}!} \right)$$

$$\exp\left(\sum_{i \neq i(\eta)} \sum_{\ell=1}^{\eta-1} \sum_{n, m \ge 0} s_{n+m-1} B_{m}(0) R(\sigma, \ell) \frac{\left(z\ell/\eta - z \left\lfloor \frac{L_{\eta}+\ell}{\eta} \right\rfloor \right)^{n}}{n!} \frac{z^{m-1}}{m!} \right).$$
(6.8)

Analyzing the floor function, we have

$$\sum_{\ell=1}^{\eta-1} \frac{\left(z\ell/\eta - z\left\lfloor\frac{L_{\eta}+\ell}{\eta}\right\rfloor\right)^n}{n!} = \sum_{\ell=1}^{\eta-1} \frac{(z(-L_{\eta}+\ell)/\eta)^n}{n!}.$$

Now (6.8) is equal to $P(Q_{\sigma,\eta} \frac{\partial}{\partial Q_{\sigma,\eta}})(J^Q_{\mu})$, for

$$P(a) = \exp\left(\sum_{i \neq i(\eta)} \sum_{\ell=1}^{\eta-1} \sum_{n,m \ge 0} s_{n+m-1} B_m(0) R(\sigma,\ell) \frac{(z(-a+\ell)/\eta)^n}{n!} \frac{z^{m-1}}{m!}\right).$$

The inductive argument at the end of the proof of Theorem 4.6 of [16] shows that (after specializing $Q_{\sigma,\eta} = Q$) the series $J^{\text{mod}}_{\mu}(\mathbf{x}, Q, -z)$ is a $H^*_{T,\text{loc}}(\text{Spec } \mathbb{C}, \mathbb{Q})[[Q_{\sigma,\eta}, \mathbf{x}]]$ -valued point of \mathcal{L}_{μ} , and

orbifold quantum Riemann-Roch then shows that $\underline{I}_{\mu}(\mathbf{x}, Q, -z)$ is a $H^*_{T, \text{loc}}(\text{Spec } \mathbb{C}, \mathbb{Q})[[Q_{\sigma, \eta}, \mathbf{x}]]$ -valued point of $\mathcal{L}^{T \operatorname{Sym}^d \mathbb{P}^r}_{\mu}$.

6.3 $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, z)$ is equal to the *J*-function of $\text{Sym}^d \mathbb{P}^r$

The beginning terms of $I_{Sym^d \mathbb{P}^r}(t, \mathbf{x}, Q, z)$ are

$$I_{\operatorname{Sym}^{d}\mathbb{P}^{r}}(t, \mathbf{x}, Q, z) = 1z + \sum_{\sigma} \sum_{i=0}^{\prime} t_{i}[H_{\sigma,i}] + \sum_{\varpi \in \operatorname{Part}(d)} x_{\varpi} 1_{\varpi} + O(z^{-1}).$$

On the other hand, by definition, there is a unique $\Lambda_{nov}^T[[t, \mathbf{x}]]$ -valued point of $\mathcal{L}_{Sym^d \mathbb{P}^r}$ of this form, namely the *J*-function

$$J_{\operatorname{Sym}^{d}\mathbb{P}^{r}}(Q,\theta,z) = 1z + \theta + \sum_{\beta,n} \frac{Q^{\beta}}{n!} \left\langle \theta, \dots, \theta, \frac{\gamma_{\phi}}{z - \overline{\psi}} \right\rangle_{0,n+1,\beta}^{\operatorname{Sym}^{d}\mathbb{P}^{r},T} \gamma^{\phi},$$
(6.9)

where $\theta = \sum_{\sigma} \sum_{i=0}^{r} t_i [H_{\sigma,i}] + \sum_{\varpi \in \text{Part}(d)} x_{\varpi} 1_{\varpi}$. Thus:

 ${\rm Corollary \ 6.3.1.} \ I_{{\rm Sym}^d\,\mathbb{P}^r}(t,{\bf x},Q,z)=J_{{\rm Sym}^d\,\mathbb{P}^r}(Q,\theta,z).$

6.4 The origins of the *I*-function

6.4.1 Quasimaps

One of the main challenges of proving a mirror theorem such as Theorem 6.1.2 is that the series $I_{\text{Sym}^d \mathbb{P}^r}(t, \mathbf{x}, Q, z)$ must be guessed in some way (or, in the original story, predicted by physicists).

Recently, there has been major progress on this front, due to work of Ciocan-Fontanine and Kim [13] on quasimaps. For $X = W /\!\!/ G$ a GIT quotient of an affine variety, they produce a collection of moduli spaces $\overline{\mathcal{Q}}_{0,n}^{\epsilon,\delta}(X,\beta)$ of (ϵ,δ) -stable quasimaps, such that when $\epsilon,\delta \gg 0$, we have $\overline{\mathcal{Q}}_{0,n}^{\epsilon,\delta}(X,\beta) \cong \overline{\mathcal{M}}_{0,n}(X,\beta)$. (When $\epsilon \gg 0$ they are spaces of weighted stable maps.)

Precisely, these spaces parametrize maps to [W/G], with weighted markings, where points of C may map into the G-unstable locus of W. The parameters $\epsilon > 0$ and $0 < \delta \le 1$ specify a stability condition which governs the number of marked points that may collide, and the maximum order of contact the curve may have with the unstable locus at a point.

The authors then define *quasimap invariants*, a generalization of Gromov-Witten invariants, via a perfect obstruction theory on $\overline{Q}_{0,n}^{\epsilon,\delta}(X,\beta)$. As in Gromov-Witten theory, twisted invariants and equivariant invariants can also be defined.

6.4.2 Graph spaces and *I*-functions

By the stability condition on stable maps, the space $\overline{\mathcal{M}}_{0,n}(X,\beta)$ is empty when $\beta = 0$ and n < 3. Indeed, in the definition of a point of the Givental cone, the sum is over $n, \beta \ge 0$ with $(n+1,\beta) \notin \{(1,0), (2,0)\}$. The first two terms -1z and $\mathbf{t}(z)$ of such a function f may be understood as corrections, corresponding to these two "missing" cases (respectively).

This can be made precise. There is a moduli space, called a graph moduli space and denoted $\overline{\mathcal{M}}G_{0,n}(X,\beta)$, that parametrizes stable maps with a single parametrized component (on which the stability condition is not required). This space has a \mathbb{C}^* -action, and one may exhibit a Gromov-Witten invariant of X appearing in f as a localization contribution to a certain integral on $\overline{\mathcal{M}}G_{0,n}(X,\beta)$. The advantage of this point of view is that $\overline{\mathcal{M}}G_{0,n}(X,\beta)$ exists for any n and β . Applying the same computation when $\beta = 0$ and n < 3 yields a conventional way of defining "extra" Gromov-Witten invariants. These exactly give the terms -1z and $\mathbf{t}(z)$ in f. This technique was used in Givental's proof of a mirror theorem for quintic threefolds [28].

The same technique can be adapted to quasimaps, as follows. The moduli space $\overline{\mathcal{Q}}_{0,n}^{\epsilon,\delta}(X,\beta)$ is instead empty whenever

$$-2 + \delta n + \epsilon \beta \le 0. \tag{6.10}$$

There is a quasimap graph space $\overline{Q}G_{0,n}^{\epsilon,\delta}(X,\beta)$ that is defined for all n and β , even in the limit $\epsilon, \delta \to 0$. The same operation as above of defining conventional extra invariants works here. Note that in the limit $\epsilon, \delta \to 0$, condition (6.10) is *always* satisfied, i.e. *all* quasimap invariants are defined via this convention. The *I*-function \mathbb{I}_X of *X* is a generating function analogous to **f** (with the simple value $\mathbf{t}(z) = \mathbf{t}(\{t_{\phi}\}, Q, z) = \sum_{\phi} t_{\phi} \gamma_{\phi})$, but defined using quasimap invariants in the limit $\epsilon, \delta \to 0$.

In many special cases, \mathbb{I}_X may be computed explicitly. The authors conjecture that the *I*-function defined this way is on the Givental cone \mathcal{L}_X , and prove it in a special case:

Theorem 3.3 of [14]. If X has a T-action with isolated fixed points and 1-dimensional orbits, then \mathbb{I}_X is on the (equivariant) Givental cone \mathcal{L}_X .

6.4.3 *I*-functions of orbifolds

In the case that X does not admit such a torus action, we may still hope that the I-function produced this way would lie on the Lagrangian cone, and that this could be proven by other means. In general this is a very promising method, but there is an obstruction when X is an orbifold.² To explain this

²An orbifold GIT quotient is the stack quotient $[W^{ss}/G]$ of the semistable locus by G. For example, $X = \text{Sym}^d \mathbb{P}^r$ is a GIT quotient via the presentation $\mathbb{C}^{(r+1)d} / (\mathbb{C}^*) \text{ wr } S_d$.

obstruction, we need to be more explicit about how to express a quasimap invariant as a localization contribution on $\overline{\mathcal{Q}}G_{0,n}^{\epsilon,\delta}(X,\beta)$.

The group \mathbb{C}^* acts on $\overline{\mathcal{Q}}G_{0,n}^{\epsilon,\delta}(X,\beta)$ by scaling on the parametrized component of each curve C. If a point $(f: C \to [W/G])$ is to be \mathbb{C}^* -fixed, then this component, identified with \mathbb{P}^1 , must map to a single point of $X = [W^{ss}/G]$, except for 0 and ∞ , which may map into the unstable locus $[W^{uns}/G]$. Similarly marked points and nodes must be at 0 or ∞ . If there are nodes at 0 and ∞ , then chopping C at 0 and ∞ yields two quasimaps to X, of degrees adding up to β .

Consider the case where there is a node at 0, but just the marked point b_{n+1} at ∞ . This locus $F_{n,\beta}$ is an open and closed substack of the \mathbb{C}^* -fixed locus of $\overline{\mathcal{Q}}G_{0,n+1}^{\epsilon,\delta}(X,\beta)$. It is isomorphic to $\overline{\mathcal{Q}}_{0,n+1}^{\epsilon,\delta}(X,\beta)$. The normal bundle $N_{F_{n,\beta}}$ parametrizes smoothings of the node 0, as well as infinitesimal movements of the special points at 0 and ∞ on \mathbb{P}^1 . It has equivariant Euler class

$$e(N_{F_{n,\beta}}) = -z^2(-z - \psi),$$

where ψ is the cotangent class of the (n + 1)st marked point under the identification with $\overline{\mathcal{Q}}_{0,n+1}^{\epsilon,\delta}(X,\beta)$, and z is the equivariant parameter. Thus (using the fact that ev_{n+1} is identified with itself under this isomorphism) the contribution of $F_{n,\beta}$ to an integral $\langle \mathbf{t}(\psi), \ldots, \mathbf{t}(\psi), \phi \rangle_{0,n+1,\beta}^{X,G}$ over the graph space is precisely the *non-graph* quasimap invariant $\langle \mathbf{t}(\psi), \ldots, \mathbf{t}(\psi), \frac{\phi}{-z^2(-z-\psi)} \rangle_{0,n+1,\beta}^X$. Up to the factor $-z^2$, this is the invariant appearing in f.

The previous paragraph only makes sense when condition (6.10) does not hold, since in that case there can be no node at zero. When it does hold, however, the locus $F_{n,\beta}$ still makes sense; it parametrizes maps $\mathbb{P}^1 \to [W/G]$, where *n* marked points are at 0, one marked point b_{n+1} is at ∞ , and where the point 0 maps into the unstable locus. This moduli space is now, at least in many cases, isomorphic to X via ev_{n+1} . Computing the virtual normal bundle $N_{F_{n,\beta}}^{vir}$ is now more difficult, but by doing so one essentially obtains the *I*-function.

We can now see why this strategy fails when X is an orbifold. For one thing, above we had all n marked points of C concentrated at the point $0 \in \mathbb{P}^1$. But there is no known way to make sense of multiple orbifold points colliding in a family. Second, there are additional complications if one would like orbifold points to map into the unstable locus, which generally has an infinite stabilizer.

There is a modification presented in [12] for dealing with these problems. The authors simply restrict to the case n = 0. This gives an *I*-function with significantly less data than hoped (it is a Λ_{nov}^T -valued point of \mathcal{L}_X , rather than a $\Lambda_{nov}^T[[x_1, \ldots, x_n]]$ -valued point), but the authors show that in the case where X is a toric orbifold, by carefully picking a GIT presentation all of the data may be recovered.

Let us see why this modification is not available in the case $X = \operatorname{Sym}^d \mathbb{P}^r$. When n = 0, the source curve C has a single orbifold point, at ∞ . A quasimap $f : C \to \operatorname{Sym}^d \mathbb{P}^r$ consists of

a principal S_d -bundle over C, and an S_d -equivariant quasimap to $(\mathbb{P}^r)^d$. However, there are no nontrivial S_d -bundles over a \mathbb{P}^1 with one orbifold point. Therefore in some sense, the *I*-function obtained via this modification will not see the fact that $\operatorname{Sym}^d \mathbb{P}^r$ is an orbifold at all! Precisely, the *I*-function will take values in the usual cohomology of $\operatorname{Sym}^d \mathbb{P}^r$, rather than the Chen-Ruan cohomology. In general, we only expect the recovery process mentioned above to work in rare cases, possibly only for toric orbifolds.

6.4.4 *I*-functions of orbifolds, attempt 2

We propose a method that in some simpler sense allows orbifold points to collide. We will show how this method gives rise to the *I*-function of Theorem 6.1.2, which gives strong evidence that the method is good.

Let $f : C \to X$ be a map from an orbifold curve into an orbifold X. By pulling back a cover of orbifold charts of X, we may cover C with open sets $\{U_i\}$ such that $f|_{U_i}$ is a map to a quotient $[M_i/G_i] \subseteq X$, i.e. $f|_{U_i}$ is given locally by a principal G_i -bundle $\tilde{U}_i \to U_i$ and a G_i -equivariant map $\tilde{U}_i \to M_i$, where G_i is finite. This still works if $f : C \to X$ is a family of maps over a base scheme S.

Suppose $S = \mathbb{C}^*$, and as $s \to 0$, two orbifold points of C approach each other. (It is easy to make rigorous sense of this, as in the case of usual moduli of weighted stable maps.) From above, we have the data of a principal G_i -bundle on C. Is it possible to find a flat limit of C over \mathbb{C} such that the map and principal G_i -bundle extend? First, we deal with the question of how to extend C.

Example 6.4.1. Consider the family $[\{xy = tz^2\}/S_2] \subseteq [\mathbb{P}^2/S_2] \times \mathbb{C}^*$, where μ_r acts by permuting x and y. This is a family over \mathbb{C}^* of orbifold curves, each of which has two orbifold points of order 2. When we fill it in the natural way, the limit is $[\{xy = 0\}/\mu_2]$, a *singular* Deligne-Mumford curve with coarse moduli space isomorphic to \mathbb{P}^1 .

This construction generalizes as follows. We replace the group S_2 with the dihedral group D_{2n} , where a reflection acts by permuting x and y, and a rotation acts by scaling x and y by an nth root of unity. Then one checks that this is also a family of orbifold curves, each with two orbifold points of order 2. The limit is again singular with coarse moduli space \mathbb{P}^1 , but it is not isomorphic to the previous one (e.g. it has isotropy group D_{2n} at the singular point). This shows that in general there are many possible limiting curves resulting from the collision of two orbifold points. (In fact, they may come in a positive-dimensional family, if the base S has higher dimension.) These particular singularities were studied by Ekedahl [23].

In any example like this, properness of BG_i allows one to extend the principal G_i -bundle to the filled-in family. However, because the behavior of degenerating orbifold curves is extremely

complicated, we do not know how to define a quasimap moduli space where orbifold points may collide. Even if we could do so, we would not know how to compute with it.

Instead we propose the following solution. Again suppose we are given a family of orbifold stable maps over $S = \mathbb{C}^*$. There is a well-defined monodromy, a conjugacy class in G_i , around a marked section. In the limit, there is also a well-defined monodromy around the collection of colliding marked sections, given by taking a small loop that encloses those sections. Rather than taking the limiting curve as $s \to 0$, we will puncture C at the marked sections (which does not lose any data) and then fill in a limiting punctured curve. This curve has a natural principal G_i -bundle with monodromy around the collided marked points equal to the one just described. We may use this monodromy to fill in a curve and a map to X, but the resulting curve does **not** fit in a flat family with C. Therefore our moduli space will parametrize quasimaps from punctured curves, with specified monodromy data at the punctures.

Here is how we use this to compute, when X = [M/G] is a global quotient by a finite group. The part of the fixed locus $F_{n,\beta}$ in $\overline{Q}G_{0,n+1}^{\epsilon,\delta}(X,\beta)$ parametrizes maps where all marked points have collided at 0. A point of $F_{n,\beta}$ now carries not only the information of the target point in X, but also the monodromy data of all n collided marked points. Thus it contains an open and closed substack for each possible assignment of monodromy data; these are indexed by factorizations of the identity in G. In particular, for $X = \text{Sym}^d \mathbb{P}^r$, we have $G = S_d$. This is where the Hurwitz number appearing in (6.1) comes from.

The other interesting part of (6.1) is the product at the end, which comes from the normal bundle to $F_{n,\beta}$. For $(f : C \to \operatorname{Sym}^d \mathbb{P}^r) \in F_{n,\beta}$, the étale cover $C' \to C$ is simply a union of covers, fully ramified over 0 and ∞ . The monodromy is given by the partition $\sigma = \operatorname{Mon}(n+1)$. The degrees of the map $C' \to \mathbb{P}^r$ on each rational connected component give a $\mathbb{Z}_{\geq 0}$ -labeling of σ with sum β . The deformations of the maps $C' \to \mathbb{P}^r$ are a product over these connected components, and the weights are computed using Kontsevich's argument. The result is the product in (6.1) (when done equivariantly).

BIBLIOGRAPHY

- [1] Dan Abramovich, Alessio Corti, and Angelo Vistoli. Twisted bundles and admissible covers. *Communications in Algebra*, 31(8):3547–3618, 2003.
- [2] Dan Abramovich, Tom Graber, and Angelo Vistoli. Gromov–Witten theory of Deligne-Mumford stacks. *American Journal of Mathematics*, 130(5):1337–1398, 2008.
- [3] Dan Abramovich and Angelo Vistoli. Compactifying the space of stable maps. *Journal of the American Mathematical Society*, 15:27–75, 2002.
- [4] Alejandro Adem, Johann Leida, and Yongbin Ruan. *Orbifolds and Stringy Topology*. Cambridge Tracts in Mathematics. Cambridge University Press, 2007.
- [5] Michael Atiyah and Raoul Bott. The moment map and equivariant cohomology. *Topology*, 23(1):1–28, 1984.
- [6] Kai Behrend and Barbara Fantechi. The intrinsic normal cone. *Inventiones mathematicae*, 128(1):45–88, March 1997.
- [7] Aaron Bertram. Another way to enumerate rational curves with torus actions. *Inventiones mathematicae*, 142(3):487–512, 2000.
- [8] Jeff Brown. Gromov–Witten invariants of toric fibrations. *International Mathematics Research Notices*, 2014(19):5437–5482, 2014.
- [9] Jim Bryan and Tom Graber. The crepant resolution conjecture. In *Algebraic Geometry–Seattle* 2005. Part 1, pages 23–42, 2009.
- [10] Philip Candelas, Xenia De La Ossa, Paul Green, and Linda Parkes. A Pair of Calabi-Yau manifolds as an exactly soluble superconformal theory. *Nuclear Physics*, B359:21–74, 1991.
- [11] Huai-Liang Chang, Young-Hoon Kiem, and Jun Li. Torus localization and wall crossing for cosection localized virtual cycles. *ArXiv e-prints*, January 2015. arXiv:1502.00078.
- [12] Daewoong Cheong, Ionuţ Ciocan-Fontanine, and Bumsig Kim. Orbifold quasimap theory. *Mathematische Annalen*, 363(3):777–816, 2015.
- [13] Ionuţ Ciocan-Fontanine and Bumsig Kim. Wall-crossing in genus zero quasimap theory and mirror maps. *Algebraic Geometry*, 1(4):400–448, 2014.

- [14] Ionuţ Ciocan-Fontanine and Bumsig Kim. Big *i*-functions. In *Development of Moduli Theory: Kyoto 2013*, volume 69 of *Advanced Studies in Pure Mathematics*, pages 323–348, 2016.
- [15] Ionuţ Ciocan-Fontanine, Bumsig Kim, and Davesh Maulik. Stable quasimaps to GIT quotients. *Journal of Geometry and Physics*, 75:17–47, January 2014.
- [16] Tom Coates, Alessio Corti, Hiroshi Iritani, and Hsian-Hua Tseng. Computing genus-zero twisted Gromov-Witten invariants. *Duke Mathematical Journal*, 147(3):377–438, 2009.
- [17] Tom Coates, Alessio Corti, Hiroshi Iritani, and Hsian-Hua Tseng. Some applications of the mirror theorem for toric stacks. *ArXiv e-prints*, January 2014. arXiv:1401.2611.
- [18] Tom Coates, Alessio Corti, Hiroshi Iritani, and Hsian-Hua Tseng. A mirror theorem for toric stacks. *Compositio Mathematica*, 151:1878–1912, October 2015.
- [19] Tom Coates, Hiroshi Iritani, and Yunfeng Jiang. The crepant transformation conjecture for toric complete intersections. *ArXiv e-prints*, September 2014. arXiv:1410.0024.
- [20] Tom Coates, Hiroshi Iritani, and Hsian-Hua Tseng. Wall-crossings in toric Gromov–Witten theory I: crepant examples. *Geometry & Topology*, 13(5):2675–2744, 2009.
- [21] Dan Edidin. Notes on the construction of the moduli space of curves. In Geir Ellingsrud, William Fulton, and Angelo Vistoli, editors, *Recent Progress in Intersection Theory*, pages 85–113. Birkhäuser Boston, Boston, MA, 2000.
- [22] Dan Edidin and William Graham. Localization in equivariant intersection theory and the Bott residue formula. *American Journal of Mathematics*, 120(3):619–636, 1998.
- [23] Torsten Ekedahl. Boundary behaviour of Hurwitz schemes. In *The moduli space of curves* (*Texel Island*), volume 129 of *Progress in Mathematics*, pages 173–198. Birkhäuser, 1995.
- [24] Honglu Fan and Y.-P. Lee. On Gromov-Witten theory of projective bundles. *ArXiv e-prints*, 2016. arXiv:1607.00740.
- [25] Bohan Fang, Chiu-Chu Melissa Liu, and Zhengyu Zong. All genus open-closed mirror symmetry for affine toric Calabi-Yau 3-orbifolds. *Proceedings of Symposia in Pure Mathematics*, 93:1, 2015.
- [26] W Fulton and R Pandharipande. Notes on stable maps and quantum cohomology. In Algebraic Geometry Santa Cruz 1995, Part 2, volume 62 of Proceedings of Symposia in Pure Mathematics, pages 45–96, 1997.
- [27] Alexander Givental. A mirror theorem for toric complete intersections. In Masaki Kashiwara, Atsushi Matsuo, Kyoji Saito, and Ikuo Satake, editors, *Topological Field Theory, Primitive Forms and Related Topics*, pages 141–175. Birkhäuser Boston, Boston, MA, 1998.
- [28] Alexander Givental. The mirror formula for quintic threefolds. *ArXiv Mathematics e-prints*, July 1998. arXiv: 9807070.

- [29] Tom Graber and Rahul Pandharipande. Localization of virtual classes. *Inventiones mathematicae*, 135(2):487–518, 1999.
- [30] Joe Harris and David Mumford. On the Kodaira dimension of the moduli space of curves. *Inventiones Mathematicae*, 67:23–86, 1982.
- [31] Brendan Hassett. Moduli spaces of weighted pointed stable curves. *Advances in Mathematics*, 173(2):316–352, 2003.
- [32] Kentaro Hori, Sheldon Katz, Albrecht Klemm, Rahul Pandharipande, Richard Thomas, Cumrun Vafa, Ravi Vakil, and Eric Zaslow. Mirror symmetry, clay mathematics monographs 1. *American Mathematical Society, Providence, RI*, 1, 2003.
- [33] Yi Hu and Jun Li. Derived resolution property for stacks, Euler classes and applications. *Mathematical Research Letters*, 18(4):677–690, 2011.
- [34] Tyler J. Jarvis and Takashi Kimura. The orbifold quantum cohomology of the classifying space of a finite group. In *Contemporary Mathematics: Orbifolds in Mathematics and Physics*, volume 310, 2002.
- [35] Yunfeng Jiang, Hsian-Hua Tseng, and Fenglong You. The quantum orbifold cohomology of toric stack bundles. *Letters in Mathematical Physics*, 107(3):439–465, 2017.
- [36] Paul Johnson. Equivariant Gromov-Witten theory of one-dimensional stacks. *Communications in Mathematical Physics*, 327(2):333–386, 2014.
- [37] Sheldon Katz. On the finiteness of rational curves on quintic threefolds. *Compositio Mathematica*, 60(2):151–162, 1986.
- [38] Sean Keel and Shigefumi Mori. Quotients by groupoids. *Annals of Mathematics, second series*, 145(1):193–213, 1997.
- [39] Bumsig Kim. Quantum hyperplane section theorem for homogeneous spaces. *Acta mathematica*, 183(1):71–99, 1999.
- [40] Joachim Kock. Notes on psi classes, 2001. Available at http://mat.uab.es/ kock/GW/notes/psinotes.pdf.
- [41] Maxim Kontsevich. Enumeration of rational curves via torus actions. In Robbert H. Dijkgraaf, Carel F. Faber, and Gerard B. M. van der Geer, editors, *The Moduli Space of Curves (Texel Island)*, pages 335–368. Birkhäuser Boston, Boston, MA, 1995.
- [42] Jun Li and Gang Tian. Virtual moduli cycles and Gromov-Witten invariants of algebraic varieties. *Journal of the American Mathematical Society*, 11(1):119–174, 1998.
- [43] Si Li. *Calabi-Yau Geometry and Higher Genus Mirror Symmetry*. PhD thesis, Harvard University, December 2011.
- [44] Bong Lian, Kefeng Liu, and Shing-Tung Yau. Mirror principle I. Asian Journal of Mathematics, 1(4):729–763, 1997.

- [45] Bong Lian, Kefeng Liu, and Shing-Tung Yau. Mirror principle III. Asian Journal of Mathematics, 3(4):771–800, 1999.
- [46] Chiu-Chu Melissa Liu. Localization in Gromov-Witten theory and orbifold Gromov-Witten theory. In Gavril Farkas and Ian Morrison, editors, *Handbook of Moduli*, volume II of *Advanced Lectures in Mathematics*, pages 353–425. International Press of Boston, Inc., Boston, MA, 2013.
- [47] Andrey Losev and Yuri Manin. New moduli spaces of pointed curves and pencils of flat connections. *The Michigan Mathematical Journal*, 48(1):443–472, 2000.
- [48] Han-Bom Moon. *Birational geometry of moduli spaces of curves of genus zero*. PhD thesis, Seoul National University, 2011.
- [49] Andrei Mustață and Anca Magdalena Mustață. Gromov-witten invariants for varieties with \mathbb{C}^* action. ArXiv e-prints, 2015. arXiv:1505.01471.
- [50] Andrei Okounkov and Rahul Pandharipande. Quantum cohomology of the Hilbert scheme of points in the plane. *Inventiones mathematicae*, 179(3):523–557, 2010.
- [51] Martin Olsson. On (log) twisted curves. *Compositio Mathematica*, 143(2):476–494, February 2007.
- [52] Miles Reid. Minimal models of canonical 3-folds. In *Algebraic Varieties and Analytic Varieties*, Advanced Studies in Pure Mathematics 1, pages 131–180, 1983.
- [53] Yongbin Ruan. The cohomology ring of crepant resolutions of orbifolds. In Tyler Jarvis, Takashi Kimura, and Arkady Vaintrob, editors, *Contemporary Mathematics: Gromov-Witten Theory of Spin Curves and Orbifolds*, volume 403. American Mathematical Society, 2006.
- [54] Hsian-Hua Tseng. Orbifold quantum Riemann-Roch, Lefschetz and Serre. *Geometry and Topology*, 14:1–81, 2010.
- [55] Ravi Vakil. Murphys law in algebraic geometry: badly-behaved deformation spaces. *Inventiones mathematicae*, 164(3):569–590, 2006.
- [56] Jonathan Wise. The genus zero Gromov-Witten invariants of the symmetric square of the plane. *Comm. Anal. Geom*, 19(5):923–974, 2011.
- [57] Aleksey Zinger. The reduced genus 1 Gromov-Witten invariants of Calabi-Yau hypersurfaces. *Journal of the American Mathematical Society*, 22(3):691–737, 2009.